

The Role of Social Class in the Formation of Voting Intentions and Product Choices: A Social Cognitive Perspective

Inaugural dissertation submitted in partial fulfillment of the requirements for the degree Doctor of Social Sciences in the Graduate School of Economic and Social Sciences at the University of Mannheim

By

Fabienne Unkelbach

Supervisors: Prof. Dr. Michaela Wänke (1st Supervisor) Prof. Dr. Herbert Bless

Dean of the Faculty of Social Sciences: Prof. Dr. Michael Diehl

Thesis Evaluators: Prof. Dr. Herbert Bless Prof. Dr. Jochen Gebauer

Thesis Defense: April 23, 2024

Preface

This cumulative dissertation is based on the three manuscripts listed below. The first two manuscripts have been published in peer-reviewed journals and the third manuscript has been submitted for publication. Manuscript 2 was originally submitted as a Registered Report as part of the GLES Open Science Challenge 2021.

- Unkelbach, F., Brütting, T., Schilling, N., & Wänke, M. (2023). Looking competent does not appeal to all voters equally: The role of social class and politicians' facial appearance for voting likelihood. *Personality and Social Psychology Bulletin*, 0(0). <u>https://doi.org/10.1177/01461672231181465</u>
- Unkelbach, F., John, M., & Vogel, V. (2023). Jumping on the bandwagon: The role of voters' social class in poll effects in the context of the 2021 German federal election. *Politische Vierteljahresschrift*, 64(1), 51–78. <u>https://doi.org/10.1007/s11615-022-00417-3</u>
- Unkelbach, F., Kehder, P., & Wänke, M. (2023). Do consumers of all social classes prefer best sellers? Three pre-registered replication studies of Na et al. (2016). Manuscript submitted for publication.

One additional manuscript that is not part of this dissertation but adds to its insights is currently in preparation. This manuscript is listed below:

Batruch, A., Unkelbach, F., Prisacaru, N., & Wänke, M. (2023). Social class and interdependent self-construal: Two large-scale studies from the U.S. and the German context. Manuscript in preparation.

Acknowledgments

I would like to thank several wonderful people without whom this dissertation would not have been possible:

First of all, I am deeply grateful to you, **Michaela**, for your guidance, invaluable feedback, and encouragement at all stages of my PhD. Thank you so much for everything I have learned from you!

Second, I would like to thank you, **Herbert** and **Jochen**, for several stimulating conversations about social class over the last few years and for agreeing to provide the expert reviews for this dissertation.

Third, I want to thank all of my **co-authors** for their great teamwork. It was a real pleasure to work with you!

Fourth, I want to thank all my dear colleagues and friends – **Belinda, Freddy, Hilal, Lara, Laura, Melvin, Moritz, Ness, and Nils** – for your emotional and scientific support, joyful lunch breaks and for providing me with endless coffee when needed. You made this journey a great experience!

Fifth, a special thank you goes to my wonderful friends who have been with me through all stages of this dissertation and always believed in me – **Moni, Franzi, and Julius.** I am so grateful to have you in my life!

Finally, I am deeply grateful to my **Parents** and of course to you, **Max**, for being my greatest supporters in all my endeavors and for always being by my side. This dissertation is dedicated to you!

Table of Contents

1	Intro	oduction and Overview	1	
2	2 Social Cognitive Perspective on Social Class			
	2.1	Conceptualization of Social Class in Social Psychology	3	
	2.2	Class-Based Differences in the 'Big Two'	4	
	2.3	Social Cognitive Effects of Social Class: A Brief Overview	5	
3	Seleo	cted Effects of Social Class in the Political and Consumer Context	6	
	3.1	Social Class and the Importance of Politicians' Perceived Competence	6	
	3.2	Social Class and the Susceptibility to Majority Influence	7	
4	Shor	rtcomings of Prior Research	8	
5	Sum	mary of the Manuscripts	9	
	5.1	Manuscript 1: Social Class and Facial-Appearance Effects	0	
	5.2	Manuscript 2: Social Class and the Political Bandwagon Effect 12	2	
	5.3	Manuscript 3: Social Class and Majority Influence in Product Choices 1	3	
	5.4	Summary of the Main Findings	4	
6	Gen	eral Discussion1	5	
	6.1	Theoretical Implications for Social Cognitive Accounts of Social Class 1	5	
	6.2	Limitations and Directions for Future Research	9	
	6.3	Practical Implications	1	
	6.4	Conclusion	2	
7	Refe	rences	3	
A	Appendix: Manuscripts			

1 Introduction and Overview

In recent years, political actors around the world have become increasingly aware of how large disparities in individuals' socio-economic living conditions can threaten the functioning of modern societies (see e.g., Department of Economic and Social Affairs of the United Nations, 2020; OECD Centre for Opportunity and Equality, 2017). For example, the United Nations declared socio-economic inequality to be one of the major challenges of the 21st century (Department of Economic and Social Affairs of the United Nations, 2016). Such disparities in social class have far-reaching implications for almost every aspect of individuals' lives (e.g., Kraus et al., 2012), including their mental and physical well-being (Assari, 2017; Fein, 1995; McLeod et al., 2012; Singh-Manoux et al., 2005), and even life expectancy (Stringhini et al., 2017).

While the role of social class in life outcomes has been studied in other disciplines for centuries (e.g., Bourdieu, 1979/1984; Marx & Engels, 1848/2017; Weber, 1904-05/1958), psychologists have long overlooked its relevance for individuals' psychological functioning (Kraus & Stephens, 2012). In the past two decades there has been a radical shift, beginning with the development of psychological models of social class that emphasize the effects of social class on people's thoughts, feelings, and behaviors (Kraus et al., 2012; Kraus & Stephens, 2012). Within these models, social class is conceptualized as including both *objective* social status (SSS), which refers to a person's perception of their relative standing in the societal hierarchy (Kraus et al., 2012; Kraus et al., 2013).

One fundamental premise of social cognitive models of social class is that making repeated experiences in certain social class contexts shapes individuals' psychological makeup (see e.g., Kraus et al., 2012). Recent approaches conceptualized these class-based differences in psychological tendencies referring to the two broad dimensions of personality and social judgment – agency and communion (Bakan, 1966; see also Abele & Wojciszke, 2007; Wiggins, 1991). While higher-class individuals have been assumed to show a larger tendency to be oriented towards promoting their self (i.e., agentic), lower-class individuals have been assumed to show a larger tendency to be oriented towards promoting their self (i.e., agentic), lower-class individuals have been assumed to show a larger tendency to be oriented towards others and their social environment (i.e., communal) (e.g., Kraus et al., 2012; Rucker et al., 2018). In line with these assumptions, a large number of studies found fundamental differences in how individuals from different social classes perceive and judge their self and others, perceive emotions, build relationships with others, behave in social situations, and make decisions (for overviews, see Fiske & Markus, 2012; Kraus et al., 2012; Manstead, 2018).

Based on three manuscripts with a focus on the political and the consumer context, this dissertation provides comprehensive new insights into the role of social class in individuals' decision-making. *First*, it sheds some light on class-based differences in the effects of politicians' perceived competence on the likelihood of voting for a particular candidate. Initial evidence from the U.S. has suggested that higher-class individuals value competence in politicians more than lower-class individuals (e.g., Callaghan et al., 2022). Manuscript 1 extends this research by examining the role of social class in facial appearance effects on voting intentions in the German context. *Second*, this dissertation provides empirical tests of the role of social class in majority influence. Prior research based on U.S. samples has shown that people from lower social classes are more susceptible to the

preferences of others when choosing products than people from higher social classes (Na et al., 2016; Stephens et al., 2007). To examine the generalizability of these social class-based differences in conformity, Manuscript 2 investigated them in the context of poll effects on voting intentions. Manuscript 3 completes the picture by examining the replicability of findings from previous research on class-based differences in susceptibility to majority influence in the context of product choice, using German and U.S. samples. All manuscripts addressed methodological limitations of previous research. Overall, the present research contributes to a deeper understanding of the generalizability and potential boundary conditions of social cognitive class-based differences. In the following, I provide brief summaries of the manuscripts and an overview of the structure of this dissertation.

Manuscript 1 had two main goals. First, we¹ aimed to clarify whether voters with higher SES value politicians' perceived competence, a sub-facet of agency, relatively more than voters with lower SES when forming voting intentions. Second, we aimed to shed light on competence as a schematic trait among higher-class voters. To achieve these goals, we used correlational survey data from a large-scale representative German sample and conducted three online experimental studies which manipulated politicians' perceived competence in the distribution of politicians (second previous research by demonstrating that individuals with higher SES place greater weight on politicians' faces. Additionally, this research provides evidence for self-perceived competence as an underlying psychological mechanism. In conclusion, Manuscript 1 highlights the importance of class-based differences in individuals' perspective on themselves, their evaluations of politicians and thus the formation of voting intentions.

Manuscript 2 had the goal to identify potential social class-based differences in voters' tendency to conform to a perceived majority in their voting intentions. Notably, conformity is typically associated with communion (see Gebauer et al., 2013). We investigated whether social class moderates so-called political bandwagon effects in the 2021 German federal election. These effects refer to the tendency to adjust one's voting intention to support a party perceived as the "winner" in pre-election poll results (see e.g., Schmitt-Beck, 2015). For this purpose, we combined data from the German Longitudinal Election Study (GLES) Rolling Cross-Section 2021 (RCS; GLES, 2022), which consists of representative samples assessed daily for 55 days before the 2021 German federal election, with the results of published pre-election polls. The findings indicated limited poll effects and did not support the assumption that the tendency to follow a perceived majority preference differs across social classes. Thus, they raise questions about the nature of bandwagon effects in multiparty systems and cast doubt on the relationship between social class and this aspect of communion.

Manuscript 3 further examined the replicability and cross-cultural generalizability of the relationship between social class and the tendency to follow a majority in the German and the U.S. context, with a focus on purchase decisions. Specifically, we conducted three replication studies of Na et al.'s (2016) original research. The authors had shown that individuals from lower social class backgrounds change their product choices more frequently

¹ In the following, "we" refers to all authors who contributed to the respective manuscript.

to conform to the perceived majority product choice. Across three studies, two conducted in Germany and one in the U.S., our results suggest that the relationship between social class and conformity in product choices is less robust than previously assumed. Additionally, we investigated whether social class is related to a chronic interdependent self-construal. This perspective on the self is related to communion (Abele et al., 2016; Abele & Wojciszke, 2007) and was proposed as an underlying mechanism of the social class effects found by Na et al. (2016). However, we did not find the assumed negative relationship between social class and an interdependent self-construal. In summary, these findings raise important concerns about the generalizability of previously found social class-based differences in communion.

After this introduction, the present dissertation is structured as follows: Section 2 introduces the conceptualization of social class in recent social psychological research and the assumption of fundamental class-based differences in agency and communion. Additionally, it provides a brief overview of psychological effects of social class. Section 3 presents relevant findings on the role of social class in the formation of voting intentions and product choices, followed by a description of the limitations of previous research in Section 4. Section 5 includes summaries of the three manuscripts underlying this dissertation and explains how they address the limitations of prior research. Finally, Section 6 provides a comprehensive discussion of the implications of the present findings, their limitations, and directions for future research on the role of social class in the political and consumer contexts.

2 Social Cognitive Perspective on Social Class

2.1 Conceptualization of Social Class in Social Psychology

Only in the last decades has social class moved into the focus of social psychologists as an important predictor of central psychological processes (see e.g., Kraus & Stephens, 2012). Building on earlier social scientific accounts that recognized shared characteristics of individuals with similar social class backgrounds (e.g., Bourdieu, 1979/1984; Kohn, 1969), Kraus et al. (2012) proposed a seminal social cognitive framework of social class to explain how social class shapes individuals' thoughts, feelings, and behaviors. This framework defines social class as an important social context in which individuals grow up and live (see also Stephens et al., 2014) and argues that social class contexts are characterized by both objective and subjective components of social class (e.g., Kraus & Stephens, 2012).

Objective components of social class particularly refer to individuals' level of access to material resources or, in other words, to the question of how much one possesses. These are typically measured via SES, using one or several indicators such as level of educational attainment, household income level, and occupational prestige (Kraus & Stephens, 2012; Oakes & Rossi, 2003).

Subjective components of social class are often operationalized as SSS, which refers to an individual's chronic perception of their position in the societal hierarchy relative to others (Kraus & Stephens, 2012; Kraus et al., 2013). In other words, SSS describes how much one believes to possess compared to others. Thus, it includes an assessment of one's material resources but also, going beyond SES, social comparisons on these typical indicators of SES (see Tan et al., 2020). SSS has additionally been found to reflect psychological variables such as the evaluation of the personal present and future economic circumstances (Singh-Manoux et al., 2003, see also Tan et al., 2020). A widely used measure for SSS is the MacArthur Scale

(Adler et al., 2000), which assesses how individuals rank their standing in society on a ladder with ten rungs representing societal ranks regarding income, educational attainment, and occupational prestige.

Both objective SES and SSS shape and systematically organize the social contexts in which individuals live (e.g., Kraus et al., 2012). However, they are conceptualized as distinct components of social class, as reflected in merely a moderate association between the two constructs (Tan et al., 2020: meta-analytic r = .32). The effects of SSS can go above and beyond the effects of SES, as is the case for health outcomes (Adler et al., 2000; Singh-Manoux et al., 2005) or even point into the opposite direction, as is the case for system justification (Li et al., 2020). In order to adequately account for the complexity of the construct, I examine the role of both components of social class in this dissertation.

2.2 Class-Based Differences in the 'Big Two'

A central premise of social cognitive approaches to social class is that being socialized and living in different social class contexts promotes the development of divergent psychological orientations, which are shared by individuals within a social class (Kraus et al., 2012; Kraus & Stephens, 2012). As higher social class contexts are characterized by abundant resources, more stability, freedom, and a high perceived societal rank, individuals making repeated experiences in these contexts tend to be more oriented *towards their self*, including own goals and internal states (e.g., Carey & Markus, 2017; Kraus et al., 2012; Kraus & Stephens, 2012). In contrast, individuals making repeated experiences in lower class-contexts, which are characterized by limited resources, high levels of constraints, few opportunities to make free choices, as well as a low perceived standing in society, tend to be more oriented *towards their social relationships and external environment* (Carey & Markus, 2017; Kraus et al., 2012; Kraus & Stephens, 2012).

Recent accounts have attempted to systematize these broad, divergent psychological tendencies among higher- compared to lower-class individuals by referring to the "Big Two" dimensions of personality and social judgments – agency and communion² (Abele & Wojciszke, 2007; Bakan, 1966). More concretely, individuals from higher social classes are assumed to be more agentic, whereas individuals from lower social classes are assumed to be more communal (Rucker et al., 2018). Within the Big Two model, agency is broadly defined as a general tendency to promote the self ("getting ahead", Hogan, 1982) and to strive to distinguish oneself from others (Abele et al., 2016; Abele & Wojciszke, 2007, 2014; Gebauer et al., 2013). Importantly, agency comprises the sub-facets of competence, referring to one's abilities, and assertiveness, emphasizing motivational aspects of pursuing one's goals (Abele et al., 2016). In contrast, communion is defined as a general tendency to promote one's relationships with others ("getting along", Hogan, 1982) and reflects one's perspective on the self in relation to others and striving for assimilation with others. Warmth and morality have

² Other theoretical accounts have used different terms referring to similar underlying constructs to describe psychological tendencies related to social class, for example solipsism and contextualism (Kraus et al., 2012) or self- and other-orientation (Gobel & Miyamoto, 2024). In this dissertation, I draw on the concepts of agency and communion as they not only constitute broad, comprehensive dimensions of self- and other-perceptions (Abele et al, 2016), but also provide a useful framework for studying how individuals make choices (Kurt & Frimer, 2015).

been identified as sub-facets of this dimension (Abele et al., 2016). Notably, recent approaches argue that agency and communion constitute separate orthogonal dimensions, suggesting that it is generally possible for individuals to be both highly agentic *and* highly communal (Abele & Wojciszke, 2007; Rucker et al., 2018).

2.3 Social Cognitive Effects of Social Class: A Brief Overview

Over the past two decades, numerous empirical studies have been conducted based on the assumption of class-based differences in broad psychological tendencies. In the following, I present a brief overview of influential findings in social psychological research on social class, including its effects on self-perception, the perception of and the interaction with others, and decision-making.

First and foremost, prior research has established class-based differences in the perspective on the self. Specifically, living in higher-class contexts has been associated with a higher self-esteem (e.g., Kan et al., 2014) and a higher perceived control over life outcomes (John et al., 2023; Johnson & Krueger, 2006; Kraus et al., 2009; Lachman & Weaver, 1998). Using concepts from cultural psychology, further research showed that higher-class individuals (as measured via parental education) are more likely than lower-class individuals to develop an *independent* self-construal (e.g., Fernández, 2005). This type of construal of the self refers to viewing the self as a unique, separate entity (e.g., Markus & Kitayama, 1991; Singelis, 1994) and is associated with the broad dimension of agency (e.g., Abele et al., 2016). Emphasizing the existence of a positive relationship between social class and self-perceived agency, further studies suggest that higher-class individuals tend to perceive themselves as more competent than lower-class individuals do (Abele, 2003; Belmi et al., 2020; Oldmeadow & Fiske, 2010).

Living in lower-class contexts, on the other hand, has been associated with a tendency to explain life outcomes by referring to contextual influences outside one's personal control (Kraus et al., 2009). Additionally, there is some research showing that lower-class individuals are more likely than higher-class individuals to develop an *interdependent* self-construal (Fernández et al. 2005; Grossmann & Varnum, 2011). This type of construal of the self refers to viewing the self as embedded in social relationships (e.g., Markus & Kitayama, 1991; Singelis, 1994) and is associated with the broad dimension of communion (e.g., Abele et al., 2016). However, the empirical evidence for class-based differences in interdependent self-construal is mixed. Some research has found no significant association (Stephens et al., 2007), while other research has found support only for certain measures of interdependent self-construal and not for others (Na et al., 2010). In summary, there exists conclusive evidence of a positive relationship between social class and self-perceived agency, whereas evidence for the often-assumed negative relationship between social class and self-perceived communion appears to be more limited (see also Boileau, 2022).

Second, prior research has established social class-based differences in social perception and individuals' relationships with others. In line with the assumption of lowerclass individuals being more oriented towards others, or communal, than higher-class individuals, they tend to have a higher accuracy when detecting others' emotions (Kraus et al., 2010). Furthermore, it was found that lower-class individuals tend to show more otheroriented emotions, such as compassion, while higher-class individuals tend to show more selforiented emotions, such as pride (Piff & Moskowitz, 2018). Adding to this line of research, it could be demonstrated that when perceiving adversities, lower-class individuals tend to focus on their connections with others, while higher-class individuals tend to rely on their personal wealth (Piff et al., 2012). Consistent with this finding, as social class increases, the proportion of people in one's social network who belong to the inner circle decreases (Na et al., 2010). Taken together, these findings on social class-based differences in social perception and relationships support the notion of class-based differences in agency and communion.

Third, effects of social class have been found in the context of decision making. One line of research has demonstrated that the higher perceived uncertainty and lack of stability associated with a lower social class leads to a higher risk aversion, more prosocial decisions, and a greater tendency to prioritize current over future goals (e.g., Amir et al., 2018; for an overview, see Sheehy-Skeffington, 2020). Importantly, there may also be social class-based differences in the meaning attached to the act of decision making (e.g., Snibbe & Markus, 2005; Stephens et al., 2011). While higher-class individuals tend to view choices as a means of establishing their uniqueness and expressing their own preferences, lower-class individuals tend to make choices with the goal to strengthen their relationships with others (Stephens et al., 2011; Stephens et al., 2007; see also Snibbe & Markus, 2005). In line with this assumption, prior research found that lower-class individuals are more likely to make choices that are similar to the choices of others (Stephens et al., 2007) and tend to follow others' choices (Na et al., 2016). Section 3.2 outlines these and related findings in more detail. In summary, these findings suggest that higher-class individuals tend to be more agentic and lower-class individuals tend to be more communal in their decision-making (see Rucker et al., 2018).

Looking at this brief, by no means exhaustive overview of the empirical evidence for social class effects found in social psychological research over the last decades, it seems reasonable to conclude that individuals' social class plays an important role in almost every aspect of their lives and, in particular, in the way they make decisions. Extending the abovementioned findings, the manuscripts underlying the present dissertation focus on the role of social class in two important decision-making contexts in modern democracies, the political and the consumer context. More concretely, they investigated social class-based differences in a) the importance of politicians' perceived competence (Manuscript 1) and b) the tendency to follow a majority's preference in voting intentions (Manuscript 2) and product choices (Manuscript 3). In the following, I outline relevant prior research in the political and the consumer context and describe its main limitations, before providing a summary of the three manuscripts that form the basis of this dissertation.

3 Selected Effects of Social Class in the Political and Consumer Context

3.1 Social Class and the Importance of Politicians' Perceived Competence

In the political context, an important interpersonal aspect that influences voting decisions is voters' perception of politicians' character traits (Funk, 1999; Kinder et al., 1980; Laustsen & Bor, 2017; Miller et al., 1986). In particular, perceived competence as one subfacet of agency has been found to increase the likelihood of voting for a candidate (Antonakis & Dalgas, 2009; Todorov et al., 2005). Research on the effects of facial appearance even suggests that merely looking competent is beneficial for a politicians' electoral success

(Antonakis & Dalgas, 2009; Ballew & Todorov, 2007; Olivola & Todorov, 2010a, 2010b; Todorov et al., 2015).

Importantly, initial evidence based on data from the American National Election Study (ANES) suggests that voters with higher SES (measured via educational attainment) value competence in politicians more compared to voters with lower SES (Kinder et al., 1980; Miller et al., 1986). In line with this result, a first experimental study based on verbal descriptions of fictional politicians found that the higher the voters' SES (measured via a composite score based on household income and education), the higher the likelihood of voting for a competent candidate (Callaghan et al., 2022). Furthermore, this research found that a higher social class was associated with greater interpersonal closeness with candidates who appeared more competent. This might be one explanation for the effect and is consistent with the finding that higher-class individuals perceive themselves as more agentic (see e.g., Oldmeadow & Fiske, 2010). However, it should be noted that Callaghan et al. (2022) relied on verbal descriptions of fictitious candidates presented as either high in competence or high in warmth. This approach neglects the fact that voters perceive politicians on both dimensions simultaneously and that politicians can vary in their level of perceived competence.

Taken together, these studies provide initial evidence for a positive relationship between social class and the importance placed on politicians' perceived competence as a subfacet of agency. One goal of this dissertation is to extend these findings (see Section 5.1, Manuscript 1). A second objective is to shed light on how fundamental class-based differences in agency and communion are manifested in conformity to the preferences of others. The next section presents preliminary evidence from prior studies regarding this relationship.

3.2 Social Class and the Susceptibility to Majority Influence

The fact that individuals are influenced in their decision-making by the perceived preferences of others has long been established in different disciplines. In social psychological research, individuals' tendency to follow a majority viewpoint has often been conceptualized as *conformity* (e.g., Asch, 1956). Similarly, political scientists have used the term *political bandwagon effect* to describe the phenomenon that voters tend to shift their voting intentions to support parties that are perceived as preferred by a majority (Moy & Rinke, 2012; Mutz, 1998; Schmitt-Beck, 1996). Importantly, this type of bandwagon effect is categorized as a poll effect as individuals particularly learn about majority's voting intentions based on the results of pre-election polls (e.g., Schmitt-Beck, 2015). Finally, in the consumer context, Cialdini (1993) coined the term *social proof heuristic* to describe the tendency of consumers to buy products that appear to be popular. Taken together, majority influence plays a pervasive role in individuals' decisions across several contexts.

Importantly, initial evidence suggests that individuals' tendency to follow the decision of a majority depends on their social class. In support of the idea that social class is negatively related with communion, prior research found that individuals from lower social class (as measured via parental educational attainment) were more likely to follow the product choice of another participant compared to individuals from higher social class (Stephens et al., 2007). Similarly, they preferred to let others choose for them as opposed to making their own choices (Stephens et al, 2011) which emphasizes their focus on others' preferences. In contrast, higher-class individuals did not prefer any of these options over the other. Finally, Na et al. (2016) found that individuals with lower social class (measured via maternal education and SSS) are more likely to change their product choice in order to align with the perceived majority preference than individuals from higher social classes (Na et al., 2016, Study 1). In summary, these findings support the notion that lower-class individuals follow the typical communal motive to assimilate with others whereas higher-class individuals tend to distinguish themselves from others and focus on their own interests through their choices (see e.g., Rucker et al., 2018).

Nevertheless, important questions remain unanswered. First, the role of social class in the susceptibility to majority influence has merely been investigated in the context of product choices and aesthetic preferences. It remains open whether social class could also moderate political bandwagon effects. Second, there is currently insufficient empirical evidence regarding the psychological mechanism underlying these class-based differences. While researchers have postulated that an interdependent self-construal might mediate the relationship between social class and conformity (Na et al., 2016; Stephens et al., 2007), empirical evidence linking social class and this type of self-construal does not appear to be robust (see Section 2.3). Even though Na et al. (2016) showed that the relationship between social class and conformity turns non-significant when controlling for primed interdependent self-construal versus independent self-construal, this approach is not sufficient to establish a comprehensive mediation model (see Pirlott & MacKinnon, 2016).

4 Shortcomings of Prior Research

Having summarized previous research on the role of social class in a) the weight attached to politicians' competence and b) the susceptibility to majority influence, it is important to note that both strands of research have several methodological limitations that may restrict the generalizability of their findings.

First, prior research on the role of social class in the political and the consumer context has often neglected the multifaceted nature of the construct. As outlined above, initial evidence on social class-based differences in the weight attached to politicians' competence using ANES data is based on voters' educational attainment as the sole indicator of SES (Kinder et al., 1980; Miller et al., 1986). Similarly, seminal studies on class-based differences in consumer choices have relied on own or parental educational attainment (for students) as (artificially dichotomized) single indicator of SES and interpreted it as proxy for social class (Snibbe & Markus, 2005; Stephens et al., 2011; Stephens et al., 2007). However, this approach does not take into account that social class contexts are shaped by multiple objective and subjectively perceived resources, which are related but represent distinct aspects linked with different experiences (e.g., Kraus et al., 2012; Kraus & Stephens, 2012; Kraus et al., 2013). In line with this premise, prior research found that SSS predicts outcomes such as policy preferences (Brown-Iannuzzi et al., 2015) or political efficacy (Kraus et al., 2015) over and above indicators of SES. To capture the multifaceted nature of social class conclusively and allow for comparisons of effects across different indicators, all manuscripts of the present dissertation relied on multiple indicators of SES as well as on a measure of SSS.

Second, prior empirical research has been restricted in generalizability due to predominantly focusing on the U.S. context (e.g., Callaghan et al., 2022; Na et al., 2016;

Stephens et al., 2007). Notably, there are several characteristics of this context expected to enhance social class effects. Importantly, the U.S. is characterized by a comparatively high economic inequality (Gini index 2021: 37.4; Solt, 2019) - a factor assumed to increase social class effects due to a higher salience of social disparities (Schneider, 2019). For example, the perception of higher-class individuals as more competent than lower-class individuals increases with higher levels of economic inequality (Connor et al., 2021; Durante et al., 2017). Furthermore, the U.S.A. is considered to be the most individualistic country in the world (Hofstede et al., 2010). Previous research has shown that the positive relationship between social status and individualistic traits is stronger in more individualistic cultures as individuals who endorse values that are important in a culture are more likely to achieve higher social status (Gobel & Miyamoto, 2024; Zhang et al., 2021). Additionally, individuals from higher social classes are more expected to promote cultural ideals (e.g., Miyamoto et al., 2018). With regard to the present research, this suggests that a) the positive relationship between social class and the importance of politicians' competence and b) the negative relationship between social class and conformity might be particularly pronounced in the U.S. context. To develop a better understanding of social class effects and to increase external validity, it appears crucial to shed light on the role of the cultural context as a potential boundary condition. Thus, Manuscripts 1 and 2 examined social class-based differences with German instead of U.S. samples, and Manuscript 3 is based on data from both countries. Germany as standard of comparison appears informative as it is characterized by a smaller level of income inequality (Gini index 2021: 29.7; Solt, 2019) and a less individualistic culture compared to the U.S.A. (Hofstede et al., 2010), two factors that might reduce the social class effects of interest.

Third, particularly prior research on social class-based differences in the susceptibility to social influence predominantly relied on relatively small samples (median sample size per study in Na et al., 2016; Stephens et al., 2011; Stephens et al., 2007: N = 94 participants) and tested social class effects with a rather low statistical power. As a low statistical power reduces the chance that a significant finding reflects a true effect and increases the likelihood of inflated estimates for found effects (Button et al., 2013), the replicability of prior findings appears questionable. Furthermore, most of these studies relied on student samples, which are likely to be limited in the variance of participants' social class effects, all manuscripts of the present dissertation incorporated pre-registered studies based on highly powered samples that were diverse regarding social class. The data was collected in the course of several experimental online studies (Manuscripts 1 and 3) or from representative large-scale surveys (Manuscripts 1 and 2). Additionally, to assess the replicability of previously found social class-based differences in the susceptibility to social influence, Manuscript 3 was based on three replication studies of research by Na et al. (2016).

5 Summary of the Manuscripts

The three manuscripts of this dissertation had the primary goal of assessing the role of social class as a moderator of social factors previously identified as influencing voting intentions (Manuscripts 1 and 2) and product choices (Manuscript 2) (for a schematic overview, see Figure 1). Manuscript 1 tested whether politicians' perceived competence is

valued differently across voters' social class spectrum. Addressing the limitations of prior research in this field, it provides the first investigation of the role of social class as a moderator of facial appearance effects on voting likelihood. This approach allowed for a subtle and nuanced manipulation of perceived competence, thus, contributing to a more comprehensive understanding of the nature of class-based political preferences. Manuscripts 2 and 3 examined whether social class-based differences in the tendency to follow a perceived majority preference generalize to the political context (Manuscript 2) and are replicable in the consumer context (Manuscript 3). Thus, Manuscript 2 represents the first attempt to integrate social cognitive accounts of social class with research on the political bandwagon effect in order to shed new light on both the role of poll effects in modern democracies and the generalizability of social class effects. Manuscript 3 consists of three replication studies of Na et al.'s (2016) research in both the U.S. and German contexts. It provides deeper insights into the generalizability of the role of social class in the tendency to follow others' product choices, interdependent self-construal as underlying mechanism, and the role of the cultural context.

Figure 1

Schematic Overview of the Three Manuscripts Underlying the Present Dissertation



Note. Social class is investigated as a moderator of the effects of politicians' perceived competence respectively the perceived preference of a majority on voting intentions (Manuscripts 1 and 2). Additionally, social class is investigated as moderator of the effect of the perceived preference of a majority on product choices (Manuscript 3).

5.1 Manuscript 1: Social Class and Facial-Appearance Effects

The main goal of Manuscript 1 was to assess social class-based differences in the weight attached to politicians' perceived competence when forming voting intentions. We hypothesized that voters with higher SES would place greater importance on the perceived competence of political candidates than voters with lower SES. Additionally, the manuscript aimed to establish initial evidence for individuals' self-construal concerning competence as an underlying mechanism. We conducted four studies following a mixed methodological approach to provide comprehensive evidence. In all studies, SSS was additionally assessed via the MacArthur Scale (e.g., Adler et al., 2000).

Study 1 aimed to provide the first correlational evidence based on self-report data from a sample representative of the German general population regarding gender, age and educational attainment ($N_1 = 2239$). Participants rated the importance of competence in political candidates. Furthermore, they indicated how competent they perceived themselves. Consistent with the hypothesis, the higher voters' objective SES (computed as a composite score based on z-standardized educational attainment and monthly net household income, see Kraus & Keltner, 2009) the higher was the rated importance of competence in politicians. This relationship remained robust after controlling for the importance ratings of other traits such as assertiveness, warmth, and trustworthiness. Furthermore, we found preliminary evidence for a mediation via voters' self-perceived competence. However, voters' explicitly reported importance ratings might differ from what they actually take into consideration when forming voting intentions. To respond to this shortcoming, Study 2a/b aimed to extend the social class-based differences found in Study 1 to facial appearance effects on voting likelihood.

In Studies 2a/b ($N_{2a\&2b} = 396$), participants were presented with 10 pictures of male politicians' faces unknown to them individually in a randomized order. These pictures were selected based on pretest data to create one set of pictures with low and one with high perceived competence while keeping perceived warmth (as one aspect of communion) at a constant level. After rating each politician's perceived competence and warmth as a manipulation check, participants indicated their likelihood of voting for each candidate. Finally, we assessed participants' objective SES via educational attainment (for students: parental educational attainment) and current household income (for students: family household income). The results of a joint analysis indicated an overall higher likelihood of voting for politicians who are perceived as more competent. Importantly, this effect was moderated via participants' SES, with competence having a greater impact among those with higher SES.

Varying politicians' perceived competence through their facial appearance appears to be a less obtrusive method than manipulations used in prior research (e.g., Callaghan et al., 2022). Yet, measuring perceived candidate traits *before* the voting likelihood might have increased their accessibility and, consequently, their weighting. To prevent this, we altered the question order in Study 3 ($N_3 = 400$). Participants first indicated the voting likelihood for each candidate and subsequently rated perceived warmth and competence. Additionally, we expanded the number of pictures, covering a broader range of perceived competence levels. The results replicated a higher voting likelihood for politicians perceived as more competent. Although the moderation by participants' SES pointed in the same direction as in Studies 2a/b, it did not reach statistical significance (p = .052).

Taken together, the results across all four studies suggest that voters' SES moderates the effect of politicians' perceived competence on voting likelihood in the German context. Voters with higher SES appear to value competence in (male) politicians more compared to lower-class voters. Additionally, Manuscript 1 provides initial evidence that this effect might be explained by a higher self-perceived competence among voters with higher SES. Notably, the pattern for SSS diverged from the one for SES. A higher SSS was related to a higher reported importance of competence as well as of several other desirable candidate traits (Study 1). However, we did not find support for a moderation of competence effects on voting likelihood via SSS (Studies 2a/b, 3). A possible explanation for these results is outlined in the *General Discussion*. Although not in the focus of our research, we found preliminary evidence that voters' political orientation affects the size of the effects of perceived competence independently of SES.

5.2 Manuscript 2: Social Class and the Political Bandwagon Effect

Manuscript 2 examines whether social class moderates the influence of another element of voters' social environment on voting intentions – the perceived voting intentions of a majority of others as based on published poll results. Drawing on social cognitive accounts of social class (e.g., Kraus et al., 2012; Kraus & Stephens, 2012) and findings from consumer research (Na et al., 2016; Stephens et al., 2007), we predicted that these bandwagon effects on voting intentions would be larger for lower-class compared to higher-class voters. Thus, Manuscript 2 had two main objectives. First, we aimed to assess the generalizability of previously found social class-based differences in the tendency to follow a majority by investigating them in a different context. Second, we attempted to enhance the understanding of the size of the political bandwagon effect and clarify the inconclusive results of prior political science research (Barnfield, 2020) by introducing social class as a potential boundary condition.

We tested our hypothesis in the context of the 2021 German federal election, relying on data from the GLES RCS 2021 (GLES, 2022), a large-scale, cross-sectional survey. The RCS data was collected on a daily basis over the 55 days preceding the election date with the pre-set goal to assess a random sample of 130 respondents per day (total *N* after pre-registered exclusions = 5291). Importantly, the RCS included measures of respondents' voting intentions, indicators of their SES (educational attainment and current gainful employment status), which we combined into an SES score (e.g., Kraus & Keltner, 2009), the MacArthur Scale (Adler et al., 2000) assessing SSS³, information on whether they had noticed poll results in the previous week as well as further established predictors of voting intentions serving as control variables (i.e., interest in politics/the current election campaign, party identification, issue orientation, candidate orientation, respondent gender, and age).

To test potential poll effects, we matched data from the RCS with the results of published pre-election polls from the eight leading German polling institutes with a time lag of one day (see Faas et al., 2008; Hoffmann & Klein, 2013). As the data was nested in field days, we computed separate multilevel logistic regressions for each of the six parties which were part of the German national parliament. The interaction term between poll results for the party and voters' SES (and both main effects) were included as predictors of voting intentions for each political party. Merely for the SPD were poll results significantly positively related to voting intentions one day after the polls were published. Supporting the assumption of a bandwagon effect for the SPD, this relationship was larger when respondents had previously perceived polls. However, in contrast to our hypothesis, there were no significant social class-based differences in the size of poll effects for any of the parties. These results remained robust when a) controlling for the afore-mentioned standard predictors of voting intentions, b) using SSS instead of SES as a measure of social class, and c) using a time lag of zero or two days instead of one day.

³ SSS was assessed in the RCS postelection survey conducted after the federal election which was completed by only 63% of the respondents from the preelection survey.

In sum, these results do not support the assumption that voters of different social classes vary substantially in their tendency to follow others' political preferences as presented in published poll results. Thus, we did not find evidence that social class serves as a boundary condition of political bandwagon effects. These findings do not only have implications for future research on bandwagon effects in multiparty systems, but also call into question the generalizability of social class-based differences in conformity. Notably, prior studies on the relationship between social class and conformity focused on product choices and were predominantly conducted in the U.S. context (e.g., Na et al., 2016). Thus, the results of Manuscript 2 did not provide conclusive evidence on this matter. They could suggest that either the cultural or the decision-making context, or both, act as boundary conditions for these social class effects. Alternatively, they could point to a lack of replicability of the association altogether.

5.3 Manuscript 3: Social Class and Majority Influence in Product Choices

Given the inconclusive findings on the relationship between social class and conformity in Manuscript 2, Manuscript 3 refocused on consumer choices. Its main goal was to assess the replicability of the previously found negative relationship between social class and conformity in product choice scenarios. By conducting two pre-registered replication studies following the procedure employed by Na et al. (2016) in the German context and one high-powered replication study in the U.S. context (as the original research), Manuscript 3 additionally aimed at shedding light on the potential influence of the cultural context as a boundary condition. Furthermore, in response to shortcomings of the original research, we assessed the role of individuals' chronic interdependent self-construal as an underlying psychological mechanism.

Study 1a ($N_{1a} = 77$) constituted a first attempt to replicate the negative relationship between social class and conformity in product choices with a German sample, following the procedure of Na et al.'s (2016) Study 1. In this online study, participants viewed pictures of 60 product pairs and indicated their preferred option in each pair. Within each pair, products merely differed regarding minor design features. Manipulated feedback on social preferences followed each choice, indicating that either a majority or a minority of other participants had chosen the same product (consistent vs. conflicting feedback). In a second choice phase, participants again indicated their preferred option for each pair, relying on current feelings. Afterwards, SES was measured based on educational attainment and household income, alongside additional indicators such as maternal education. SSS was measured with the MacArthur Scale. The number of changes in choice served as the dependent variable. Results showed that people changed their product choices more often when deviating from a majority preference compared to being in line with a majority preference. However, in contrast to the findings of Na et al., we did not find any support for social class-based differences in conformity.

In Study 1b ($N_{1b} = 203$), we tested the generalizability of social class-based differences in conformity in the German context with a larger sample size and, thus, larger statistical power compared to Study 1a. Again, we followed the experimental design from Na et al.'s (2016) Study 1, but we additionally experimentally manipulated SSS before the product choice task to increase its variance and provide the basis for a causal interpretation of social class effects. Furthermore, Study 1b explored social class-based differences in perceived similarity with other participants as a potential mediator. As in Study 1a, participants made more changes in choices in conflicting compared to consistent feedback trials, but neither (manipulated) SSS nor (measured) SES moderated this effect. Exploratory analyses revealed a positive relationship between perceived similarity with other participants and conformity. However, we found no social class-based differences in perceived similarity. In sum, Study 1a/b did not find evidence in support of a negative relationship between social class and conformity in product choice in the German context. Importantly, these results could either point to the cultural context as a boundary condition for social class effects on conformity or to an overall limited replicability.

To provide a more complete picture regarding the role of the cultural context, Study 2 $(N_2 = 312)$ aimed to replicate the findings of Na et al. (2016) in the same national context as the original research - the U.S. context. We closely followed the experimental design of Na et al.'s (2016) Study 1. Still, we extended the study by assessing interdependent self-construal with the self-construal scale by Singelis (1994) to examine its potential role as a mediator. The manipulation of social preferences produced results consistent with the original study, with participants making more changes in product choice when the majority had ostensibly chosen a different compared to the same product. However, contrary to the results of the original research and different from Study 1a/b, both SSS and SES were positively associated with conformity, suggesting that U.S. participants with higher social class showed more conformity compared to participants with lower social class. Additionally, in contrast to theoretical accounts of social class (e.g., Kraus et al., 2012; Rucker et al., 2018), participants with higher SES indicated a more interdependent self-construal. For SSS, this relationship was non-significant. Additionally, interdependent self-construal was not significantly associated with conformity in product choices. Thus, we did not find evidence for its role as a mediator.

In summary, unlike Na et al. (2016), we found no evidence in three pre-registered replication studies that conformity is more pronounced among individuals from lower social classes compared to those from higher social classes in the context of product choice. Whereas effect sizes were small and non-significant in the German context (Studies 1a/b), a high-powered replication study in the U.S. context (Study 2) revealed a *positive* relationship – thus, in the contrary direction as the original research. Furthermore, the results did not support a mediation through an interdependent self-construal as suggested by Na et al. Consequently, the generalizability of social class-based differences in conformity in product choices is called into question.

5.4 Summary of the Main Findings

The key findings of the three manuscripts of this dissertation can be summarized as follows: The perceived competence of a politician increased the likelihood of voting for that candidate, but importantly, voters with higher SES valued competence relatively more than voters with lower SES (Manuscript 1). With regard to majority influence, the results were less clear (Manuscripts 2 and 3). While the effects of perceived social preferences on voting intentions were small and non-significant for most political parties, conformity emerged consistently in product choices. Importantly, the previously established moderation of majority influence by social class was not found in the context of voting intentions, nor was it

replicated for product choice in German samples. Unexpectedly, higher social class was even associated with greater conformity in product choices in U.S. samples.

Additionally, findings on social class-based differences in self-construal were mixed. On the one hand, individuals from higher social classes reported higher levels of selfperceived competence which appeared to mediate the relationship between social class and the importance of competence in politicians. This result aligns with the assumption that higher-class individuals are more agentic (e.g., Rucker et al., 2018). On the other hand, individuals from lower social classes did not report a higher self-perceived warmth or a more pronounced interdependent self-construal, seemingly contradicting the assumption of them being more communal (Kraus et al., 2012; Rucker et al., 2018) and casting further doubt on the replicability of class-based differences in conformity. In the *General Discussion*, I further discuss these findings on self-construal and their theoretical implications in light of new results from an additional follow-up project based on representative data from both the German and the U.S. context.

In summary, the three manuscripts underlying this dissertation shed light on the role of social class in the formation of voting intentions and product choices. Thereby, they addressed central shortcomings of prior research such as single measures of social class, a narrow focus on the U.S. context and low statistical power to provide a nuanced picture of social class effects. On the one hand, the present findings emphasize the association between social class and competence by showing how it is reflected in voting intentions. On the other hand, they call into question the generalizability and replicability of social class-based differences in aspects related to communion such as conformity and an interdependent self-construal.

6 General Discussion

While detailed discussions of the specific findings can be found in the respective manuscripts, this section shows broader implications for theoretical perspectives on social class, particularly regarding the relationship of social class to agency and communion. Additionally, I reflect on the limitations of the present empirical studies and point to questions that warrant investigation in future research. Finally, I outline practical implications of the present findings, particularly in the context of electoral and marketing campaigns targeting individuals across the social class spectrum.

6.1 Theoretical Implications for Social Cognitive Accounts of Social Class

Social cognitive accounts of social class are based on the central premise that higherclass individuals tend to be more oriented towards the self, respectively agentic, and lowerclass individuals tend to be more oriented towards others, respectively communal (Kraus et al., 2012; Rucker et al., 2018). By investigating social class-based differences in how individuals form voting preferences and make product choices, the three manuscripts of this dissertation provide some insights into the relationship between social class and agency, respectively, communion. In the following, I describe how the present findings extend the knowledge on social cognitive effects of social class but also challenge existing theoretical assumptions.

6.1.1 Social Class and Competence: Evidence for a Deeply Rooted Link

First of all, the present findings regarding the formation of voting intentions synthesize prior research on facial appearance effects in the political context (e.g., Todorov et al., 2005) and models of the psychological effects of social class (e.g., Kraus et al., 2012; Rucker et al., 2018). As expected, social class as measured via SES moderates facial appearance effects, such that higher-class voters value politicians' perceived competence to a larger degree than lower-class voters. Moreover, initial evidence suggests that higher-class voters perceive themselves as more competent and, importantly, perceive this trait in others in a more nuanced way than lower-class voters. Hence, it appears justified to extend prior models of social class by the notion that competence can act as a schematic trait (for an overview on self-schemata see Markus, 1977) for higher-class voters. As part of their chronically activated knowledge about their self, competence becomes a lens through which these individuals perceive and evaluate others (Fong & Markus, 1982).

Supporting this assumption, preliminary evidence suggests that perceived similarity may explain social class-based differences in the weighting of competence. Within their congruency model of voting behavior, Caprara and Zimbardo (2004) argue that voters tend to prefer politicians who they perceive as more similar to themselves and perceive their claims as more convincing and credible (Caprara et al., 2007). Thus, it is plausible to assume that higher-class voters value competence in politicians more as they perceive themselves as more competent.

Additionally, self-interest might serve as a second psychological mechanism underlying the relationship between social class and the importance of politicians' competence. As high perceived similarity is often interpreted as a signal that a politician shares important values with oneself (see Caprara & Zimbardo, 2004), higher-class voters might infer that politicians with a more competent appearance are more likely to legislate in accordance with their own interests compared to less competent-appearing politicians. Indeed, prior research found initial evidence for self-interest as a further mediator of social classbased differences in the weight placed on politicians' competence (Callaghan et al., 2022). Thus, it can be argued that the particularly pronounced preference for seemingly competent candidates among higher-class voters reflects a strategy to advance their own interests, which is a manifestation of agency itself.

Notably, social class is only positively related to the importance of politicians' competence and not to the importance of other agency-related traits such as assertiveness and dominance. Viewing the voting process as a matter of *conferral of social rank* (see Gobel & Miyamoto, 2024) can provide further insights into the nature of these relationships. Research in evolutionary psychology suggests that human social hierarchies are determined primarily by two strategies for increasing social rank (Cheng, 2020; Cheng et al., 2013; Maner & Case, 2016): On the one hand, social rank can be attained by expressing *dominance*, which refers to a strategy based on intimidation and coercion. On the other hand, individuals can increase their rank through *prestige*, a strategy that involves displaying and sharing knowledge and skills, which leads others to voluntarily subordinate themselves. In modern societies, prestige is the more widely endorsed route to rank conferral and is associated with a greater stability compared to hierarchies based on dominance which tend to produce higher levels of resistance (Cheng, 2020; see also Gobel & Miyamoto, 2024). Importantly, prestige and

UNKELBACH | SOCIAL CLASS

dominance are related to the broad dimension of agency. While both strategies reflect a high level of agency, prestige appears to be more closely linked to the sub-facet competence, whereas dominance rather refers to the sub-facet assertiveness (see e.g., Cheng et al., 2013). Based on this reasoning, the present research suggests that particularly voters with higher social class favor politicians who appear to endorse prestige as a strategy for achieving rank. The lack of a significant relationship between social class and the reported importance of politicians' assertiveness may be explained by its association with a dominance-based rank attainment strategy. Partially negative connotations of assertiveness, consistent with a forceful way of attaining rank, may counteract the effects of perceived similarity among higher-class voters. In accordance with this argumentation, looking highly dominant was not advantageous for politicians in the present research.

In sum, these findings offer important implications for widely assumed class-based differences in agency (see Rucker et al., 2018). While they emphasize the strength of the link between social class and competence in line with prior theoretical accounts, they also point at the need to differentiate the sub-facets of agency, competence and assertiveness, when investigating the role of social class in the formation of voting intentions.

Regarding communion, the results of additional analyses in Manuscript 1 suggest that the picture is less clear. Consistent with recent work by Boileau (2022), we found no evidence that lower-class individuals perceive themselves as more communal (as measured via warmth and trustworthiness). Similarly, they did not place a higher value on communion in politicians compared to higher-class individuals. While overarching models predict that lower-class individuals are more communal (Kraus et al., 2012; Rucker et al., 2018) – often formulated as a compensatory mechanism for having lower agency than higher-class individuals – the present findings suggest that this perspective is too simplified. In the following section, I will discuss this aspect in more detail in light of the findings on the role of social class in majority influence.

6.1.2 Social Class and Majority Influence: Rethinking Established Assumptions

While this dissertation establishes that social class acts as a moderator of politicians' perceived competence, the present findings raise doubts about its widely assumed role in majority influence. Contrary to previous research, Manuscripts 2 and 3 found that German individuals from lower social classes did not express a greater tendency to follow perceived majority preferences in either the political or consumer context. The results of prior research could also not be replicated in the U.S. context which rules out the cultural context as an alternative explanation for the null effects found with German samples. While these findings shed light on the role of social influence in individuals' decision-making, they challenge the fundamental assumption of social cognitive theories of social class which suggest a negative relationship between social class and an orientation towards others, i.e. communion (e.g., Kraus et al., 2012; Rucker et al., 2018).

But how can the present findings be explained? To answer this question, it might be necessary to rethink the nature of the relationship between social class and conformity. In prior research (see Na et al., 2016) as well as in the present studies, researchers followed the implicit assumption of an underlying linear relationship, suggesting that conformity increases with decreasing social class. However, the relationship can also take a different form (see Bless & Wänke, 2023). Considering how social class is distributed within societies, an

inverted U-shaped relationship between social class and conformity might also appear plausible. In Germany, merely a minority (7.7%) of individuals is rich (i.e., has an equivalence income above 200% of the population's median equivalence income) or is at risk of poverty (16.7%) (i.e., has an equivalence income below 60% of the population's median equivalence income) (Microcensus, 2022a, 2022b) while individuals in the middle range of incomes represent the majority of the population. Hence, individuals in the middle of the income range, and thus, the social class spectrum, may perceive the preference of a majority of others as more diagnostic due to their larger representation in general society (see also Moschis, 1976). As a consequence, conformity might be highest at medium levels of social class and lower at the ends of the social class spectrum – reflecting an inverted U-shaped relationship. A closely related so-called "middle-status conformity" hypothesis is well established in organizational and management research (e.g., Phillips & Zuckerman, 2001). At the individual level, research supporting this hypothesis found that organizational leaders with middle status make relatively more conventional choices compared to those with lower or higher status (Durand & Kremp, 2016).

An inverted U-shaped relationship between social class and conformity could explain the mixed findings from Manuscript 3 and previous research, given that the studies only covered certain subsections of the social class spectrum. Na et al. (2016) may have found a negative relationship between social class and conformity because their sample only included students from a selective public U.S. college, which is mainly attended by students from relatively wealthy families (Chetty et al., 2020). Therefore, their sample may have selectively covered middle to middle-high levels of social class⁴. In contrast, the samples in Manuscript 3 were likely more diverse in terms of social class as they were not limited to students and were collected via an online crowdsourcing platform (Study 1a and 2). However, there was a lack of participants with extremely low or high social class in all of our studies. Collecting data from individuals with predominantly lower to middle levels of social class may have resulted in the relationship with conformity being nonsignificant (Study 1a/b) or even positive (Study 2). Exploratory tests of a quadratic trend yielded inconsistent results in both the German samples and the U.S. sample in Manuscript 3. However, these should only be interpreted with caution due to the lack of representativeness regarding social class. For future research, it may be fruitful to consider the possibility of a nonlinear relationship between social class and conformity in order to resolve the mixed findings. This requires assessing adequately powered samples that cover the entire social class spectrum within each cultural context of interest.

Finally, although not the primary focus of Manuscript 3, it provides some insights into the commonly assumed negative relationship between social class and an interdependent selfconstrual, the assumed mediator of class-based differences in conformity. In contrast to the prediction of social cognitive models of social class, SES was significantly *positively* related to an interdependent self-construal. Thereby, the present findings contribute to the mixed empirical evidence on this matter (Fernández et al., 2005; Grossmann & Varnum, 2011; Stephens et al., 2007). In a large-scale replication project together with co-authors (Batruch et al., 2023), using a high-powered German sample (n = 1,324) and a representative U.S. sample (n = 5,771), we were unable to replicate the negative relationship between social class and an

⁴ In Na et al.'s (2016) Study 1, no further descriptive statistics were reported for measures of SES, except for the number of students classified as working or middle class based on maternal education.

interdependent self-construal again⁵. In both samples, preliminary findings show that correlations between the different measures of an interdependent self-construal and social class indicators were either non-significant or positive.

In conclusion, together with the insignificant relationship between social class and self-perceived communion found in Manuscript 1, these inconsistent findings imply that the nature of the relationship between social class and communion is less robust than expected – and potentially of a fundamentally different nature – than the relationship between social class and agency and, thus, requires further research.

6.2 Limitations and Directions for Future Research

The present dissertation sheds light on how social class shapes the influence of a) politicians' perceived competence and b) social preferences on voting preferences respectively product choices. Yet, the underlying empirical studies have some limitations which point to open questions and fruitful approaches for future research.

First, although all manuscripts captured social class as a multifaceted construct by using two measures of SES, educational attainment and household income (employment status as a proxy for income level in Manuscript 2), as well as one measure of SSS, the current approach still has some drawbacks. With respect to SES, we followed previous research by using both indicators to form a joint score (e.g., Callaghan et al., 2022; Kraus & Keltner, 2009). However, this approach assumes continuity of the variables, which is not guaranteed. Additionally, it neglects that educational attainment and income shape different aspects of social class contexts and may, thus, affect outcomes differently. For example, with respect to political orientation, both indicators showed divergent correlations, with higher income being associated with a more right-wing and higher educational attainment being related with a more left-wing political orientation (Manuscript 1). For this reason, the analyses for all manuscripts were reported separately by indicator type in the respective appendix. Apart from this aspect, a limitation of the present research is the lack of assessment of occupational status as a third indicator of SES. Given that perceived similarity mediates classbased differences in the importance of politicians' competence (as holders of highly prestigious offices) (Manuscript 1), one might expect even stronger associations for occupational status than for other SES indicators. Generally, future research on psychological effects of social class should investigate whether social class effects that were previously demonstrated with only a single indicator of the construct are generalizable to other components of social class contexts (see also Antonoplis, 2023).

Furthermore, the present dissertation highlights the need to clarify the nature of the subjective components of social class. In Manuscript 1, voters with higher SSS rated all facets of agency and communion as more important in politicians than voters with lower SSS. However, in contrast to SES, SSS did not moderate the effect of perceived competence on the likelihood of voting for the respective candidate. One possible explanation for this divergent pattern for SES and SSS could be the artificial inflation of correlations of SSS with other variables measured through self-report due to common method variance (see Tan et al., 2020).

⁵ Interdependent self-construal was assessed via the Singelis scale (1994) in the German sample and via the self-construal scale by Vignoles et al. (2016), need for inclusion Valcke et al. (2020) and collectivism (Yoo et al., 2011) in the U.S. sample.

Thus, it seems fruitful for future research on SSS to follow the approach taken in Manuscript 1 and assess outcomes using both self-report and other types of measures to gain a deeper understanding of the robustness and the processes underlying the effects of SSS.

Second, although the manuscripts underlying this dissertation went beyond previous research in this field by investigating social class effects in the German instead of solely the U.S. context, important questions regarding their cross-cultural generalizability remain open. Prior research suggests that the fundamental class-based differences in agentic and communal self-construal which are expected to underly the investigated social class effects are influenced by two characteristics of the investigated context: the level of economic inequality and cultural norms (see e.g., Gobel & Miyamoto, 2024). Although Germany is characterized by a considerably lower economic inequality compared to the U.S.A., it is also a Western, highly individualistic country (Hofstede et al., 2010). Recent accounts trying to integrate cross-cultural findings on social cognitive effects of social class primarily focused on comparing Western and East Asian countries (Gobel & Miyamoto, 2024; Miyamoto, 2017). However, this categorization might not be sufficiently fine-grained, considering that Manuscript 3 reports initial evidence on differences in the relationship between SES and conformity between the German and the U.S. context. To gain a more nuanced understanding of the generalizability of class-based differences in the formation of political preferences and consumer choices, future research should adopt a broader cross-cultural perspective. Economic inequality, culture, as well as their interplay should be investigated as potential boundary conditions with samples from multiple countries.

Third, the present studies have measured social class cross-sectionally, limiting the ability to model fluctuations in social class over the lifespan and their potential effects on individuals' psychological makeup (see e.g., Kraus & Stephens, 2012). Additionally, the present research did not establish causal evidence for social class effects. Previous research has proposed experimental manipulations of SSS to establish causality (Brown-Iannuzzi et al., 2015; Kraus et al., 2015). However, this approach did not seem sufficient for the present research as I expected stable class-based differences in the self-construal as underlying mechanisms. These differences are likely to develop over a longer period of time when living in certain social class contexts and, therefore, cannot be adequately manipulated experimentally. To address both limitations, future research would benefit from collecting nationally representative longitudinal data with multiple measurement points over several years.

Fourth, this dissertation focuses specifically on the role of social class in individuals' decision making while treating other social categories as control variables. However, personal characteristics such as gender and ethnicity can also serve as status-based contexts (Kraus & Stephens, 2012) and influence fundamental differences in agency and communion (Rucker et al., 2018). For instance, preliminary evidence suggests that gender moderates cultural differences in independent and interdependent self-construal (Cai et al., 2022). Previous accounts have argued that psychological orientations are determined by the most salient form of social hierarchy in a given situation (Rucker et al., 2018). Nevertheless, an intersectional perspective which considers how social class-based differences are intertwined with the effects of categories such as gender and ethnicity may provide a more comprehensive understanding. Future research could put a focus on actively examining intersectional

differences in the formation of voting preferences or product choices. Additionally, research on facial appearance effects should examine the generalizability of the present findings by using a more diverse set of stimuli regarding these characteristics.

6.3 **Practical Implications**

6.3.1 Competence in Political Campaigns: Targeting Higher-Class Voters

To gain a deeper understanding of the real-life consequences of the present findings for the political context, it is important to consider that in many modern democracies, individuals with higher SES not only represent the majority of members of parliament (e.g., for Germany; see Deutscher Bundestag, 2023), but also tend to exhibit a higher level of political engagement (e.g., Kraus et al., 2015; Manuscript 1). Therefore, the previously found positive effect of politicians' perceived competence on electoral success (Ballew & Todorov, 2007; Todorov, 2005) may be partially explained by higher turnout rates of higher-class voters in actual election scenarios.

For political actors, the present findings suggest that they can benefit from taking into account the social class of their target group of voters when deciding whether to use competence-related claims in political campaigns. Although voters across the social class spectrum generally value competence, competence-related political communication is likely to be more effective for a higher-class target group. Furthermore, it appears worthwhile to consider the role of perceived candidate traits not only from verbal but also non-verbal forms of communication such as portraits on advertising posters (e.g., Olivola & Todorov, 2010a; for prestige, see Witkower et al., 2020). However, addressing lower-class voters appears to be more challenging. Contrary to common compensatory views on agency and communion, lower-class voters do not generally value warmth or trustworthiness more than higher-class voters (Manuscript 1) but merely do so under certain conditions such as a heightened perception of sincerity (Tan & Kraus, 2018).

However, it should be noted that establishing a political context where politicians rely primarily on competence claims may carry the risk of perpetuating unequal levels of political participation. Particularly lower-class voters may feel less addressed, which could further reduce their political engagement. However, further research is needed to confirm this hypothesis. In an unpublished study (N = 260) (Pschibilski, 2022), German voters perceived political parties that used slogans expressing high competence (vs. slogans expressing high warmth) as being more likely to target individuals with higher income, education, and occupational prestige. This preliminary finding may reflect individuals' awareness of which social groups are (or are not) addressed by political communication that emphasizes competence.

6.3.2 Context Matters for Majority Influence: Poll Effects and Product Claims

Although no class-based differences in poll effects were found, the present findings contribute to the ongoing public debate on the conditions for publishing pre-election poll results. Based on the present research, the likelihood of public opinion becoming a self-fulfilling prophecy as described in the political bandwagon effect (Schmitt-Beck, 2015) appears to be rather small in multiparty contexts. Fluctuating poll results seem to play a much smaller role for voting intentions than voters' identification with political parties and the evaluation of leading party candidates (potentially influenced by the evaluation of candidate

traits as outlined above). However, it is possible that the picture looks different in two-party systems in which a 'winner' in polls is more clearly defined (Meffert et al., 2011). This may also be the case in situations with high uncertainty in voting preferences, such as time periods with long temporal distance to an election.

Regarding ambiguous product choices, the present findings underscore the importance of majority influence. When consumers lack a clear preference, information about the preference of a majority can sway their decision and even prompt them to reconsider their initial choice. Therefore, the use of popularity claims remains an important tool for marketers to influence consumer preferences (see Cialdini, 1993). However, it is not yet possible to draw profound practical implications regarding the potential benefits of using such claims to target individuals from certain social classes due to the mixed empirical evidence.

Finally, for voters and consumers across the social class spectrum, reflecting on the factors influencing their choices and the validity for their actual political preferences or consumption goals might not only increase perceived control over decisions (Caprara et al., 2004), but also improve (political) information processing and, consequently, decision quality.

6.4 Conclusion

Recent research on social class has demonstrated the encompassing effects of social class on psychological tendencies. Based on social cognitive models of social class, the present dissertation takes a closer look at two specific social class effects in the political and the consumer context, respectively. Social class was investigated as a moderator of a) the effects of politicians' perceived competence on voting intentions and b) the size of majority influence in voting intentions and product choices. The empirical findings of the three underlying manuscripts provide a mixed picture. On the one hand they reveal class-based differences in the preference for competence in politicians which is in line with the well-established positive relationship between social class and agentic traits. On the other hand, they do not support the notion of a negative relationship between social class and conformity. Both in the political and the consumer context this relationship seems to be more limited in generalizability than suggested by prior research. In sum, the present findings imply that rethinking the linear assumption of lower-class individuals being more communal than higher-class individuals is crucial to advance psychological models of social class.

In conclusion, this dissertation addresses the discrepancy between the extent of what social psychology currently understands about fundamental social class-based differences in psychological tendencies and their importance in the face of increasing social disparities worldwide. By narrowing the research gap on how social class shapes the formation of voting intentions and consumption choices, the present research contributes to identifying the mechanisms that shape individuals' living contexts and potentially perpetuate existing social inequalities. Not only does it provide information for political actors on how to improve political communication, but it also constitutes one step in helping voters and consumers to make well-informed, reflected, and thus potentially better, choices – across the social class spectrum.

7 References

- Abele, A. E. (2003). The dynamics of masculine-agentic and feminine-communal traits: Findings from a prospective study. *Journal of Personality and Social Psychology*, 85(4), 768–776. https://doi.org/10.1037/0022-3514.85.4.768
- Abele, A. E., Hauke, N., Peters, K., Louvet, E., Szymkow, A., & Duan, Y. (2016). Facets of the fundamental content dimensions: Agency with competence and assertivenesscommunion with warmth and morality. *Frontiers in Psychology*, 7, 1810. https://doi.org/10.3389/fpsyg.2016.01810
- Abele, A. E., & Wojciszke, B. (2007). Agency and communion from the perspective of self versus others. *Journal of Personality and Social Psychology*, 93(5), 751–763. https://doi.org/10.1037/0022-3514.93.5.751
- Abele, A., & Wojciszke, B. (2014). Communal and agentic content in social cognition: A dual perspective model. *Advances in Experimental Social Psychology*, 50, 195-255. https://doi.org/10.1016/B978-0-12-800284-1.00004-7
- Adler, N. E., Epel, E. S., Castellazzo, G., & Ickovics, J. R. (2000). Relationship of subjective and objective social status with psychological and physiological functioning: Preliminary data in healthy white women. *Health Psychology*, *19*(6), 586–592. https://doi.org/10.1037/0278-6133.19.6.586
- Amir, D., Jordan, M. R. & Rand, D. G. (2018). An uncertainty management perspective on long-run impacts of adversity: The influence of childhood socioeconomic status on risk, time, and social preferences. *Journal of Experimental Social Psychology*, 79, 217–226. https://doi.org/10.1016/j.jesp.2018.07.014
- Antonakis, J., & Dalgas, O. (2009). Predicting elections: Child's play! *Science*, *323*(5918), 1183. https://doi.org/10.1126/science.1167748
- Antonoplis, S. (2023). Studying socioeconomic status: Conceptual problems and an alternative path forward. *Perspectives on Psychological Science*, 18(2), 275–292. https://doi.org/10.1177/17456916221093615
- Asch, S. E. (1956). Studies of independence and conformity: I. A minority of one against a unanimous majority. *Psychological Monographs: General and Applied*, *70*(9), 1–70. https://doi.org/10.1037/h0093718
- Assari, S. (2017). Social determinants of depression: The intersections of race, gender, and socioeconomic status. *Brain Sciences*, 7(12). https://doi.org/10.3390/brainsci7120156
- Bakan, D. (1966). *The duality of human existence: Isolation and communion in Western man.* Beacon Press.
- Ballew, C. C., & Todorov, A. (2007). Predicting political elections from rapid and unreflective face judgments. *Proceedings of the National Academy of Sciences*, 104(46), 17948–17953. https://doi.org/10.1073/pnas.0705435104
- Barnfield, M. (2020). Think twice before jumping on the bandwagon: Clarifying concepts in research on the bandwagon effect. *Political Studies Review*, 18(4), 553–574. https://doi.org/10.1177/1478929919870691
- Batruch, A., Unkelbach, F., Prisacaru, N., & Wänke, M. (2023). Social class and interdependent self-construal: Two large-scale studies from the U.S. and the German context. Manuscript in preparation.

- Belmi, P., Neale, M. A., Reiff, D., & Ulfe, R. (2020). The social advantage of miscalibrated individuals: The relationship between social class and overconfidence and its implications for class-based inequality. *Journal of Personality and Social Psychology*, *118*(2), 254–282. https://doi.org/10.1037/pspi0000187
- Bless, H., & Wänke, M. (2023). The crucial role of linearity when comparing effects across studies. *Nature Reviews Psychology*, 2, 516–517. https://doi.org/10.1038/s44159-023-00220-w
- Boileau L.-A. (2022). Social class and the big two: Investigating a widespread yet understudied assumption on social class differences in self-concepts. [Doctoral dissertation, University of Mannheim].
- Bourdieu, P. (1984). *Distinction: A social critique of the judgment of taste* (R. Nice, Trans.). Harvard University Press. (Original work published 1979)
- Brown-Iannuzzi, J. L., Lundberg, K. B., Kay, A. C., & Payne, B. K. (2015). Subjective status shapes political preferences. *Psychological Science*, *26*(1), 15–26. https://doi.org/10.1177/0956797614553947
- Button, K. S., Ioannidis, J. P., Mokrysz, C., Nosek, B. A., Flint, J., Robinson, E. S., & Munafò, M. R. (2013). Power failure: Why small sample size undermines the reliability of neuroscience. *Nature Reviews Neuroscience*, 14, 365–376. https://doi.org/10.1038/nrn3475
- Cai, N., Dang, S., & Hewitt, P. (2022, September 3). Independent and interdependent selfconstrual: An examination of migration history and gender in Chinese- and European-Canadians. PsyArXiv. https://doi.org/10.31234/osf.io/47hfj
- Callaghan, B., Kraus, M. W., & Dovidio, J. F. (2022). Social class predicts preference for competent politicians. *Journal of Experimental Social Psychology*, 100, 104298. https://doi.org/10.1016/j.jesp.2022.104298
- Caprara, G. V., Vecchione, M., Barbaranelli, C., & Fraley, R. C. (2007). When likeness goes with liking: the case of political preference. *Political Psychology*, *28*(5), 609–632. https://doi.org/10.1111/j.1467-9221.2007.00592.x
- Caprara, G. V., & Zimbardo, P. G. (2004). Personalizing politics: A congruency model of political preference. *American Psychologist*, 59(7), 581–594. https://doi.org/10.1037/0003-066X.59.7.581
- Carey, R. M., & Markus, H. R. (2017). Social class shapes the form and function of relationships and selves. *Current Opinion in Psychology*, 18, 123–130. https://doi.org/10.1016/j.copsyc.2017.08.031
- Cheng, J. T. (2020). Dominance, prestige, and the role of leveling in human social hierarchy and equality. *Current Opinion in Psychology*, *33*, 238–244. https://doi.org/10.1016/j.copsyc.2019.10.004
- Cheng, J. T., Tracy, J. L., Foulsham, T., Kingstone, A., & Henrich, J. (2013). Two ways to the top: Evidence that dominance and prestige are distinct yet viable avenues to social rank and influence. *Journal of Personality and Social Psychology*, 104(1), 103–125. https://doi.org/10.1037/a0030398
- Chetty, R., Friedman, J. F., Saez, E., Turner, N., & Yagan, D. (2020). Baseline crosssectional estimates of child and parent income distributions by college [Data set]. Retrieved from Opportunity Insights

https://opportunityinsights.org/data/?geographic_level=100&topic=0&paper_id=3084 #resource-listing

- Cialdini, R. B. (1993). *Influence: Science and practice* (3rd ed.). HarperCollins College Publishers.
- Connor, P., Varney, J., Keltner, D., & Chen, S. (2021). Social class competence stereotypes are amplified by socially signaled economic inequality. *Personality & Social Psychology Bulletin*, 47(1), 89–105. https://doi.org/10.1177/0146167220916640
- Department of Economic and Social Affairs of the United Nations (2016). *Inequality a defining challenge of our time*. United Nations. https://www.un.org/en/desa/inequality-%E2%80%93-defining-challenge-our-time
- Department of Economic and Social Affairs of the United Nations (2020). *World social report 2020: Inequality in a rapidly changing world.* https://www.un.org/development/desa/dspd/wpcontent/uploads/sites/22/2020/02/World-Social-Report2020-FullReport.pdf
- Deutscher Bundestag (2023). Das Datenhandbuch zur Geschichte des Deutschen Bundestages. Alters- und Sozialstruktur. Schul- und Hochschulbildung [Data set]. Retrieved from https://www.bundestag.de/resource/blob/272942/6cf360e6389ed7280699f6667e43c5e 5/Kapitel_03_09_Schul-_und_Hochschulbildung-pdf.pdf
- Durand, R., & Kremp, P.-A. (2016). Classical deviation: Organizational and individual status as antecedents of conformity. *Academy of Management Journal*, *59*(1), 65–89. https://doi.org/10.5465/amj.2013.0767
- Durante, F., Tablante, C. B., & Fiske, S. T. (2017). Poor but warm, rich but cold (and competent): Social classes in the stereotype content model. *Journal of Social Issues*, 73(1), 138–157. https://doi.org/10.1111/josi.12208
- Faas, T., Mackenrodt, C., & Schmitt-Beck, R. (2008). Polls that mattered: Effects of media polls on voters' coalition expectations and party preferences in the 2005 German parliamentary election. *International Journal of Public Opinion Research*, 20(3), 299– 325. https://doi.org/10.1093/ijpor/edn034
- Fein, O. (1995). The influence of social class on health status: American and British research on health inequalities. *Journal of General Internal Medicine*, 10(10), 577–586. https://doi.org/10.1007/BF02640369
- Fernández, I., Páez, D., & González, J. L (2005). Independent and interdependent selfconstruals and socio-cultural factors in 29 nations. *Revue Internationale de Psychologie Sociale*, 18(1-2), 35-63.
- Fiske, S. T., & Markus, H. R. (2012). *Facing social class: How societal rank influences interaction*. Russell Sage Foundation.
- Fong, G. T., & Markus, H. (1982). Self-schemas and judgments about others. *Social Cognition*, 1(3), 191–204. https://doi.org/10.1521/soco.1982.1.3.191
- Funk, C. L. (1999). Bringing the candidate into models of candidate evaluation. *The Journal of Politics*, 61(3), 700–720. https://doi.org/10.2307/2647824
- Gebauer, J. E., Paulhus, D. L., & Neberich, W. (2013). Big two personality and religiosity across cultures. *Social Psychological and Personality Science*, 4(1), 21–30. https://doi.org/10.1177/1948550612442553

- GLES (2022). GLES Rolling Cross-Section 2021. GESIS, Köln. ZA7703 Datenfile Version 2.0.0. https://doi.org/10.4232/1.13876
- Gobel, M. S., & Miyamoto, Y. (2024). Self- and other-orientation in high rank: A cultural psychological approach to social hierarchy. *Personality and Social Psychology Review*, 28(1), 54-80. https://doi.org/10.1177/10888683231172252
- Grossmann, I., & Varnum, M. E. W. (2011). Social class, culture, and cognition. Social Psychological and Personality Science, 2(1), 81–89. https://doi.org/10.1177/1948550610377119
- Hoffmann, H., & Klein, M. (2013). Koalitionen, Kandidaten, Kommunikation: Analysen zur Bundestagswahl 2009 (Schriftenreihe des Arbeitskreises "Wahlen und politische Einstellungen" der Deutschen Vereinigung für Politische Wissenschaften (DVPW)). Springer VS.
- Hofstede G., Hofstede G. J., & Minkov M. (2010). *Cultures and organizations—Software of the mind: Intercultural cooperation and its importance for survival* (3rd ed.). McGraw-Hill.
- Hogan, R. (1982). A socioanalytic theory of personality. In M. Page (Ed.), *Nebraska symposium on motivation* (Vol. 30, pp. 55–89). Lincoln: University of Nebraska Press.
- Kan, C., Kawakami, N., Karasawa, M., Love, G. D., Coe, C. L., Miyamoto, Y., Ryff, C. D., Kitayama, S., Curhan, K. B., & Markus, H. R. (2014). Psychological resources as mediators of the association between social class and health: Comparative findings from Japan and the USA. *International Journal of Behavioral Medicine*, 21(1), 53–65. https://doi.org/10.1007/s12529-012-9249-y
- Kinder, D. R., Peters, M. D., Abelson, R. P., & Fiske, S. T. (1980). Presidential prototypes. *Political Behavior*, 2(4), 315–337. https://doi.org/10.1007/BF00990172
- Kohn, M. L. (1969). Class and conformity: A study in values. Dorsey.
- Kraus, M., Anderson, C., & Callaghan, B. (2015). The inequality of politics: Social class rank and political participation. SSRN Electronic Journal. https://doi.org/10.2139/ssrn.2600107
- Kraus, M. W., Côté, S., & Keltner, D. (2010). Social class, contextualism, and empathic accuracy. *Psychological Science*, 21(11), 1716-1723. https://doi.org/10.1177/0956797610387613
- Kraus, M. W., & Keltner, D. (2009). Signs of socioeconomic status: A thin-slicing approach. *Psychological Science*, 20(1), 99–106. https://doi.org/10.1111/j.1467-9280.2008.02251.x
- Kraus, M. W., Piff, P. K., & Keltner, D. (2009). Social class, sense of control, and social explanation. *Journal of Personality and Social Psychology*, 97(6), 992–1004. https://doi.org/10.1037/a0016357
- Kraus, M. W., Piff, P. K., Mendoza-Denton, R., Rheinschmidt, M. L., & Keltner, D. (2012). Social class, solipsism, and contextualism: How the rich are different from the poor. *Psychological Review*, 119(3), 546–572. https://doi.org/10.1037/a0028756
- Kraus, M. W., & Stephens, N. M. (2012). A road map for an emerging psychology of social class. Social and Personality Psychology Compass, 6(9), 642–656. https://doi.org/10.1111/j.1751-9004.2012.00453.x

- Kraus, M. W., Tan, J. J. X., & Tannenbaum, M. B. (2013). The social ladder: A rank-based perspective on social class. *Psychological Inquiry*, 24(2), 81–96. https://doi.org/10.1080/1047840X.2013.778803
- Kurt, D., & Frimer, J. (2015). Agency and communion as a framework to understand consumer behavior. In M. I. Norton, D. D. Rucker, & C. Lamberton (Eds.), *The Cambridge Handbook of Consumer Psychology* (pp. 446–475). Cambridge University Press. https://doi.org/10.1017/CBO9781107706552.017
- John, M., Boileau, L. L.-A., & Bless, H. (2023). Effect of social class on personal control beliefs. *Journal of Personality*, 00, 1-14. https://doi.org/10.1111/jopy.12872
- Johnson, W., & Krueger, R. F. (2006). How money buys happiness: Genetic and environmental processes linking finances and life satisfaction. *Journal of Personality* and Social Psychology, 90(4), 680–691. https://doi.org/10.1037/0022-3514.90.4.680
- Lachman, M. E., & Weaver, S. L. (1998). The sense of control as a moderator of social class differences in health and well-being. *Journal of Personality and Social Psychology*, 74(3), 763–773. https://doi.org/10.1037/0022-3514.74.3.763
- Laustsen, L., & Bor, A. (2017). The relative weight of character traits in political candidate evaluations: Warmth is more important than competence, leadership and integrity. *Electoral Studies*, 49, 96–107. https://doi.org/10.1016/j.electstud.2017.08.001
- Li, W., Yang, Y., Wu, J., & Kou, Y. (2020). Testing the status-legitimacy hypothesis in China: Objective and subjective socioeconomic status divergently predict system justification. *Personality and Social Psychology Bulletin*, 46(7), 1044–1058. https://doi.org/10.1177/0146167219893997
- Maner, J. K., & Case, C. R. (2016). Dominance and prestige: Dual strategies for navigating social hierarchies. *Advances in Experimental Social Psychology*, 54, 129–180. https://doi.org/10.1016/bs.aesp.2016.02.001
- Manstead, A. S. R. (2018). The psychology of social class: How socioeconomic status impacts thought, feelings, and behaviour. *British Journal of Social Psychology*, 57(2), 267–291. https://doi.org/10.1111/bjso.12251
- Markus, H. (1977). Self-schemata and processing information about the self. *Journal of Personality and Social Psychology*, *35*(2), 63–78. https://doi.org/10.1037/0022-3514.35.2.63
- Markus, H. R., & Kitayama, S. (1991). Culture and the self: Implications for cognition, emotion, and motivation. *Psychological Review*, 98(2), 224–253. https://doi.org/10.1037/0033-295X.98.2.224
- Marx, K., & Engels, F. (2017). *The communist manifesto* (S. Moore, Trans.). Pluto Press. (Original work published 1848)
- McLeod, C. B., Hall, P. A., Siddiqi, A., & Hertzman, C. (2012). How society shapes the health gradient: Work-related health inequalities in a comparative perspective. *Annual Review of Public Health*, 33, 59–73. https://doi.org/10.1146/annurev-publhealth-031811-124603
- Meffert, M. F., Huber, S., Gschwend, T., & Pappi, F. U. (2011). More than wishful thinking: Causes and consequences of voters' electoral expectations about parties and coalitions. *Electoral Studies*, *30*(4), 804–815. https://doi.org/10.1016/j.electstud.2011.08.001
- Microcensus (2022a). Armutsgefährdungsquoten, Bundesländer (Bundesmedian, Landesmedian) ab 2020 [Data set]. Retrieved from

https://www.statistikportal.de/de/sbe/ergebnisse/einkommen-armutsgefaehrdung-und-soziale-lebensbedingungen/armutsgefaehrdung-und-0

- Microcensus (2022b). Einkommensreichtumsquoten, Bundesländer (Bundesmedian, Landesmedian) ab 2020 [Data set]. Retrieved from https://www.statistikportal.de/de/sbe/ergebnisse/einkommen-armutsgefaehrdung-undsoziale-lebensbedingungen/armutsgefaehrdung-und-1
- Miller, A. H., Wattenberg, M. P., & Malanchuk, O. (1986). Schematic assessments of presidential candidates. *American Political Science Review*, 80(2), 521–540. https://doi.org/10.2307/1958272
- Miyamoto, Y. (2017). Culture and social class. *Current Opinion in Psychology, 18,* 67–72. https://doi.org/10.1016/j.copsyc.2017.07.042
- Miyamoto, Y., Yoo, J., Levine, C. S., Park, J., Boylan, J. M., Sims, T., Markus, H. R., Kitayama, S., Kawakami, N., Karasawa, M., Coe, C. L., Love, G. D., & Ryff, C. D. (2018). Culture and social hierarchy: Self- and other-oriented correlates of socioeconomic status across cultures. *Journal of Personality and Social Psychology*, *115*(3), 427–445. https://doi.org/10.1037/pspi0000133
- Moschis, G. P. (1976). Social comparison and informal group influence. *Journal of Marketing Research*, 13(3), 237-244. https://doi.org/10.1177/002224377601300304
- Moy, P., & Rinke, E. M. (2012). *Opinion polls and the media: Reflecting and shaping public opinion (Opinion Polls and the Media)*. Palgrave Macmillan.
- Mutz, D. C. (1998). Impersonal influence in American politics. Cambridge studies in political psychology and public opinion. Cambridge University Press. https://doi.org/10.1017/CBO9781139175074
- Na, J., Grossmann, I., Varnum, M. E. W., Kitayama, S., Gonzalez, R., & Nisbett, R. E. (2010). Cultural differences are not always reducible to individual differences. *Proceedings of the National Academy of Sciences of the United States of America*, 107(14), 6192–6197. https://doi.org/10.1073/pnas.10019 11107
- Na, J., McDonough, I. M., Chan, M. Y., & Park, D. C. (2016). Social-class differences in consumer choices: Working-class individuals are more sensitive to choices of others than middle-class individuals. *Personality & Social Psychology Bulletin*, 42(4), 430– 443. https://doi.org/10.1177/0146167216634043
- Oakes, J. M., & Rossi, P. H. (2003). The measurement of SES in health research: Current practice and steps toward a new approach. *Social Science & Medicine*, *56*(4), 769–784. https://doi.org/10.1016/S0277-9536(02)00073-4
- OECD Centre for Opportunity and Equality (2017). Understanding the socio-economic divide in Europe (Background Report). https://www.oecd.org/els/soc/cope-divide-europe-2017-background-report.pdf
- Oldmeadow, J. A., & Fiske, S. T. (2010). Social status and the pursuit of positive social identity: Systematic domains of intergroup differentiation and discrimination for highand low- status groups. *Group Processes & Intergroup Relations: GPIR, 13*(4), 425– 444. https://doi.org/10.1177/1368430209355650
- Olivola, C. Y., & Todorov, A. (2010a). Elected in 100 milliseconds: Appearance-based trait inferences and voting. *Journal of Nonverbal Behavior*, *34*(2), 83–110. https://doi.org/10.1007/s10919-009-0082-1

- Olivola, C. Y., & Todorov, A. (2010b). Fooled by first impressions? Reexamining the diagnostic value of appearance-based inferences. *Journal of Experimental Social Psychology*, 46(2), 315–324. https://doi.org/10.1016/j.jesp.2009.12.002
- Phillips, D. J., & Zuckerman, E. W. (2001). Middle-status conformity: Theoretical restatement and empirical demonstration in two markets. *American Journal of Sociology*, 107(2), 379–429. https://doi.org/10.1086/324072
- Piff, P. K., & Moskowitz, J. P. (2018). Wealth, poverty, and happiness: Social class is differentially associated with positive emotions. *Emotion*, 18(6), 902–905. https://doi.org/10.1037/emo0000387
- Piff, P. K., Stancato, D. M., Martinez, A. G., Kraus, M. W., & Keltner, D. (2012). Class, chaos, and the construction of community. *Journal of Personality and Social Psychology*, 103(6), 949–962. https://doi.org/10.1037/a0029673
- Pirlott, A. G., & MacKinnon, D. P. (2016). Design approaches to experimental mediation. Journal of Experimental Social Psychology, 66, 29–38. https://doi.org/10.1016/j.jesp.2015.09.012
- Pschibilski (2022). Wärme und Kompetenz bei der Wahrnehmung von Parteien
 Die Rolle des sozialen Status der Wählenden [Unpublished Bachelor's thesis].
 University of Mannheim.
- Rucker, D. D., Galinsky, A. D., & Magee, J. C. (2018). The agentic–communal model of advantage and disadvantage: How inequality produces similarities in the psychology of power, social class, gender, and race. *Advances in Experimental Social Psychology*, 58, 71–125. https://doi.org/10.1016/bs.aesp.2018.04.001
- Schmitt-Beck, R. (1996). Mass media, the electorate, and the bandwagon. A study of communication effects on vote choice in Germany. *International Journal of Public Opinion Research*, 8(3), 266–291. https://doi.org/10.1093/ijpor/8.3.266
- Schmitt-Beck, R. (2015). Bandwagon effect. In G. Mazzoleni (Ed.), *The International Encyclopedia of Political Communication* (pp. 1–5). Wiley. https://doi.org/10.1002/9781118541555.wbiepc015
- Schneider, S. M. (2019). Why income inequality is dissatisfying—perceptions of social status and the inequality-satisfaction link in Europe. *European Sociological Review*, *35*(3), 409–430. https://doi.org/10.1093/esr/jcz003
- Sheehy-Skeffington, J. (2020). The effects of low socioeconomic status on decision-making processes. *Current Opinion in Psychology*, 33, 183–188. https://doi.org/10.1016/j.copsyc.2019.07.043
- Singelis, T. M. (1994). The measurement of independent and interdependent self-construals. *Personality and Social Psychology Bulletin, 20*(5), 580–591. https://doi.org/10.1177/0146167294205014
- Singh-Manoux, A., Adler, N. E., & Marmot, M. G. (2003). Subjective social status: Its determinants and its association with measures of ill-health in the Whitehall II study. *Social Science and Medicine*, 56(6), 1321-1333. https://doi.org/10.1016/S0277-9536(02)00131-4
- Singh-Manoux, A., Marmot, M. G., & Adler, N. E. (2005). Does subjective social status predict health and change in health status better than objective status? *Psychosomatic Medicine*, 67(6), 855–861. https://doi.org/10.1097/01.psy.0000188434.52941.a0

- Snibbe, A. C., & Markus, H. R. (2005). You can't always get what you want: Educational attainment, agency, and choice. *Journal of Personality and Social Psychology*, 88(4), 703–720. https://doi.org/10.1037/0022-3514.88.4.703
- Solt, F. (2019). The standardized world income inequality database, Versions 8-9. *Harvard Dataverse, V11*. https://doi.org/10.7910/DVN/LM4OWF
- Stephens, N. M., Fryberg, S. A., & Markus, H. R. (2011). When choice does not equal freedom. *Social Psychological and Personality Science*, 2(1), 33–41. https://doi.org/10.1177/1948550610378757
- Stephens, N. M., Markus, H. R., & Phillips, L. T. (2014). Social class culture cycles: How three gateway contexts shape selves and fuel inequality. *Annual Review of Psychology*, 65(1), 611–634. https://doi.org/10.1146/annurev-psych-010213-115143
- Stephens, N. M., Markus, H. R., & Townsend, S. S. M. (2007). Choice as an act of meaning: The case of social class. *Journal of Personality and Social Psychology*, 93(5), 814– 830. https://doi.org/10.1037/0022-3514.93.5.814
- Stringhini, S., Carmeli, C., Jokela, M., Avendaño, M., Muennig, P., Guida, F., Ricceri, F., d'Errico, A., Barros, H., Bochud, M., Chadeau-Hyam, M., Clavel-Chapelon, F., Costa, G., Delpierre, C., Fraga, S., Goldberg, M., Giles, G. G., Krogh, V., Kelly-Irving, M., . . . Kivimäki, M. (2017). Socioeconomic status and the 25 × 25 risk factors as determinants of premature mortality: A multicohort study and meta-analysis of 1.7 million men and women. *Lancet*, 389(10075), 1229–1237. https://doi.org/10.1016/S0140-6736(16)32380-7
- Tan, J. J. X., & Kraus, M. W. (2018). Judgments of interpersonal warmth predict class differences in political candidate support. *Social Cognition*, 36(1), 106–133. https://doi.org/10.1521/soco.2018.36.1.106
- Tan, J. J. X., Kraus, M. W., Carpenter, N. C., & Adler, N. E. (2020). The association between objective and subjective socioeconomic status and subjective well-being: A metaanalytic review. *Psychological Bulletin*, 146(11), 970–1020. https://doi.org/10.1037/bul0000258
- Todorov, A., Mandisodza, A. N., Goren, A., & Hall, C. C. (2005). Inferences of competence from faces predict election outcomes. *Science*, *308*(5728), 1623–1626. https://doi.org/10.1126/science.1110589
- Todorov, A., Olivola, C. Y., Dotsch, R., & Mende-Siedlecki, P. (2015). Social attributions from faces: Determinants, consequences, accuracy, and functional significance. *Annual Review of Psychology*, 66(1), 519–545. https://doi.org/10.1146/annurev-psych-113011-143831
- Valcke, B., van Hiel, A., van Assche, J., van Roey, T., Onraet, E., & Roets, A. (2020). The need for inclusion: The relationships between relational and collective inclusion needs and psychological well-and ill-being. *European Journal of Social Psychology*, 50(3), 579–596. https://doi.org/10.1002/ejsp.2641
- Vignoles, V. L., Owe, E., Becker, M., Smith, P. B., Easterbrook, M. J., Brown, R., González, R., Didier, N., Carrasco, D., Cadena, M. P., Lay, S., Schwartz, S. J., Des Rosiers, S. E., Villamar, J. A., Gavreliuc, A., Zinkeng, M., Kreuzbauer, R., Baguma, P., Martin, M., . . . Bond, M. H. (2016). Beyond the 'east–west' dichotomy: Global variation in
cultural models of selfhood. *Journal of Experimental Psychology: General, 145*(8), 966–1000. https://doi.org/10.1037/xge0000175

- Weber, M. (1958). *The protestant ethic and the spirit of capitalism* (T. Parsons, Trans.). Scribner. (Original work published 1904 and 1905)
- Wiggins, J. S. (1991). Agency and communion as conceptual coordinates for the understanding and measurement of interpersonal behavior. In D. Cicchetti & W. M. Grove (Eds.), *Thinking clearly about psychology, Volume 2: Personality and psychopathology* (pp. 89–113). University of Minnesota Press.
- Witkower, Z., Tracy, J. L., Cheng, J. T., & Henrich, J. (2020). Two signals of social rank: Prestige and dominance are associated with distinct nonverbal displays. *Journal of Personality and Social Psychology*, 118(1), 89–120. https://doi.org/10.1037/pspi0000181
- Yoo, B., Donthu, N., & Lenartowicz, T. (2011). Measuring Hofstede's five dimensions of cultural values at the individual level: Development and validation of CVSCALE. *Journal of International Consumer Marketing*, 23, 93-210.
- Zhang, H., Talhelm, T., Yang, Q., & Hu, C. S. (2021). High-status people are more individualistic and analytic-thinking in the west and wheat-farming areas, but not ricefarming areas. *European Journal of Social Psychology*, 51(6), 878–895. https://doi.org/10.1002/ejsp.2778

Appendix: Manuscripts

The three manuscripts that underly this dissertation are attached in this Appendix in the following order:

- Unkelbach, F., Brütting, T., Schilling, N., & Wänke, M. (2023). Looking competent does not appeal to all voters equally: The role of social class and politicians' facial appearance for voting likelihood. *Personality and Social Psychology Bulletin*, 0(0). https://doi.org/10.1177/01461672231181465
- Unkelbach, F., John, M., & Vogel, V. (2023). Jumping on the bandwagon: The role of voters' social class in poll effects in the context of the 2021 German federal election. *Politische Vierteljahresschrift*, 64(1), 51–78. <u>https://doi.org/10.1007/s11615-022-00417-3</u>
- Unkelbach, F., Kehder, P., & Wänke, M. (2023). Do consumers of all social classes prefer best sellers? Three pre-registered replication studies of Na et al. (2016). Manuscript submitted for publication.

The online supplements for the three manuscripts can be retrieved from:

Manuscript 1: https://doi.org/10.1177/01461672231181465

Manuscript 2: https://doi.org/10.1007/s11615-022-00417-3

Manuscript 3: https://osf.io/fmc29/?view_only=34bc5c1f50df4bb9b46c33eb5b81e50d

TITLE: Looking Competent Does Not Appeal to All Voters Equally: The Role of Social Class and Politicians' Facial Appearance for Voting Likelihood

© 2023 by the Society for Personality and Social Psychology, Inc. This version of the article has been accepted for publication after peer review but is not the copy of record and may not exactly replicate the final, authoritative version of the article. Please do not copy or cite without authors' permission. The final published article is available online at:

Unkelbach, F., Brütting, T., Schilling, N., & Wänke, M. (2023). Looking competent does not appeal to all voters equally: The role of social class and politicians' facial appearance for voting likelihood. *Personality and Social Psychology Bulletin*, 0(0). https://doi.org/10.1177/01461672231181465

Abstract

Voters generally value competence in politicians. Four studies, all conducted in Germany, show that this is especially pronounced in people of higher compared to lower social class. The first study, with a representative sample ($N_1 = 2239$), found that the reported importance of competence in politicians increased with increasing socio-economic status (SES). This was mediated by self-perceived competence which was higher in participants of higher SES. In three further studies (two pre-registered, $N_{2a\&2b} = 396$, $N_3 = 400$) participants merely saw pictures of politicians' faces. Perceived competence based on facial appearance increased the likelihood of voting for a politician. Again, this effect was stronger among participants of higher compared to lower SES. This moderation persisted after controlling for participants' political orientation and politicians' perceived warmth and dominance. We discuss implications for future research on the psychological underpinnings of social class as well as appearance effects in the political context.

Political psychology; Appearance-based politics; Big Two; Voting behavior; Self-Concept

Looking Competent Does Not Appeal to All Voters Equally: The Role of Social Class and Politicians' Facial Appearance for Voting Likelihood

Voting decisions are to some extent influenced by candidates' presumed personality characteristics (Funk, 1999; Kinder et al., 1980; Laustsen & Bor, 2017; Miller et al., 1986; Olivola & Todorov, 2010a; Todorov et al., 2005). Especially perceived competence appears to be relevant for voters. Voters' ratings of the perceived competence of political candidates were positively correlated with their overall evaluation (Funk, 1999; Miller et al., 1986) and with voting for them (Laustsen & Bor, 2017). Even in the absence of further information merely looking competent seems important for electoral success (Antonakis & Dalgas, 2009; Ballew & Todorov, 2007; Olivola & Todorov, 2010a, 2010b; Todorov et al., 2005; Todorov et al., 2015). For example, in Todorov et al.'s (2005) seminal study, participants' competence judgments based on portraits of candidates predicted not only the actual election outcome but also the difference in votes between candidates.

Despite numerous evidence for the importance of perceived competence, one may wonder whether competence is equally relevant for all voters. We argue that the importance of politicians' perceived competence for voting likelihood depends on voters' social class with those of higher social class valuing competence more.

Why competence may appeal more to members of high social class

Before we delineate our hypothesis a definition of the key concepts is in order. Social class can be conceptualized by objective components that describe the material resources an individual possesses (Kraus et al., 2012) such as financial resources, educational attainment and occupational prestige. We refer to these as objective socioeconomic status (SES). Previous research shows that SES is linked with individuals' perceived rank in society in comparison to others, a concept often referred to as subjective social status (SSS; Adler et al., 2000; Kraus et al., 2013). While SSS and SES tend to be moderately positively correlated, SSS goes beyond SES by capturing social comparison processes aside from objective assessments (e.g., Tan et al., 2020).

With regard to our second central concept, competence, we follow the concepualization of the Big Two framework (Abele et al., 2016; Abele & Wojciszke, 2007, 2014) which postulates two main content dimensions of social judgment, namely agency and communion. In this model, competence is conceptualized as one subfacet of agency and defined as the ability to accomplish tasks, i.e. intelligence and skill. In contrast, assertiveness, the second facet of agency, is understood as one's motive to promote the self, i.e., ambitiousness and self-confidence. Communion, on the other hand, encompasses the two subfactes warmth (reflecting empathy and likeability) and trustworthiness (reflecting sincerety and honesty).

Importantly, stereotypes of high and low status groups differ on these dimensions. High-status people are believed to be more competent than low-status people (Cuddy et al., 2008; Fiske et al., 2002; Fiske et al., 2007). This stereotype of higher competence of high status groups seems to be shared in many societies (Durante et al., 2013) and, crucial for our argument, is shared by high status people about themselves (Abele, 2003; Belmi et al., 2020; Oldmeadow & Fiske, 2010). Assuming that competence plays a larger role in the self-concept of high- compared to low-status people (see Kraus et al., 2012) we argue that this trait can be considered as chronic self-relevant knowledge, i.e., a schematic trait.

Schematic traits guide the processing of information about oneself and about others (Fong & Markus, 1982; Markus, 1977; Shoda & McConnell, 2013). When perceiving others people primarily attend to information related with their own self-schema (Fong & Markus, 1982; Riggs & Cantor, 1984) and weigh these information more heavily when forming an impression of the target (Carpenter, 1988; Sedikides & Skowronski, 1993). Indeed, people from a higher social class distinguish others especially with regard to their perceived competence (Oldmeadow & Fiske, 2010). Accordingly, one may expect that higher-class voters base their evaluation of politicians more on perceived competence than lower-class voters.

This assumption is in line with findings that voters generally evaluate politicians more positively when they share personality characteristics (Caprara et al., 2007; Caprara & Zimbardo, 2004). As higher-class voters feel more competent, they may perceive candidates who appear more competent as more similar.

Specifically, we argue that people of higher SES are competence-schematics and should therefore value competence in politicians more than people of lower SES. People of higher SES share specific experiences during their socialization in an environment that is determined by educational and financial achievement (e.g., Stephens et al., 2014), and as competence is instrumental to both the concept is omnipresent and easily accessible. These experiences might foster a class-specific tendency to establish competence as schematic trait and distinguish and evaluate others accordingly.

For SSS, we do not have a firm hypothesis regarding a preference for competence. One may assume that people who think of themselves as high in social status also ascribe themselves the attributes stereotypically ascribed to high status people. This assumption would predict similar effects for SES and SSS if self-schema is responsible for the effects. However, it is unclear whether the self-schema of people high in SSS parallels that of people high in SES. It is noteworthy that previous studies on the importance of candidate traits found diverging effects of SES and (manipulated or chronic) SSS (Callaghan et al., 2022; Tan & Kraus, 2018).

For our main hypothesis that voters high in SES weigh competence in a politician more heavily than lower-class voters there is already some supporting evidence. Analyses of data from the American National Election Studies (ANES) show that voters with higher educational attainment were more likely to mention competence when asked to name aspects that might make them vote for/against presidential candidates than less educated voters (Miller et al., 1986). Importantly, this group of voters was more likely to state that competence (vs. warmth) is an important characteristic in an ideal US president (Kinder et al., 1980). Likewise, a more recent analysis of ANES data (Laustsen & Bor; 2017; Table SI.11b) reveals a significant interaction between voters' educational level and politicians' perceived competence on vote choice in the proposed direction.⁶

Moreover, Callaghan et al. (2022) presented descriptions of politicians in form of word clouds expressing competence or warmth. Compared to participants of lower SES those of higher SES indicated a higher voting likelihood for the candidate presented as competent

⁶ The authors do not interpret this effect.

and were more likely to prefer the competent over the warm candidate. The present research goes beyond the previous evidence by not only relying on self-reported importance but moreover by testing whether the effect manifests for the impact of competence perceived from facial appearance on voting preferences.

The Present Research

Four studies tested the hypothesis that the perceived competence of a political candidate has a larger impact on the propensity of voting for this candidate among voters with higher SES.

In contrast to previous research which has focused on US samples, we investigate the association between social class and important candidate traits in the German context. Beyond reasons of external validation, a conceptual replication outside the USA is particularly relevant in this case as the societal structure of the USA may have contributed to the previously observed effects.

The fact that people of higher social class consider competence as more important than people of lower social class may be particularly pronounced in the USA for two reasons: Firstly, competence is a trait which is specifically valued in individualistic cultures. Within a culture people who excel on the dimensions valued in this culture are more likely to ascend to higher positions and in turn may hold those values particularly dear (Gobel & Miyamoto, 2022). Thus, the association of higher status with more individualistic traits is particularly pronounced in individualistic cultures (Zhang et al., 2020; Gobel & Miyamoto, 2022). It should be noted that although Germany is more individualistic than collectivistic (Hofstede et al., 2010) it is much less individualistic than the USA, the most individualistic country in the world. Possibly the association between social class and a preference for competence is stronger in the USA due to its extreme position on the individualism-collectivism scale. In Germany, as a less individualistic society, social classes may differ less in how they value such individualistic traits.

Secondly, compared to the USA, Germany is characterized by a lower level of economic inequality (Gini index 2018: 41.4 vs. 31.7; World Bank, 2023). In regions with higher income inequality social status appears to be more salient (Paskov et al., 2013) and social class effects on psychological outcomes tend to be larger (e.g., Schneider, 2019). Also, social class stereotyping, i.e., the perception of higher social classes as more competent compared to lower social classes, increases with higher inequality (Durante et al., 2017). All this considered, social class effects may be particularly pronounced in the USA.

Investigating the effects across a different political and societal context, our studies provide important information on their robustness and generalizability. In Study 1, we relied on self-reports. Participants of a representative sample (N = 2239) rated the importance of competence and other attributes in a politician, and also rated how competent they perceived themselves which allowed us to explore the presumed mediation of self-concept.

In Studies 2a/b and 3, we employed a less obtrusive methodology. Going beyond previous studies we did not rely on self-reported trait importance, which may be prone to self-presentation effects. Rather we drew on findings showing an advantage for competent-looking politicians (Todorov et al., 2005). Participants were shown portrait photographs of politicians who varied in their perceived competence according to pretest ratings. For each politician

participants indicated their voting likelihood. Using pictures rather than self-reported importance tested whether perceived competence is spontaneously more appealing to voters of higher compared to lower social class.

In all studies we explore the role of chronic SSS in addition to SES. Study 1 explored whether SSS and SES had similar effects on people's self-schema and whether possible differences in self-schema might be responsible for the diverging effects of SES and SSS found in previous research by Callaghan et al. (2022).

We had no firm expectation about the importance of warmth in politicians depending on SES or SSS. Assuming that social classes differ in how important they consider certain traits in a politician because of differences in their self-concept, it is not clear what to expect regarding the importance of warmth. Research shows mixed results on the class stereotype on warmth depending on country (Durante et al., 2013, but see also Durante et al., 2017). More crucially, regarding self-concept there was either no (Abele, 2003) or even a weak negative relationship (Boileau, 2022). Finally, there is evidence that individuals with lower SES only prefer warm candidates when their warmth appears to be genuine (Tan & Kraus, 2018).

Two studies were preregistered (Study 2b: <u>https://aspredicted.org/436zf.pdf</u>, Study 3: <u>https://aspredicted.org/s3743.pdf</u>). All preregistrations included study design, stopping rule, exclusion criteria and planned analyses. All preregistered analyses are reported in the manuscript and any deviations were marked. For each study, sample size was determined before data analysis. All measures, manipulations, and exclusions of the presented studies are reported.

Study 1

In Study 1, we investigate if people's social class is related to how important they consider competence in political candidates. Specifically, we rely on self-reported importance ratings which were part of a large-scale survey with a nationally representative German sample that was originally conducted as part of a different, unrelated research project. To isolate the association between social class and the reported importance of competence, we tested this effect relative to the other three main facets of social judgments in the Big Two model (Abele et al., 2016) – assertiveness, warmth and trustworthiness.

Especially separately investigating competence (as ability) and assertiveness (as motivational component) might provide new insights. While most of the previous research in this domain referred to competence but actually measured a combination of both traits (e.g., Callaghan et al., 2022; Funk, 1999), there are first studies indicating that competence and assertiveness should be dissociated when investigating weight attached to candidate traits (Mignon et al., 2016). Consequently, a differentiation between both traits can help to better understand social class effects.

We also assessed participants' self-schema to explore its role as possible mediator. In addition to investigating the role of SES, we explored whether SSS showed a similar pattern.

Finally, we attempt to rule out individuals' political orientation as alternative explanation. A higher income has often been linked with a more right political orientation (e.g., Page et al., 2013). Additionally, political orientation is associated with the value attached to assertiveness (Eriksson, 2018). Even though competence has often been considered a non-ideological dimension (e.g., Mignon et al., 2016), it is crucial to clarify if

political orientation plays a role in the association between social class and the importance of competence in politicians.

Method

All materials (exceptions for Study 2a/b and 3: politicians' pictures (shared upon request); materials of the pretest), R code for all reported analyses, data and codebooks are available at: <u>https://osf.io/rfvbj/</u>

Materials. Participants indicated how important they considered being "competent", "assertive", "likeable" and "trustworthy" for a politician on a scale from 1 *(not at all important)* to 5 *(very important)*.

To explore the presumed mediation by self-schema, we assessed participants' selfperception on these characteristics with three attributes each (Cronbach's $\alpha = .77-.87$) on a scale from 1 (*not at all*) to 5 (*completely*) (see Supplement, Tables S1/2).

Participants. Data from 2469 German participants were collected via an online access panel provider. The sample was representative for the German population regarding age, gender and education. After excluding participants who did not know their annual household income and participants who did not want their answers to be used, we arrived at a final sample of 2239 participants (1095 female, 1136 male, 8 diverse, $M_{age} = 46.02$ years, $SD_{age} = 14.33$). This sample size was sensitive to detecting an effect size of r = .05 or higher given 80% power and alpha = 0.05 (Faul et al., 2009).

Procedure. The questions were part of a larger online study which was conducted with SoSci Survey (Leiner, 2019) and introduced as a study on political attitudes. All variables of interest for Study 1, aside from people's self-perceived traits, were assessed before an experimental manipulation that was part of a different, unrelated research project. Self-perceived traits did not differ significantly between the experimental and control groups (all ps > .200).

Firstly, demographics were assessed (gender, age, federal state). Then, SSS was measured with the MacArthur Scale (e.g. Adler et al., 2000) and participants indicated their educational level on a scale with 8 options plus an "other" option (*Median* = secondary school certificate). Next followed further questions on (political) attitudes that were not relevant to the present paper, as well as one item for global political orientation (1 = left; 9 = right). Then, participants rated the importance of competence and further candidate traits as described above as well as the importance of further characteristics irrelevant to the present paper (e.g., having a good concept for climate protection). After an experimental manipulation and further variables irrelevant to this paper (see Supplement (1) for a complete list) participants indicated their self-reported competence, assertiveness, warmth and trustworthiness. Finally, they reported their current monthly net household income on category options from 1 (*below* 500€) to 13 (10000€ and more) (*Median* = 2000€ – below 2500€) plus a "don't know" option.

Results

Reported importance of competence. Overall, participants indicated that competence in a politician was rather important (M = 4.39, SD = 0.85, 5-point scale). We coded and z-standardized educational level and household income and created an index of SES by taking their mean (see Kraus & Keltner, 2009). Consistent with our hypothesis, a higher SES was significantly positively associated with reporting a higher importance of competence in political candidates, r(2237) = .08, 95% CI [0.04, 0.13], p < .001.

However, the investigated sample was representative only with regard to education but not income. A comparison of the income distribution in our sample with the German population (German Federal Statistical Office, 2022) shows that our sample is characterized by an overrepresentation of individuals with lower household income and an underrepresentation of individuals from the highest income category of 5000 and above (see Supplement, Table S3). Thus, we created a subset which was representative for the German population regarding household income. For this purpose, we divided the sample into ten income categories parallel to those described in the Microcensus and randomly drew individuals from these subgroups to obtain a new dataset with the same ratio of income categories as the German population. In this subset (N = 1200), again a higher SES was significantly associated with a higher importance of politicians' competence, r(1198) = .10, 95% CI [0.04, 0.15], p < .001.

Robustness checks. Overall, SES was not associated with political orientation, r(2237) = .01, 95% CI [-0.03, 0.05], p = .597. Importantly, the association between SES and the rated importance of competence remained robust when controlling for political orientation, b = .09, 95% CI [0.05, 0.14], SE = 0.02, t(2239) = 4.06, p < .001.

As shown in Table 1, SES was not significantly related to the reported importance of assertiveness or warmth, but unexpectedly, people with higher SES indicated a higher importance of trustworthiness. When controlling for the reported importance of the other three facets of the Big Two model (Abele et al., 2016), SES continued to positively predict the importance of competence in politicians, b = .06, 95% CI [0.03, 0.09], SE = .01, t(2234) = 3.92, p < .001.

Subjective social status (SSS). Overall, SSS was positively correlated with SES, r (2237) = .53, 95% CI [0.50, 0.56], p < .001. Similar to SES, the importance of competence increased with higher SSS, r(2237) = .07, 95% CI [0.03, 0.11], p < .001. However, SSS was significantly positively correlated with the reported importance of all of the candidate traits (see Table 1). In contrast to SES, SSS was not significantly related to the importance of competence of competence when controlling for the importance of assertiveness, warmth and trustworthiness, b = .01, 95% CI [-0.00, 0.02], SE = .01, t(2234) = 1.23, p = .219.

Table 1

Descriptive Statistics and Bivariate Correlations among Objective SES, Subjective Social Status, Political Orientation, the Rated Importance of Candidate Traits and Voters' Traits (Study 1).

	1	2	3	4	5	9	L	8	6	10	11
1. SES											
2. SSS	.53										
3. Political Orientation	.01	.07									
4. Importance Competence	.08	.07	06								
5. Importance Assertiveness	.01	.07	.02	.65							
6. Importance Warmth	00	.07	00.	.47	.48						
7. Importance Trustworthiness	.05	90.	05	.72	.65	.54					
8. Own Competence	.19	.25	.05	.35	.35	.28	.33				
9. Own Assertiveness	.14	.26	.13	.13	.23	.19	.15	.62			
10. Own Warmth	.01	.10	03	.37	.37	.36	.38	.67	.36		
11. Own Trustworthiness	.05	.07	04	.43	.39	.32	.42	.68	.34	.83	
M	0.00	5.03	5.63	4.39	4.16	3.84	4.29	3.66	3.12	3.86	3.99
SD	0.79	1.75	1.86	0.85	0.86	0.96	0.87	0.77	0.85	0.85	0.83
Note: $N = 2239$. SES = Socio-economic	status. SS	SS = Subi	ective soc	ial status.	Political (Orientatio	n was asse	essed on a	scale fron	1(<i>left</i>) to	9 (right).

VI IZIUI). (ichi) r Self-perceived traits were mean scores based on three items per trait.

All correlations $\geq .05$ or $\leq ..05$ are statistically significant (p < .05).

Mediation via self-perceived competence. SES and self-perceived competence correlated positively, r(2237) = .19, 95% CI [0.15, 0.23], p < .001. A mediation analysis using the R package lavaan (version 0.6-12; Rosseel, 2012) suggests that the effect of SES on the importance of competence was completely mediated via self-perceived competence, b = 0.07, SE = 0.01, z = 8.00, p < .001, 95% CI [0.05, 0.09]. The subsample representative for household income showed the same result, b = 0.09, SE = 0.01, z = 6.69, p < .001, 95% CI [0.06, 0.11].

Furthermore, SSS was positively associated with the self-schema of all of the four traits (see Table 1).

Discussion

Supporting our hypothesis and using a representative German sample, we found that the higher the SES, the higher the importance of politicians' competence. This association was independent of political orientation. Furthermore, the data offer preliminary support for differences in the self-concept as presumed cause for this relationship. More concretely, voters with higher SES perceived themselves as more competent and this self-view completely mediated the relationship between SES and the importance of politicians' competence. This is in line with previous findings from the USA that higher-class voters perceive a greater interpersonal closeness with competent politicians (Callaghan et al., 2022). However, as all variables were measured our study offers merely correlational evidence. Thus, we cannot rule out that other potential mediators may also play a role.

Additionally, we acknowledge that the way we assessed self-concept does not inform us to what extent the traits were part of the spontaneously accessible self-concept (schema) that would guide the perception of others (Fong & Markus, 1982; Markus, 1977). While people of high SES also rated themselves as more assertive than people of low SES there was no difference in the importance of this trait for a politician. It seems plausible that people of higher SES are schematic on competence but not on assertiveness as there are many situations and experiences over a lifetime that form and reinforce their perception of being competent. Moreover, competence is unambiguously considered positive. Experiences of assertiveness are probably much less frequent for many people and assertiveness has also negative connotations.

We had predicted the influence of SES on the importance of politicians' competence but had no hypothesis regarding assertiveness which we only assessed as a control variable. Our findings suggest that a distinction between competence that is merely related to ability and other aspects of taking effective action (namely assertiveness) as suggested by Abele et al. (2016) seems appropriate when assessing the appeal of politicians.

For SSS we found a different pattern. Whereas the rated importance of competence increased with SSS similarly to SES, this relationship was not robust when controlling for the importance of other traits. Apparently, in contrast to people of high SES, people with high SSS do not specifically value competence but generally value desirable traits in politicians more compared to people with lower SSS. A recent meta-analysis (Tan et al., 2020) suggests that high positive correlations between the SSS-ladder and other measures may reflect a positive response bias, which would explain why in our study the correlations between SSS and own traits were higher than for SES. This would imply that the higher relevance of

competence among people high in SSS does not necessarily reflect similar processes as in people high in SES. We will return to this issue in the General Discussion.

Altogether we found clear support for our hypothesis that people of higher SES value competence more in politicians compared to people of lower SES. Yet, one caveat needs to be addressed. We only measured what people explicitly reported to be important traits in politicians. It is an open question whether this would really influence their voting decisions. People may mention competence as an important characteristic because they believe that this is what is expected of rational voters. Even if they indeed consider competence as highly important other characteristics may have a larger influence when it comes to actual voting without voters being consciously aware of it. Thus, in the following studies, we investigated the influence of perceived competence on voting likelihood dependent on social class while varying perceived competence less obtrusively and not asking participants directly about how important they considered this predictor.

Study 2a & 2b

The work by Todorov and colleagues (2005; see also Ballew & Todorov, 2007; Olivola & Todorov, 2010a, 2010b) shows that competence perceived only by a politician's looks influences voting decisions. Based on the results of Study 1, we expect this effect to be stronger for voters with higher SES. To test this hypothesis, we presented pictures of (unknown) politicians' faces with high versus low perceived competence in Study 2a/b and measured voting likelihood. This procedure allowed us to test the influence of perceived competence depending on SES in a rather subtle manner. Additionally, we differentiated the effects of perceived competence from warmth. Warmth appears to be an important basis of comparison as it is typically juxtaposed with competence as a main dimension of interpersonal perception (e.g., Fiske et al., 2002) and people tend to make rapid judgments of this trait (e.g., Fiske et al., 2007). Again, we measured SSS and explored if it played the same role as SES.

Method

Materials. In each of the two studies, we used ten portrait photos selected from a sample of 63 male members of the Swiss national parliament. All pictures were taken from the parliament's website which ensured a uniform portrait style, standard business attire and a similar background. In a pretest, 80 participants rated subsets of portraits regarding the attributes "competent" and "warm" on a scale from 1 (*not at all*) to 11 (*very*) such that each portrait was rated by 20 participants. Our goal was to select one set of pictures with low and one with high perceived competence but to keep warmth at a constant level across the sets. Due to an organizational mistake in Study 2a, there were only 4 pictures in the condition of relatively low perceived competence (pretest competence: M = 5.65, SD = 0.56, warmth: M = 6.38, SD = 1.42) and 6 in the condition of relatively high perceived competence (pretest competence set: pretest competence: M = 5.54, SD = 0.40, warmth (measured via "warm", "sincere", "likeable"): M = 5.80, SD = 0.75; high competence set: pretest competence: M = 7.85, SD = 0.27, warmth: M = 5.89, SD = 0.24). The set in Study 2b contained three photos from Study 2a and seven new ones.

Participants. For Study 2a, we collected data from 282 German-speaking participants (with incomplete submissions: 330). For Study 2b, we collected data from 260 German participants on prolific.co (with incomplete submissions: 266). An a-priori power analysis using G*Power for alpha = .05 and a power of 80% to detect a small effect of f = 0.10 for the interaction effect of a 2 x 2 mixed ANOVA as a proxy resulted in a sample size of 200. We applied our pre-registered exclusion criteria for Study 2b (failing the attention check, familiarity with any of the politicians, missing/invalid values on key variables, being no native German speaker⁷) also to Study 2a and additionally excluded participants who failed a seriousness check or were below 18 years. After exclusions we arrived at a final sample size of N = 396 ($N_{2a} = 195$, $N_{2b} = 201$) participants (220 female, 172 male, 4 diverse; $M_{age} = 28.32$ years, SD = 9.93; Study 2a: 138 female, 56 male, 1 diverse; $M_{age} = 26.06$ years, SD = 9.46; Study 2b: 82 female, 116 male, 3 diverse; $M_{age} = 30.51$ years, SD = 9.90).

Procedure. The procedures in Study 2a/b were almost identical. In both online studies, participants indicated their interest in politics (1 = not at all; 7 = very much); their political orientation (1 = left; 9 = right), their voting regularity (11-point scale with higher values indicating a greater regularity) and whether they had been eligible to vote in the last German federal election. Next, SSS was assessed via the MacArthur scale.

Participants were instructed that they would see 10 pictures of randomly selected male politicians. They viewed the faces in an individualized randomized order. Based on measures of Fiske et al. (2002), participants rated perceived warmth ("warm", "friendly", "sincere"; α Study 2a/b = .82/.82) and competence ("competent", "intelligent", "ambitious"; α Study 2a/b = .74/.65) for each politician on a scale from 1 (*not at all*) to 5 (*extremely*) as manipulation check (see Supplement for the German version, Table S4). After this check, participants indicated how likely they would consider voting for this candidate from 1 (*not at all likely*) to 11 (*very likely*).

Finally, participants completed the measures of SES among further demographics (gender, age, German language skills). SES was assessed via educational level as in Study 1 (Study 2a: *Median* = high school diploma; Study 2b: *Median* = university degree) and current annual household income (on a scale from 1 (*below 15,000€*) to 8 (*over 150,000€*) (plus "don't know" option in Study 2a); Study 2a: *Median* = 50,001€ - 75,000€; Study 2b: *Median* = 35,001€ - 50,000€).

Paralleling Study 1, we created an index of SES. Due to the relatively high proportion of students, we additionally computed an *adjusted index* of SES for students and apprentices based on their parents' education (mean of father's and mother's educational level) and current family household income.

Finally, participants indicated if they recognized any of the politicians so that we could exclude these cases. In Study 2a, they indicated whether they had answered all the questions seriously.

⁷Even though we had preregistered for Study 2b to exclude non-native German speakers, we did not apply this strict criterium to Study 2a. Here, we included 7 participants who reported to speak German for more than 10 years. Excluding these did not change the results.

Results

Manipulation check. In both studies, politicians in the high competence set were perceived as significantly more competent than politicians in the low competence set, Study 2a: $M_{\text{high}} = 3.48$, $SD_{\text{high}} = 0.44$, $M_{\text{low}} = 2.95$, $SD_{\text{low}} = 0.50$, t(194) = 16.10, p < .001, d = 1.15, 95% CI for d [0.97, 1.33]; Study 2b: $M_{\text{high}} = 3.56$, $SD_1 = 0.44$, $M_{\text{low}} = 3.18$, $SD_2 = 0.43$, t(200) = 14.48, p < .001, d = 1.02, 95% CI for d [0.85, 1.19]. Although based on the pretest perceived warmth should not differ between the target politicians, competent-looking politicians were rated as less warm than less competent-looking politicians (Study 2a: $M_1 = 3.12$, $SD_1 = 0.46$, $M_2 = 3.17$, $SD_2 = 0.48$, t(194) = -1.72, p = .087, d = -0.12, 95% CI for d [-0.26, 0.02], Jeffrey-Zellner-Siow (JZS) Prior BF = 2.94; Study 2b: $M_{\text{high}} = 2.78$, $SD_{\text{high}} = 0.50$, $M_{\text{low}} = 3.21$, $SD_{\text{low}} = 0.53$, t(200) = -13.36, p < .001, d = -0.94, 95% CI for d [-1.11, -0.78], JZS Prior BF > 1000).

Likelihood of voting. Due to the parallel design of Study 2a/b and to test our hypotheses with more power, we used a joint analysis including data source (Study 2a or 2b) as additional predictor (see Supplement (3) for separate analyses and the pre-registered linear regression analysis). We conducted a multilevel linear regression with maximum likelihood estimates. The joint model specified a within-subject relationship between perceived competence and voting likelihood that we had predicted to be stronger for participants with higher SES. We grand-mean centered SES as level-2-predictor and used person-mean centering for condition as effect-coded level-1-predictor (-1 = 10w competence, 1 = high competence) (see Enders & Tofighi, 2007). Data source was included grand-mean centered as dummy-coded level-2-predictor (0 = Study 2a, 1 = Study 2b).

The results are displayed in Table 2. Competence condition had a significant effect on voting likelihood, b = 0.22, SE = 0.03, t = 7.12, p < .001, 95% CI [0.16, 0.28]), implying that participants indicated on average a higher voting likelihood for politicians from the high compared to the low competence set. Consistent with our hypothesis, this effect was moderated by SES (b = 0.12, SE = 0.04, t = 2.90, p = .004, 95% CI [0.04, 0.20]).⁸ The remaining model terms were insignificant (see Table 2, model 1).

Simple slope analyses revealed that the strongest effect of competence condition on voting likelihood was found among participants with high SES (+1 *SD*; *b* = 0.31, *SE* = 0.04, *t* = 7.08, p < .001, 95% CI [0.23, 0.40]), holding all other covariates constant. At low levels of SES, competence condition still had a positive, but smaller effect (-1 *SD*; *b* = 0.13, *SE* = 0.04, *t* = 2.99, p = .003, 95% CI [0.05, 0.22]). A Johnson-Neyman plot (see Figure 1) shows that the effect of competence condition was significant for all values of SES above -1.00 and increased with SES.

Furthermore, the results of parallel analyses using the adjusted score of SES showed similar significance levels (see Supplement, Table S7).

⁸ Separate analyses showed that the interaction effect was only significant in Study 2a, p = .021, and not in Study 2b, p = .224. In Study 2b, the interaction only reached significance for the adjusted score, p = .017. Hence, we conducted a meta-analysis across Study 2a, 2b and 3 (see below).



Figure 1. Conditional effect of competence condition on voting likelihood as a function of objective SES

Note. Johnson-Neyman plot for the effect of competence condition on voting likelihood as a function of objective SES. In addition to the point estimate, the 95% Confidence Interval is displayed. The plot was created using the R package interactions (version 1.1.5; Long, 2019).

Exploratory Analyses.

Measured competence as predictor. As the manipulation check indicated that perceived warmth and competence of the two sets were confounded, we analyzed the data using measured competence and warmth as predictors. In a parallel analysis, we included both measured traits (from the manipulation check) as level-1-predictors (person-mean centered) with random slopes instead of competence condition. Perceived competence had a significant effect on voting likelihood, b = 1.32, SE = 0.04, t = 30.80, p < .001, 95% CI [1.23, 1.40]. Additionally, warmth had a significant effect, b = 1.51, SE = 0.04, t = 39.80, p < .001, 95% CI [1.43, 1.58]. Importantly, only the interaction effect between competence and SES was significant, b = 0.19, SE = 0.06, t = 3.21, p = .001, 95% CI [0.07, 0.30]⁹, but the interaction between warmth and SES was not, b = -0.03, SE = 0.05, t = -0.61, p = .543, 95% CI [-0.13, 0.07]. Irrelevant to our hypothesis, the effect of warmth was larger in Study 2b compared to Study 2a, b = 0.18, SE = 0.08, t = 2.40, p = .017, 95% CI [0.03, 0.33]. Furthermore, the effect of competence on voting likelihood was larger for politicians with high warmth, b = 0.26, SE = 0.04, t = 6.99, p < .001, 95% CI [0.19, 0.34]. The other model terms were non-significant (see Table 2, model 2). Controlling for political orientation, political interest and voting

⁹ This was also the case for 2b, p = .007, while the effect was marginally significant in study 2a, p = .057.

regularity did also not change the pattern of results (p < .001). Further robustness checks can be found in the supplement (Table S8-11).

Subjective Social Status (SSS). When including SSS in the main analyses instead of SES neither the interaction with the competence condition nor the interaction with measured competence on voting likelihood was significant, p = .196 / p = .851 (see Supplement, Table S10).

Parameter Estimates for Multi.	level Model	of Voting 1	Likelihood (S	otudy 2a an	ıd 2b)					
			Model 1					Model 2		
- Fixed effects	p	SE b	t	d	95% CI ^a	<i>q</i>	SE b	t	d	95% CI ^a
Intercept	5.46	0.07	76.50	<.001	5.32, 5.60	5.47	0.07	76.35	<.001	5.33, 5.61
Competence - Condition	0.22	0.03	7.12	<.001	0.16, 0.28					
Meas. Warmth						1.51	0.04	39.80	<.001	1.43, 1.58
Meas. Competence						1.32	0.04	30.80	<.001	1.23, 1.40
Obj. SES	0.05	0.10	0.54	.588	-0.14, 0.24	0.05	0.10	0.52	.605	-0.14, 0.24
Source	-0.12	0.14	-0.87	.387	-0.42, 0.17	-0.15	0.14	-1.04	.297	-0.43, 0.13
Meas. Warmth x Source						0.18	0.08	2.40	.017	0.03, 0.33
Meas. Competence x Source						-0.13	0.09	-1.56	.120	-0.30, 0.04
Meas. Warmth x Meas. Competence						0.26	0.04	6.99	<.001	0.19, 0.34
Competence - Condition x Obj. SES	0.12	0.04	2.90	.004	0.04, 0.20					
Meas. Warmth x Obj. SES						-0.03	0.05	-0.61	.543	-0.13, 0.07
Meas. Competence x Obj. SES						0.19	0.06	3.21	.001	0.07, 0.30
Obj. SES x Source						0.16	0.19	0.82	.410	-0.22, 0.54
Meas. Warmth x Obj. SES x Source						-0.10	0.10	-0.94	.349	-0.31, 0.11

Table 2

MANUSCRIPT 1: SOCIAL CLASS AND COMPETENCE

15

COMPETENCE	
AND	
LASS	
AL CI	
1: SOCIA	
SCRIPT	
MANU	

Table 2 (continued)

Meas. Competence x Obj. SES x Source			0.06	0.12	0.53	.595	-0.17, 0.29
Random effects ([co-]variances)							
Intercept	1.65				1.90		
Competence-Condition (0.0 0.0	1					
Meas. Competence					0.29	0.27	
Meas. Warmth					0.28	0.04	0.00
Residual	3.74				1.22		
<i>Note. N</i> = 396, 10 pictures per Study, 3,960 observations. ICC for mo Competence-Condition -1 = Low Competence, Competence-Conditio ^a Confidence intervals were computed from the profiled likelihood.	del 1 = .31, ICC f n 1 = High Comp	or model $2 = .64$. $b = unstand$ etence. Meas. = Measured. So	lardized coeff ource 0 = Stu	cient, <i>SE</i> = sta ly 2a, Source 1	ndard error = Study 2b.		

16

Discussion

Whereas Study 1 showed that people of higher SES rated politicians' competence as more important than people of lower SES, Studies 2a/b showed that this also translates into differences in preference for competent-looking politicians between voters of higher and lower SES. Replicating the results of previous studies (Ballew & Todorov, 2007; Todorov et al., 2005), participants overall indicated a higher voting likelihood for more (vs. less) competent-looking politicians. As expected, the effect of perceived competence on voting likelihood was larger among participants with higher SES. The interaction effect was small, but the result supports our hypothesis that politicians' perceived competence is more appealing to higher- than to lower-class voters.

Consistent with the finding from Study 1 that the relationship between SSS and the importance of candidates' competence is less robust, SSS did not play a role in the weighting of perceived competence. Apparently, voters with high SSS explicitly express a higher preference for competent politicians but do not spontaneously take this cue into account. We will address this finding together with the results of SSS from Study 1 and 3 in the General Discussion.

Finally, we must consider that the procedure of assessing perceived competence (and warmth) before voting likelihood may have increased its accessibility. Although this would not explain why the effect was stronger for voters with high SES we cannot claim that people of higher SES would also spontaneously give more weight to competence than people with lower SES. To explore this issue we modified our design for Study 3.

Study 3

In Study 3, we changed the design of Studies 2a/b regarding several aspects. Firstly, we assessed voting likelihood before warmth and competence. Secondly, because the dichotomization of faces with low vs. high perceived competence reduces the power, we included a larger sample of pictures in Study 3. Thereby, we ensured that politicians with perceived competence across the whole spectrum of our pretest ratings were included. Thirdly, we replaced the attribute "ambitious" with "capable" to assess perceived competence without any aspects linked to assertiveness.

Finally, we cannot rule out that the effects observed in Studies 2a/b were due to another trait perceived from the faces that was confounded with competence. Of course, we cannot control for all possible traits but chose to control for perceived dominance in Study 3. Dominance plays an important role to achieve social rank aside from demonstrating competence (Cheng et al., 2013). Additionally, more dominant-looking people are perceived as having a higher status (Rahal et al., 2021). Thus, a more dominant-looking politician might also elicit a higher perceived similarity among higher-class voters. In Study 3, we therefore controlled for perceived dominance with separately assessed ratings of the politicians' perceived dominance.

Method

Materials. We used pictures from the same database of politicians as in Studies 2a/b but selected 32 pictures covering the whole range of competence levels based on the pretest. More concretely, we ordered the pictures according to pretest competence ratings and selected every other picture (range on 11-point scale: 5.05 - 8.5). Additionally, we standardized the

background to be completely white for all portraits.

In a further study (N = 96; 32 female, 63 male, 1 diverse; $M_{age} = 31.57$, $SD_{age} = 10.04$) run via prolific.co, participants rated the 32 pictures on either how "competent" (n = 30), "warm" (n = 35), or "dominant" (n = 31) they perceived each politician on a scale from 1 (*not at all*) to 5 (*very*).

Participants. We conducted a simulation using the R package SIMR (version 1.0.6; Green & MacLeod, 2016) to estimate the power for a cross-level interaction effect in a two-level model (Arend & Schäfer, 2019). Results indicated that 370 participants would suffice to provide 80% power to obtain a standardized effect size of 0.10 keeping alpha at 5%. We pre-registered a minimum sample size of 400 valid cases and collected complete data¹⁰ from 425 German participants on prolific.co. Following pre-registered exclusion criteria (failing any attention check, recognizing any of the depicted politicians, missing/invalid values on key variables), we arrived at a final sample of 400 participants (169 female, 220 male, 11 no answer provided; $M_{age} = 30.59$ years, $SD_{age} = 8.83$).

Procedure. Study 3 was conducted online. The beginning of the study and the assessment of covariates was identical to Study 2a/b. Then participants read that they would see 32 pictures of male politicians. In the first phase, they viewed the faces individually and indicated their voting likelihood for each as in Study 2a/b. Thereafter, half of the participants (n = 203) rated first perceived competence for all politicians and then warmth (phases 2 and 3) whereas for the others (n = 197) the order was reversed. In each study phase, pictures were presented in a randomized order. Perceived competence and warmth were rated as in Study 2a/b except that the attribute "ambitious" was replaced by "capable".

Based on these measures, we created two scores for each politician: Competence (Cronbach's α : .81 - .89) and warmth (Cronbach's α : .78 - .87).

Afterwards, participants completed the same measures of SES including educational attainment (*Median* = university degree) and household income (*Median* = 35.001 - 50.000) as in Study 2a/b among further demographics. Finally, they indicated if they had recognized any politician.

Results

Likelihood of voting. Again, we computed a composite score of SES. We then computed multilevel linear regression models for voting likelihood as criterion parallel to Study 2a/b. In model 1, we included measured competence (level 1), SES (level 2) and their interaction. In model 2, we additionally controlled for warmth (level 1) and its interaction with SES. All level-1-variables were person-mean centered and SES was grand-mean centered. We included a random intercept for participants, random slopes for the level-1-predictors and grouped the data by participants.

Again, measured competence significantly predicted voting likelihood (model 1), b = 1.10, SE = 0.03, t = 36.61, p < .001, 95% CI [1.04, 1.16]. Furthermore, warmth and

¹⁰ Among the incomplete participations were 79 participants who failed an attention check and 23 participants who did not complete the questionnaire.

competence were significantly correlated, *repeated-measures* r(12399) = .35, p < .001. Therefore, we controlled for the influence of warmth and its interaction with SES in model 2. In this model, both competence, b = 0.83, SE = 0.03, t = 28.43, p < .001, 95% CI [0.77, 0.88], and warmth, b = 0.72, SE = 0.03, t = 25.30, p < .001, 95% CI [0.67, 0.78], significantly predicted voting likelihood. Although the interaction of competence and SES showed the same trend as in the previous studies when controlling for the interaction between warmth and SES, it fails to reach significance at conventional significance levels (model 2: b = 0.08, SE = 0.04, t = 1.95, p = .052, 95% CI [-0.00, 0.15]). The rest of the model terms were non-significant (see Table 3).

Including political orientation, political interest and voting regularity as covariates did also not change the pattern and the results were robust to using the adjusted SES score (see Supplement for further robustness checks, Table S14-15).

Exploratory Analyses.

Further independent ratings of competence, warmth, and dominance. To disentangle perceived competence from dominance as possibly confounded trait, we had an independent sample rate the faces for competence, dominance, and warmth as described above. We repeated the main analyses using these independent competence ratings as predictor (level 2, grand-mean centered) as well as the interaction with SES to predict voting likelihood. In contrast to the competence ratings from Study 3, politicians' independently perceived competence did not only predict voting for them, b = 1.65, SE = 0.05, t = 33.26, p < .001, 95% CI [1.55, 1.74]), but – in line with our hypothesis – this effect was significantly moderated by SES, b = 0.15, SE = 0.06, t = 2.36, p = .018, 95% CI [0.03, 0.28]).

Additionally including independent dominance and warmth ratings showed that the interaction between competence and SES remained robust, b = 0.18, SE = 0.06, t = 2.78, p = .005, 95% CI [0.05, 0.31]). Furthermore, SES significantly moderated the effect of dominance, b = 0.13, SE = 0.05, t = 2.50, p = .013, 95% CI [0.03, 0.23]). Higher perceived dominance was significantly associated with *lower* voting likelihood at low levels of SES but not at higher levels (see Table S16 in the Supplement for complete results).

MANUSCRIPT 1: SOCIAL CLASS AND COMPETENCE

Table 3

		_
C	Y 	2
-	501	nn
7	5	220
1		2
-	2	ž
	ò	5
	2	1
-	0	20
-	4	Š
F		Ì
	b	, M
	2	1
,	1	22
Ŀ	-	2
ζ	_	
	Ç	5
_	ۍ	2
	0	2
	2	Ż
5		2
•	<	4
-	0	5
	2	2
-	2	2
	÷	11
į	10	2
2		Z
	2	_
ς	4	2
	200	3
•	11	11
	2	11
•	111	5
	ε	3
F	1	1
	20	5
•	371	2
	ì	22
	20	3
	11	2
	-	3

			Model 1					Model 2		
Fixed effects	9	SE b	t	d	95% CI ^a	В	SE b	t	d	95% CI ^a
Intercept	5.12	0.06	79.22	<.001	4.99, 5.25	5.12	0.06	79.22	<.001	4.99, 5.25
Warmth						0.72	0.03	25.30	<.001	0.67, 0.78
Competence	1.10	0.03	36.61	<.001	1.04, 1.16	0.83	0.03	28.43	<.001	0.77, 0.88
Obj. SES	0.12	0.08	1.40	.162	-0.05, 0.28	0.12	0.08	1.40	.162	-0.05, 0.28
Warmth x Obj. SES						-0.02	0.04	-0.41	.681	-0.09, 0.06
Competence x Obj. SES	0.03	0.04	0.86	.393	-0.04, 0.11	0.08	0.04	1.95	.052	-0.00, 0.15
Random effects ([co-]va	riances)									
Intercept			1.57					1.59		
Warmth								0.14		
Competence			0.16	0.29				0.14		
Residual			3.06					2.69		
Note. $N = 400, 32$ pictures, 12,800 (^a Confidence intervals were compute	observations. V ed from the pro	Varmth and co	mpetence refer d.	to the perce	ived traits as measur	ed in Study 3. b) = unstandard	ized coefficien	t, $SE = $ stand	ard error

20

Subjective Social Status (SSS). When including SSS in the main analysis (model 2) instead of SES, it did not significantly interact with measured competence, p = .082.

Discussion

Study 3 replicated the finding from Study 2a/b with more fine-grained stimulus material insofar as looking competent as well as looking warm increased a politician's electoral success. This appeared to be the case even when these dimensions were not made salient.

Regarding our main hypothesis that specifically perceived competence is more important to voters of higher SES, the results are in line with the predictions and our previous studies, but the effect was small and depending on the analysis was or was not significant. The effect fell beneath conventional significance levels (p < .052) when using the competence ratings assessed in the same study as voting likelihood, it was significant when using competence ratings from an independent study (p < .018). Moreover, an analysis of extreme cases, similar to Studies 2a/b (see Supplement (6)), also supported the hypothesis. Given that all analyses pointed in the same direction we interpret the data as supporting our hypothesis and previous studies.

Study 3 ruled out that perceived dominance may have caused the effect. Controlling for dominance strengthened rather than weakened the effect. The independence of competence and dominance is in line with the results of Study 1 which found class effects on the weighting of competence but not of assertiveness and again speaks for separating the subfacets of agency. Interestingly, we find first evidence that dominance can lead to negative evaluations of politicians among lower-class voters while it apparently does not matter to higher-class voters. However, this finding is preliminary, and its deeper discussion exceeds the scope of this paper.

Again, replicating the finding of Study 2a/b, the results for SSS did not parallel those of SES. We discuss the apparent divergent patterns for SES and SSS in the General Discussion.

Mini Meta-Analysis

Overall, Studies 2a/b and 3 point in the same direction that voters' social class moderates the effect of perceived competence from politicians' faces on voting likelihood. However, the effect sizes were small and depending on study and analysis not always significant (Study 3, p = .052). Thus, we conducted a small-scale fixed-effects meta-analysis (Goh et al., 2016) for Studies 2a, 2b and 3 (included separately, see Supplement (8)). For the interaction effect between SES and perceived competence when controlling for warmth and its interaction with SES, the mini meta-analysis showed a significant effect, M r = .14, 95% CI [0.07, 0.20], Z = 3.81, p < .001, attesting that a politician's competent appearance influenced voters of high SES more than those of low SES.

General Discussion

Four studies extend previous work on the role of perceived competence when evaluating politicians by showing that the preference for competence depends on voters' social class.

More concretely, perceived competence appears to be more important for higher than for lower class voters. In Study 1, voters of higher SES rated competence in a politician as more important compared to voters with of SES. This result complements previous findings (Callaghan et al., 2022; Laustsen & Bor, 2017). Observing this effect in the German context which is characterized by a lower economic inequality and thus potentially smaller social class effects compared to the USA speaks to the robustness of a link between social class and a preference for competence. Moreover, the effect does not seem to be limited to an extremely individualistic society as the USA.

Whereas Study 1 provides evidence from a representative sample based on explicit ratings, Studies 2 and 3 go one step further and use a more implicit method by assessing the voting likelihood for politicians varying in competent appearance based on looks. According to the results, politicians' perceived competence has a larger influence on voting likelihood among higher-class voters. This was the case even when controlling for other impressions elicited by the faces, namely warmth and dominance. Thus, our research provides first evidence that social classes do not only differ in what they say they find important in a politician but also in what they are actually considering, i.e., less controlled responses.

A potential mechanism underlying the preference for competence among higher-class voters is voters' self-perception. We argue that high competence is more likely part of higher (vs. lower) social class people's self-schema. Given that the self-schema guides the perception and evaluation of others (Carpenter, 1988; Fong & Markus, 1982; Green & Sedikides, 2001; Riggs & Cantor, 1984) people of higher social class should perceive and judge others according to competence cues. Moreover, people generally prefer politicians who are similar to them (Caprara et al., 2007). Study 1 supported our assumptions. Not only did participants of higher SES perceive themselves as more competent this self-concept fully mediated the effect of SES on the weighting of competence. Furthermore, exploratory analyses for Studies 2a/b and 3 (see Supplement (5) for details) indicated that people of higher SES did not only place a larger weight on perceived competence but, independent of this preference, also differentiated more between politicians with high and low competence than participants of lower SES. The finding that higher-class voters perceived politicians' competence in a more nuanced way than lower-class voters suggests that they are more sensitive for competencerelated cues in politicians' appearance. This supports the assumption of higher-class people being competence schematics which makes them attend more to this trait when judging others (Fong & Markus, 1982).

A comparison of Studies 2a/b and 3 suggests that the effect appears to be stronger when competence is made salient beforehand. Apparently drawing perceivers' attention to competence has different effects depending on social class. We acknowledge that further evidence is needed. But even if the effects were limited to prior activation of competence as in Studies 2a/b this may be closer to real world situations than the absence of prior activation. It seems unlikely that voters form their first impression of a candidate in the voting booth. Rather following a candidate's public appearances and media coverage makes it likely that an impression of competence is explicitly activated. Accordingly, the increased accessibility of competence elicited by the question order in Study 2a/b might resemble the actual psychological processes when evaluating politicians.

Whereas we found converging evidence for the effect of SES the results do not support similar conclusions for SSS. When explicitly asked (Study 1) high SSS increased the relevance of competence but also of all other traits and the self-ratings on these traits rendering inflated effects due to common method variance likely (see Tan et al, 2020). This would imply that the higher relevance of competence among people high in SSS is an artefact and does not reflect similar processes as in people high in SES. Likewise, the lack of effects of SSS on the more subtle measure of Studies 2a-3 speaks against the functional equivalence of SSS and SES in this regard. This is also in line with previous research that did not find effects of manipulated SSS on the relevance of politicians' competence (Callaghan et al., 2022). Our assumption of people of high SES being competence schematics was based on the fact that in their environment that is characterized by educational and financial achievement competence plays an omnipresent role and becomes important when judging the self as well as others. To the extent that SSS represents similar socialization processes its effects should be similar. However, SSS is more dependent on the current context in which it is assessed (Destin et al., 2017) and potentially less closely linked with socialization processes. Future research may shed light on the divergent findings on SES and SSS.

Our main result has implications for the well-established effect of perceived competence on electoral outcomes (e.g., Todorov et al., 2005). Generally, members of higher social class show a higher level of political participation (e.g., Kraus et al., 2015). Also, in our studies higher class participants were more likely to have voted in the last German federal election (Study 1: r = .22, p < .001) and reported to vote more regularly (Study 2a/b: r = .18, p < .001, Study 3: r = .19, p < .001). Hence, higher class voters might partially drive the effect of perceived competence on electoral success through their relatively higher turnout rate. For political actors, it appears worthwhile to consider the social class of the target audience in the campaign. Focusing on competence may be a promising strategy for winning voters of higher social class but may not pay off among voters of lower social class.

As our research focused on competence the stimulus material maximized variance on competence and any conclusions regarding other traits can only be preliminary. With this in mind, we summarize that in none of the studies did we observe a class difference on the importance of warmth. Although research reports a warmth stereotype for lower social status groups (Durante et al., 2017), based on which one may expect that warmth is more important to voters of lower social class, it should be noted that this stereotype is not observed in all countries (Durante et al., 2013). More crucially, our assumptions are based on self-stereotypes, and we did not find that people of lower SES think of themselves as warmer than people of high SES.

Limitations and Future Directions. As voting plays an essential role for the functioning of democracies, it is important to examine which factors influence voting decisions. The current studies add to research on how interpersonal perception influences electoral outcomes. However, we acknowledge some limitations.

Firstly, the characteristics of the politicians' faces that were used in Studies 2a-3 limit the generalizability of our results. Specifically, we used pictures of white¹¹, male politicians. Specifically, the weight attached to perceived competence might be different for male and female politicians. It has been shown that competence-related information is more important when evaluating female candidates (Ditonto, 2017). Also, none of the candidates depicted was characterized by an extreme lack of perceived competence. This might be due to the fact that these politicians were part of the Swiss parliament and, hence, already had had some electoral success. Future research should investigate the role of social class for weighting competence when evaluating politicians of different gender and ethnic background as well as politicians with a greater lack of perceived competence.

Secondly, the samples in Study 2a-3 were not representative of all voters as they did not include people from the extreme ends of the social class spectrum. Especially people with low education did not participate in these studies. This, however, suggests that the weighting of perceived competence which we found despite the restricted variance in SES might be even larger in real-world settings.

Thirdly, the societal and political context of our studies always poses a limitation to our research. Investigating social class effects with German samples supports the generalizability of previously found effects in the USA context, but still more studies with samples from different societies are needed to generalize the results. In fact, assuming that competence is more valued in individualistic than collectivistic societies and class differences manifest more on traits generally regarded highly in a society (Gobel & Miyamoto, 2022) one may expect a divergent pattern for collectivistic societies.

As a further issue, as we cannot manipulate SES we cannot determine whether the observed effect is not due to a third variable. Nevertheless, our hypothesis was based on assumptions about differences in the self-concept between people of higher and lower SES and we found evidence for this assumed mediating process.

Conclusion

The present research underlines the importance for politicians of being perceived as competent. But perceived competence – even when only based on facial appearance – appears more important for voters of higher compared to lower social class. The reason for this difference seems to lie in systematic differences in the self-concept of members of higher and lower class. Generally, social class has only recently been discovered as predictor for fundamental psychological processes (e.g., Kraus et al., 2012) like person perception. As social class has an omnipresent influence on a person's identity it is high time psychological research pays more attention to its impact.

Appendix A. Supplementary Material

Supplementary material to this article can be found in the online appendix.

¹¹ As are over 99% of the members of the German parliament (as of 2023).

References

- Abele, A. E. (2003). The dynamics of masculine-agentic and feminine-communal traits: Findings from a prospective study. *Journal of Personality and Social Psychology*, 85(4), 768–776. https://doi.org/10.1037/0022-3514.85.4.768
- Abele, A. E., Hauke, N., Peters, K., Louvet, E., Szymkow, A., & Duan, Y. (2016). Facets of the fundamental content dimensions: Agency with competence and assertivenesscommunion with warmth and morality. *Frontiers in Psychology*, 7, 1810. https://doi.org/10.3389/fpsyg.2016.01810
- Abele, A. E., & Wojciszke, B. (2007). Agency and communion from the perspective of self versus others. *Journal of Personality and Social Psychology*, 93(5), 751–763. https://doi.org/10.1037/0022-3514.93.5.751
- Abele, A. E., & Wojciszke, B. (2014). Communal and agentic content in social cognition. A dual perspective model. *Advances in Experimental Social Psychology*, 50(50), 195– 255. https://doi.org/10.1016/B978-0-12-800284-1.00004-7
- Adler, N. E., Epel, E. S., Castellazzo, G., & Ickovics, J. R. (2000). Relationship of subjective and objective social status with psychological and physiological functioning: Preliminary data in healthy, White women. *Health Psychology*, *19*(6), 586–592. https://doi.org/10.1037/0278-6133.19.6.586
- Antonakis, J., & Dalgas, O. (2009). Predicting elections: Child's play! *Science (New York, N.Y.)*, 323(5918), 1183. https://doi.org/10.1126/science.1167748
- Arend, M. G., & Schäfer, T. (2019). Statistical power in two-level models: A tutorial based on Monte Carlo simulation. *Psychological Methods*, 24(1), 1–19. https://doi.org/10.1037/met0000195
- Ballew, C. C., & Todorov, A. (2007). Predicting political elections from rapid and unreflective face judgments. *Proceedings of the National Academy of Sciences of the United States of America*, 104(46), 17948–17953. https://doi.org/10.1073/pnas.0705435104
- Bates, D., Mächler, M., Bolker, B., & Walker, S. (2015). Fitting linear mixed-effects models using lme4. (Version 1.1-29). *Journal of Statistical Software*, 67(1), 1–48. https://doi.org/10.18637/jss.v067.i01
- Belmi, P., Neale, M. A., Reiff, D., & Ulfe, R. (2020). The social advantage of miscalibrated individuals: The relationship between social class and overconfidence and its implications for class-based inequality. *Journal of Personality and Social Psychology*, *118*(2), 254–282. https://doi.org/10.1037/pspi0000187
- Boileau, L. L.-A. (2022). Social class and the big two: Investigating a widespread yet understudied assumption on social class differences in self-concepts [Doctoral dissertation]. University of Mannheim.
- Callaghan, B., Kraus, M. W., & Dovidio, J. F. (2022). Social class predicts preference for competent politicians. *Journal of Experimental Social Psychology*, 100, 104298. https://doi.org/10.1016/j.jesp.2022.104298
- Caprara, G. V., Vecchione, M., Barbaranelli, C., & Fraley, R. C. (2007). When likeness goes with liking: The case of political preference. *Political Psychology*, *28*(5), 609–632. https://doi.org/10.1111/j.1467-9221.2007.00592.x

- Caprara, G. V., & Zimbardo, P. G. (2004). Personalizing politics: A congruency model of political preference. *The American Psychologist*, 59(7), 581–594. https://doi.org/10.1037/0003-066X.59.7.581
- Carpenter, S. L. (1988). Self-relevance and goal-directed processing in the recall and weighting of information about others. *Journal of Experimental Social Psychology*, 24(4), 310–332. https://doi.org/10.1016/0022-1031(88)90023-6
- Cheng, J. T., Tracy, J. L., Foulsham, T., Kingstone, A., & Henrich, J. (2013). Two ways to the top: Evidence that dominance and prestige are distinct yet viable avenues to social rank and influence. *Journal of Personality and Social Psychology*, 104(1), 103–125. https://doi.org/10.1037/a0030398
- Destin, M., Rheinschmidt-Same, M., & Richeson, J. A. (2017). Status-based identity: A conceptual approach integrating the social psychological study of socioeconomic status and identity. *Perspectives on Psychological Science*, *12*(2), 270–289. https://doi.org/10.1177/1745691616664424
- Ditonto, T. (2017). A high bar or a double standard? Gender, competence, and information in political campaigns. *Political Behavior*, *39*(2), 301–325. https://doi.org/10.1007/s11109-016-9357-5
- Durante, F., Fiske, S. T., Kervyn, N., Cuddy, A. J. C., Akande, A. D., Adetoun, B. E.,
 Adewuyi, M. F., Tserere, M. M., Ramiah, A. A., Mastor, K. A., Barlow, F. K.,
 Bonn, G., Tafarodi, R. W., Bosak, J., Cairns, E., Doherty, C., Capozza, D.,
 Chandran, A., Chryssochoou, X., . . . Storari, C. C. (2013). Nations' income inequality
 predicts ambivalence in stereotype content: How societies mind the gap. *The British Journal of Social Psychology / the British Psychological Society*, *52*(4), 726–746.
 https://doi.org/10.1111/bjso.12005
- Durante, F., Tablante, C.B., & Fiske, S.T. (2017), Poor but warm, rich but cold (and competent): Social classes in the Stereotype Content Model. *Journal of Social Issues*, 73, 138-157. https://doi.org/10.1111/josi.12208
- Enders, C. K., & Tofighi, D. (2007). Centering predictor variables in cross-sectional multilevel models: A new look at an old issue. *Psychological Methods*, 12(2), 121– 138. https://doi.org/10.1037/1082-989X.12.2.121
- Eriksson, K. (2018). Republicans value agency, democrats value communion. *Social Psychology Quarterly*, *81*(2), 173–184. https://doi.org/10.1177/0190272518771060
- Faul, F., Erdfelder, E., Buchner, A., & Lang, A.-G. (2009). Statistical power analyses using G*Power 3.1: Tests for correlation and regression analyses. *Behavior Research Methods*, 41(4), 1149–1160. https://doi.org/10.3758/BRM.41.4.1149
- Fiske, S. T., Cuddy, A. J. C., & Glick, P. (2007). Universal dimensions of social cognition: warmth and competence. *Trends in Cognitive Sciences*, 11(2), 77-83. https://doi.org/10.1016/j.tics.2006.11.005.
- Fiske, S. T., Cuddy, A. J. C., Glick, P., & Xu, J. (2002). A model of (often mixed) stereotype content: Competence and warmth respectively follow from perceived status and competition. *Journal of Personality and Social Psychology*, 82(6), 878–902. https://doi.org/10.1037//0022-3514.82.6.878
- Fong, G. T., & Markus, H. (1982). Self-Schemas and Judgments about Others. Social Cognition, 1(3), 191–204. https://doi.org/10.1521/soco.1982.1.3.191

- Funk, C. L. (1999). Bringing the Candidate into Models of Candidate Evaluation. *The Journal of Politics*, 61(3), 700–720. https://doi.org/10.2307/2647824
- German Federal Statistical Office (Destatis) (2022). *Bevölkerung und Erwerbstätigkeit. Haushalte und Familien. Ergebnisse des Mikrozensus. Erstergebnisse 2021. Tabelle 1.1.* https://www.destatis.de/DE/Themen/Gesellschaft-Umwelt/Bevoelkerung/Haushalte-Familien/Publikationen/Downloads-Haushalte/haushalte-familien-2010300217004.pdf?__blob=publicationFile
- Gobel, M. S., & Miyamoto, Y. (2022, January 29). Self- and other-orientation in high rank: A cultural psychological approach to social hierarchy. https://doi.org/10.31234/osf.io/k8npc
- Goh, J. X., Hall, J. A., & Rosenthal, R. (2016). Mini meta-analysis of your own studies: Some arguments on why and a primer on how. *Social and Personality Psychology Compass*, 10(10), 535–549. https://doi.org/10.1111/spc3.12267
- Green, P., & MacLeod, C. J. (2016). Simr: An R package for power analysis of generalized linear mixed models by simulation. (Version 1.0.6). *Methods in Ecology and Evolution*, 7(4), 493–498. https://doi.org/10.1111/2041-210X.12504
- Green, J. D., & Sedikides, C. (2001). When do self-schemas shape social perception? The role of descriptive ambiguity. *Motivation and Emotion*, 25(1), 67–83. https://doi.org/10.1023/A:1010611922816
- Hofstede, G., Hofstede, G. J., & Minkov, M. (2010). *Cultures and Organizations Software of the mind: Intercultural cooperation and its importance for survival.* 3rd edition. McGraw-Hill.
- Kinder, D. R., Peters, M. D., Abelson, R. P., & Fiske, S. T. (1980). Presidential prototypes. *Political Behavior*, 2(4), 315–337. https://doi.org/10.1007/BF00990172
- Kraus, M., Anderson, C., & Callaghan, B. (2015). The inequality of politics: Social class rank and political participation. SSRN Electronic Journal. Advance online publication. https://doi.org/10.2139/ssrn.2600107
- Kraus, M. W., & Keltner, D. (2009). Signs of socioeconomic status: A thin-slicing approach. *Psychological Science*, 20(1), 99–106. https://doi.org/10.1111/j.1467-9280.2008.02251.x
- Kraus, Piff, P. K., Mendoza-Denton, R., Rheinschmidt, M. L., & Keltner, D. (2012). Social class, solipsism, and contextualism: How the rich are different from the poor. *Psychological Review*, 119(3), 546–572. https://doi.org/10.1037/a0028756
- Kraus, M. W., Tan, J. J. X., & Tannenbaum, M. B. (2013). The social ladder: A rank-based perspective on social class. *Psychological Inquiry*, *24*, 81–96.
- Laustsen, L., & Bor, A. (2017). The relative weight of character traits in political candidate evaluations: Warmth is more important than competence, leadership and integrity. *Electoral Studies*, 49, 96–107. https://doi.org/10.1016/j.electstud.2017.08.001
- Leiner, D. J. (2019). SoSci Survey (Version 3.1.06) [Computer software]. Available at https://www.soscisurvey.de
- Long J. A. (2019). *interactions: Comprehensive, user-friendly toolkit for probing interactions*. R package version 1.1.0. (Version 1.1.5). https://cran.rproject.org/package=interactions.

- Markus, H. (1977). Self-schemata and processing information about the self. *Journal of Personality and Social Psychology*, 35(2), 63–78. https://doi.org/10.1037/0022-3514.35.2.63
- Mignon, A., Mollaret, P., Rohmer, O., & Bagès, C. (2016). Effect of political orientation on judgment of agency, competence, morality, and sociability: The French presidential election of 2012. Swiss Journal of Psychology, 75(1), 35–45. https://doi.org/10.1024/1421-0185/a000169
- Miller, A. H., Wattenberg, M. P., & Malanchuk, O. (1986). Schematic Assessments of Presidential Candidates. *American Political Science Review*, 80(2), 521–540. https://doi.org/10.2307/1958272
- Oldmeadow, J. A., & Fiske, S. T. (2010). Social status and the pursuit of positive social identity: Systematic domains of intergroup differentiation and discrimination for highand low- status groups. *Group Processes & Intergroup Relations: GPIR*, *13*(4). https://doi.org/10.1177/1368430209355650
- Olivola, C. Y., & Todorov, A. (2010a). Elected in 100 milliseconds: Appearance-Based Trait Inferences and Voting. *Journal of Nonverbal Behavior*, *34*(2), 83–110. https://doi.org/10.1007/s10919-009-0082-1
- Olivola, C. Y., & Todorov, A. (2010b). Fooled by first impressions? Reexamining the diagnostic value of appearance-based inferences. *Journal of Experimental Social Psychology*, 46(2), 315–324. https://doi.org/10.1016/j.jesp.2009.12.002
- Page, B. I., Bartels, L. M., & Seawright, J. (2013). Democracy and the policy preferences of wealthy Americans. *Perspectives on Politics*, 11(1), 51–73. http://doi.org/10.1017/S153759271200360X
- Paskov, M., Gërxhani, K., & van de Werfhorst, H. G. (2013). Income inequality and status anxiety. *GINI Discussion Paper*, *90*, 1–46.
- Rahal, D., Fales, M. R., Haselton, M. G., Slavich, G. M., & Robles, T. F. (2021). Cues of social status: Associations between attractiveness, dominance, and status. *Evolutionary psychology : an international journal of evolutionary approaches to psychology and behavior*, *19*(0), 14747049211056160. https://doi.org/10.1177/14747049211056160
- Riggs, J. M., & Cantor, N. (1984). Getting acquainted. *Personality and Social Psychology Bulletin*, 10(3), 432–445. https://doi.org/10.1177/0146167284103012
- Rosseel, Y. (2012). Lavaan: An R package for structural equation modeling. (Version 0.6-12). *Journal of Statistical Software*, 48(2), 1–36. https://doi.org/10.18637/jss.v048.i02
- Schneider, S.M. (2019). Why income inequality is dissatisfying—Perceptions of social status and the inequality-satisfaction link in Europe. *European Sociological Review*, 35, 3, 409–430. https://doi.org/10.1093/esr/jcz003
- Sedikides, C., & Skowronski, J. J. (1993). The self in impression formation: Trait centrality and social perception. *Journal of Experimental Social Psychology*, 29(4), 347–357. https://doi.org/10.1006/jesp.1993.1016
- Shoda, T. M., & McConnell, A. R. (2013). Interpersonal sensitivity and self-knowledge: Those chronic for trustworthiness are more accurate at detecting it in others. *Journal* of Experimental Social Psychology, 49(3), 440–443. https://doi.org/10.1016/j.jesp.2012.12.011

- Stephens, N. M., Markus, H. R., & Phillips, L. T. (2014). Social class culture cycles: How three gateway contexts shape selves and fuel inequality. *Annual Review of Psychology*, 65, 611–634. https://doi.org/10.1146/annurev-psych-010213-115143
- Tan, J. J. X., & Kraus, M. W. (2018). Judgments of interpersonal warmth predict class differences in political candidate support. *Social Cognition*, 36(1), 106–133. https://doi.org/10.1521/soco.2018.36.1.106
- Tan, J. J. X., Kraus, M. W., Carpenter, N. C., & Adler, N. E. (2020). The association between objective and subjective socioeconomic status and subjective well-being: A metaanalytic review. *Psychological Bulletin*, 146(11), 970–1020. https://doi.org/10.1037/bul0000258
- Todorov, A., Mandisodza, A. N., Goren, A., & Hall, C. C. (2005). Inferences of competence from faces predict election outcomes. *Science*, 308(5728), 1623–1626. https://doi.org/10.1126/science.1110589
- Todorov, A., Olivola, C. Y., Dotsch, R., & Mende-Siedlecki, P. (2015). Social attributions from faces: Determinants, consequences, accuracy, and functional significance. *Annual Review of Psychology*, 66, 519–545. https://doi.org/10.1146/annurev-psych-113011-143831
- World Bank, World Development Indicators (2023). *Gini index* [Data file]. https://data.worldbank.org/indicator/SI.POV.GINI?end=2018&start=2015.
- Zhang, H., Talhelm, T., Yang, Q., & Hu, C. S. (2021). High-Status people are more individualistic and analytic-thinking in the west and wheat-farming areas, but not ricefarming areas. *European Journal of Social Psychology*, 51, 878–895. https://doi.org/10.1002/ejsp.2778

TITLE: Jumping on the Bandwagon: The Role of Voters' Social Class in Poll Effects in the Context of the 2021 German Federal Election

The version of record of this article, first published in *Politische Vierteljahresschrift*, is available online at Publisher's website: https://doi.org/10.1007/s11615-022-00417-3

PVS Politische Vierteljahresschrift German Political Science Quarterly

CRITICAL PAPER

Jumping on the Bandwagon: The Role of Voters' Social Class in Poll Effects in the Context of the 2021 German Federal Election

Fabienne Unkelbach D · Melvin John · Vera Vogel

Received: 25 September 2021 / Revised: 25 May 2022 / Accepted: 4 July 2022 / Published online: 10 August 2022 © The Author(s) 2022

Abstract Published findings of opinion polls are an important part of the political coverage before elections. Thus, researchers have long investigated whether the perceived popularity of political parties can lead to even more voters following this majority. However, empirical findings on this so-called political bandwagon effect are mixed. In the present paper, we integrate theories from political science and social psychology to explain these inconsistencies through social class as a potential moderating variable. Based on previous findings regarding consumer decisions, we hypothesized that bandwagon effects are greater among voters with lower social class. To investigate this hypothesis, we combined data from the German Longitudinal Election Study (GLES) Rolling Cross-Section 2021, which was collected over the 55 days before the 2021 German federal election, with the results of published preelection polls. Using separate multilevel models for each of the parties, we found no evidence for bandwagon effects. Only for the Social Democratic Party were poll results related to voting intentions assessed on the following day, suggesting that polls might have contributed to the party's electoral success. However, there was no evidence for a moderation of bandwagon effects by voters' social class. Accordingly, we could not resolve the mixed findings in this field of research. Our results point to important open questions in research on bandwagon effects in multiparty systems as well as on effects of social class in Germany.

Melvin John Chair of Microsociology and Social Psychology, University of Mannheim, Mannheim, Germany

Melvin John · Vera Vogel Mannheim Centre for European Social Research (MZES), University of Mannheim, Mannheim, Germany

Fabienne Unkelbach (⊠)

Chair of Consumer and Economic Psychology, University of Mannheim, A5, 6, 68159 Mannheim, Germany

E-Mail: funkelba@mail.uni-mannheim.de

Keywords Mass media \cdot Social influence \cdot Social psychology \cdot Voting intention \cdot Rolling cross-section study

Wer springt auf den *Bandwagon* auf? Die Rolle des sozialen Status der Wählenden für Umfrageeffekte im Kontext der Bundestagswahl 2021

Zusammenfassung Ergebnisse von Meinungsumfragen sind ein wichtiger Bestandteil der politischen Berichterstattung vor Wahlen. Daher wird seit Langem untersucht, ob die aufgrund von Umfrageergebnissen wahrgenommene Popularität politischer Parteien dazu führen kann, dass sich Wählende der scheinbaren Mehrheit anschließen. Die empirischen Ergebnisse zu diesem sogenannten politischen Bandwagon-Effekt sind jedoch gemischt. Im vorliegenden Artikel haben wir Theorien aus der Politikwissenschaft und der Sozialpsychologie integriert, um diese Unstimmigkeiten durch den sozialen Status als mögliche moderierende Variable zu erklären. Auf der Grundlage früherer Erkenntnisse zu Konsumentscheidungen stellten wir die Hypothese auf, dass Bandwagon-Effekte bei Wählenden mit niedrigerem sozialem Status größer ausfallen. Zur Untersuchung dieser Hypothese kombinierten wir Daten der German Longitudinal Election Study (GLES) Rolling Cross-Section-Wahlkampfstudie 2021, die in den letzten 55 Tagen vor der Bundestagswahl 2021 durchgeführt wurde, mit Umfrageergebnissen, die in diesem Zeitraum veröffentlicht wurden. In separaten Mehrebenenmodellen für jede der Parteien zeigte sich keine Evidenz für Bandwagon-Effekte. Lediglich für die SPD standen die Umfrageergebnisse in Zusammenhang mit der am Folgetag berichteten Wahlabsicht, was darauf hindeutet, dass die Umfragen zum Wahlerfolg der Partei beigetragen haben könnten. Wir fanden jedoch keine Evidenz für eine Moderation der Bandwagon-Effekte durch den sozialen Status der Wählenden. Dementsprechend konnten wir die gemischten Ergebnisse in diesem Forschungsfeld nicht auflösen. Unsere Ergebnisse weisen auf wichtige offene Fragen in der Forschung zu Bandwagon-Effekten in Mehrparteiensystemen sowie zu Effekten des sozialen Status in Deutschland hin.

Schlüsselwörter Massenmedien · Sozialer Einfluss · Sozialpsychologie · Wahlabsicht · Rolling Cross-Section Wahlkampfstudie

1 Introduction

In modern democracies and especially in times of elections, people constantly have the opportunity to get information from the mass media on others' political opinions and voting preferences. One of the most important sources of such information consists of published findings from public opinion polls (Moy and Rinke 2012). Over the past decades, polls have become an increasingly important part of the political coverage (Brettschneider 2008). For example, in the last 5 weeks before the German federal election in 2013, the proportion of people who reported to have paid attention to such polls rose drastically and reached approximately 70% just before the election (Partheymüller and Schäfer 2013). This trend is related to a pervasive tendency toward so-called horse-race journalism, which focuses more on the question of who's ahead than on substantive issues during the election campaign (Genz et al. 2001). While the great presence of polls in the media has long been critically debated, it became especially controversial during the 2021 German federal election campaign when a polling institute reported projections in the preelection phase including data from postal voters on their already cast votes (Thiel 2021).

Different kinds of poll effects have been distinguished. For example, becoming aware of public opinion via polls can elicit *strategic voting* behavior (i.e., people vote for a party/candidate because of strategic reasons even though it is not their preferred choice) as well as the so-called *underdog effect* (i.e., people favor parties/ candidates who are behind in polls) (Moy and Rinke 2012). Furthermore, it has been argued that the perceived popularity of political parties and candidates can result in the so-called *political bandwagon effect* (Moy and Rinke 2012). This refers to the phenomenon in which some people tend to follow the perceived majority and vote for candidates, parties, or political opinions that are ahead in the polls (Schmitt-Beck 2015). Thus, bandwagon effects can be understood as an instance of majority influence in the political context.

In the present research, we argue that voters' social class is a possible moderator variable of bandwagon effects. Recent findings from social psychology suggest that majority influence is stronger among individuals with lower (vs. higher) social class in other choice contexts (Na et al. 2016; Stephens et al. 2011). By transferring this finding to the political context, we investigated whether voters' social class moderates the effect of a majority party preference (as perceived via public poll results) on voters' own vote intention, i.e. the political bandwagon effect.

We examine this research question in the context of the 2021 German federal election because it provides an extraordinarily promising context to investigate poll effects for several reasons. First, this was the first federal election in the postwar history of Germany without the incumbent chancellor running. Due to this political change, preelection poll results were characterized by great dynamics before the 2021 election. This is a prerequisite for adequate statistical power to examine the existence of a bandwagon effect and voters' social class as its assumed boundary condition. For example, in August 2021 the Social Democratic Party (SPD) was ahead of the Christian Democratic Union (CDU) in poll results for the first time in almost 15 years (Grahn and Süßmann 2021). Second, the COVID-19 crisis has made existing social disparities especially salient. Contradicting its initially assumed role as a "great equalizer" (Cuomo 2020), studies suggest that the pandemic has instead increased existing social inequalities by affecting people of different social classes to a different extent (Bundeszentrale für politische Bildung 2021). Based on data from the German Longitudinal Election Study (GLES) Rolling Cross-Section 2021 (GLES 2022), the present research makes an important contribution to a) the controversy about polls as an integral part of the political news coverage and b) the profound understanding of consequences of social class differences.

2 Theoretical Argument, Literature Review, and Hypothesis

2.1 Majority Influence in the Political Context: The Political Bandwagon Effect

In the political context, voters do not only develop their own voting intention but also form an impression of others' preferences. Importantly, beliefs about the general electorate's party support are influenced by polls spread by the mass media and have the potential to influence individuals' voting preferences (e.g., Moy and Rinke 2012). Poll effects based on the tendency to follow a perceived majority opinion have often been investigated under the term bandwagon effect. It especially refers to the influence of preelection polls on individuals' voting preferences in the sense that support for the view presented as being favored by a majority of society increases (Barnfield 2020; Schmitt-Beck 2015). Thus, bandwagon effects can affect individuals' attitudes toward political issues, parties, and candidates as well as actual voting behavior (Moy and Rinke 2012). Thereby, polls constitute the most visible signal of majority support. A consequence of bandwagon effects is that the perceived public opinion can turn into a self-fulfilling prophecy (Schmitt-Beck 2015).

Different theoretical accounts have been used to explain why voters jump on the bandwagon (for an overview, see Hardmeier 2008). In the following, we will briefly highlight the most influential approaches. From a social psychological perspective, bandwagon effects have often been understood as a manifestation of *conformity*. In his seminal study, Asch (1956) found that individuals conform to the opinion of a majority of people surrounding them even when the majority position is clearly incorrect. Advancing these findings, studies showed that the perception that a majority of others evaluate something positively leads to individuals evaluating this attitude or object more positively, too (e.g., Erb et al. 1998). People thus follow a consensus heuristic that implies that what a majority is doing must be the preferable option (Erb and Bohner 2010).

Aside from psychological concepts used to explain the mechanisms of bandwagon effects, research from political science has long emphasized the role of so-called *impersonal influence* (Mutz 1998). This construct describes the effect of information about the beliefs of collectives of others who are not part of an individual's personal contacts. In this regard, bandwagon effects have long been tied to a so-called gratification mechanism that refers to voters switching to the "winning side" solely because of the expected gratification of belonging to the "winners" (Mutz 1998).

Even though many potential causal mechanisms and conceptualizations of majority influence have been discussed, there is still a lack of empirical evidence for a conclusive model of political bandwagon effects (Schmitt-Beck 2015). Indeed, it is conceivable that the concept of bandwagon effects is not inherently linked with one of the proposed mechanisms but that different mechanisms might be in play (Barnfield 2020).

Empirical research on bandwagon effects includes a variety of studies that differ in their design and political setting. However, there are two main aspects that can be used to categorize these studies (for an overview, see Barnfield 2020). First, bandwagon effects have been investigated with regard to two outcomes: the switch in vote choice from one alternative to another (conversion) and a decision to turn out
to vote (mobilization) (Barnfield 2020; Morton et al. 2015). Because most previous studies have investigated bandwagon effects on conversion, we will also focus on these effects when speaking of bandwagon effects.

Second, studies on bandwagon effects have differed in their independent variable—the aspect of opinion polls that influences voters' preferences. While there is some evidence for the effect of a rise of public support of a candidate's respective party from one time point to the second one (dynamic bandwagon effects; e.g., van der Meer et al. 2016), most research has focused on the (leading) position in poll results of one party in comparison to others at one point in time (static bandwagon effects; e.g., Schmitt-Beck 1996).

In addition to these conceptual differences, the political context plays an important role in investigations on bandwagon effects. The concept of leading in the polls is well applicable to first-past-the-post systems like that of the United States, and most evidence on bandwagon effects stems from presidential primaries in the United States (e.g., Callander 2007). But what does success/leading in the polls mean in the German multiparty context with proportional representation? Recent research suggests that the definition of success, or of being a "winner," in opinion polls is more ambiguous in multiparty contexts with proportional representation systems (cf. Barnfield 2020). Due to the common formation of coalition governments, "winning" an election is not limited to being the party with the largest vote share (Meffert et al. 2011). Recent research in proportional representation contexts has emphasized that there are several aspects that can lead to a party's being portrayed and perceived as a winner, e.g., exceeding expectations, getting the largest number of seats, overcoming the electoral threshold (Stolwijk et al. 2016), or being predicted with high certainty to be part of a government (Riambau 2018). Thus, both large and small parties might justifiably claim to be successful in their election campaigns (Hardmeier and Roth 2003). Accordingly, we investigated potential poll effects for all major German parties to be able to get a complete picture.

In the German context, there is first-hand evidence that interest in political media coverage a) influences voters' expectations regarding the election outcome and b) consequentially leads to voters choosing the party that was leading in the polls in the 1990 national election (Schmitt-Beck 1996). Similar results could be found for the 2005 parliamentary election in a study based on data from a rolling cross-section (RCS) study (Faas et al. 2008). However, a reanalysis of the GLES RCS data for the 2005 election with a different analysis approach¹ suggested that the polls had an influence only on voting turnout intention and coalition expectations but not on voting intentions for the different parties (Hoffmann and Klein 2013). Accordingly, we concluded that further investigations on poll effects in the German context were necessary.

Whereas previous research has explained the mixed empirical evidence for the political bandwagon effect through methodological issues and a lack of conceptual clarity (cf. Barnfield 2020), we argue that a third reason might play a role: Potential moderating variables of majority influence could influence the size of the political

¹ The authors used a multilevel modeling approach, a different selection of published poll results, and a different time-lag between the publishing of polls and the survey data collection.

bandwagon effect. Importantly, recent social psychological findings suggest that a sociodemographic characteristic of individuals, namely their social class, is related to the tendency to follow a perceived majority. We, thus, focused on this variable as a potential moderator of bandwagon effects in the present research.

2.2 Social Class and Susceptibility to Social Influence

In the last decade, psychological perspectives on social class have been developed: a so-called social cognitive perspective (Kraus et al. 2012) and a cultural approach (Stephens et al. 2014). Whereas the first one focuses on the way different material resources can influence basic psychological tendencies, the second one is based on the notion that people's social class is an important determinant for the sociocultural contexts in which people spend most of their lives. Within these frameworks, objective components of social class (objective socioeconomic status [SES]) refer to a person's level of access to (material) resources and are often measured via one's educational attainment and financial means, as well as occupational prestige (Oakes and Rossi 2003). By contrast, subjective components of social status are defined as a person's perception of their relative standing in society, which is derived from the comparison of one's material wealth to those of others (e.g., Kraus et al. 2012, 2017). It is assumed that objective SES and subjective social status (SSS) constitute important factors of social class contexts that influence people's experiences of being from a certain social class background and refer to different aspects of social class (Adler et al. 2000; Kraus et al. 2012). Importantly, research from both theoretical approaches on social class shows that the material conditions in which people are raised and live influence a range of psychological and behavioral outcomes (Kraus et al. 2012; Manstead 2018).

First, people of lower and higher social class differ in their self-concept (Kraus et al. 2012; Manstead 2018). A self-concept is defined as a person's thoughts, beliefs, and feelings about the self as an object (Markus and Kitayama 1991, 2010). More concretely, there is first evidence that people of lower social class are more likely to develop an *interdependent* self (Grossmann and Varnum 2011). This means that they tend to focus on their self as embedded in social relationships, a focus that is often explained by their experience of constrained opportunities for making free life choices. In contrast, people of a higher social class are more likely to develop an *independent* self-concept (Kraus et al. 2012). This means that they tend to see themselves as separate, unique entities, and this tendency has often been explained by their larger freedom to make life decisions according to their own interests and desires.

Second, these social class differences in self-concepts are reflected in the way people make choices under social influence. Individuals from lower social classes are relatively more likely to make choices that promote similarity to and connection with others, whereas individuals from higher social classes are relatively more likely to make choices that produce uniqueness and differentiation from others (Na et al. 2016; Stephens et al. 2007, 2011). For example, it was found that participants with lower objective SES (as classified via their parents' educational attainment) more often chose the same product as an ostensible former participant and liked their

chosen product more when another person apparently made the same choice (vs. a different choice) (Stephens et al. 2007, studies 2 and 3). However, the choice of another person had no influence on the ratings of the chosen product for participants with higher objective SES.

Additionally, people of lower social class are more susceptible to a perceived majority preference when it comes to product choices (Na et al. 2016). It could be demonstrated that mainly participants of lower social class aligned their product choices with a majority's preference even when that choice contradicted their personal preferences (Na et al. 2016, study 1). Importantly, this effect was found using objective SES as well as SSS. First evidence suggests that these social class differences in the sensitivity to preferences of others are indeed mediated by their independent vs. interdependent self-concepts (Na et al. 2016, studies 2 and 3).

Thus, whereas majority influence has often been considered a well-established phenomenon, recent studies suggest that these effects cannot be found uniformly across all social classes. In the present research, we transfer these findings to the political context and, more concretely, to the political bandwagon effect.

2.3 Current Research: The Role of Social Class in Bandwagon Effects

Even though multiple studies on bandwagon effects have been conducted over the past decades, empirical evidence is mixed. One potential explanation for these inconsistencies might be the presence of moderator effects. While the political system has often been considered an important context variable, few studies so far have focused on voters' characteristics as moderators of the size of bandwagon effects. This, however, is important not only to better understand the relevance of bandwagon effects for voters' behavior but also to arrive at a better understanding of poll influences overall.

In the present research, we argue that recent social psychological findings on the relationship between people's social class and their tendency to follow a majority can offer new insights into boundary conditions of bandwagon effects. More concretely, people with lower objective SES and SSS have been found to be more likely to align their product choices with social preferences (Na et al. 2016; Stephens et al. 2011). Research suggests that this can be explained by an interdependent self-concept and a stronger focus on external factors when making decisions (Kraus et al. 2012; Na et al. 2016). However, the relationship between social class and the tendency to follow a majority has so far been investigated only in product choice settings.

Based on these findings, we argue that voters' social class also plays a role in the political context, specifically in the formation of voting intentions. First evidence that this transfer is possible stems from research on environmental concerns. It has been demonstrated that perceived descriptive social norms about pro-environmental behavior, i.e., the perception of how a majority of others behave, are more predictive of support for pro-environmental action among individuals with lower objective (vs. higher) SES (Eom et al. 2018).

Furthermore, there is empirical evidence that the political bandwagon effect is stronger among less educated voters (Schmitt-Beck 1996), whereas educational attainment can be seen as a proxy for objective SES. This result suggests that lowerclass voters correctly understand the information conveyed in polls but use it systematically differently than higher-class voters do. At the same time, this finding contradicts the potential alternative assumption that people of lower social class attach less weight to poll results when making their own voting decisions because of their lower level of generalized trust (Dahlhaus and Schlösser 2021; Kim et al. 2022). Based on these assumptions and results, we derived the hypothesis that the results of preelection polls more strongly influence voting intentions among people of lower (vs. higher) social class. We investigated this hypothesis in the context of the 2021 German parliamentary election. Our research can thereby produce further insights into bandwagon effects in multiparty systems and proportional representation systems. Because of the interdisciplinary approach and the use of suitable data from the 2021 GLES RCS (Faas et al. 2008; Hoffmann and Klein 2013), our research expands the literature on bandwagon effects and on social cognitive effects of social class in several ways: By clarifying a boundary condition of the political bandwagon effect, the present research contributes to understanding the size and relevance of effects of preelection polls on political attitudes and voting behavior. Thus, it has the potential to inform the debate on regulations on the publication of these polls. At the same time, our study goes beyond previous social psychological research on social class by investigating these effects in the election context. More concretely, our research offers new insights into the generalizability of social cognitive effects of social class, which have so far only been found in limited contexts. Additionally, it allows us to investigate boundary conditions of conformity, a central social psychological concept.

Combining both strands of research makes it possible to investigate the generalizability of the relationship between social class and the tendency to follow a majority. In the long run, our findings might help prepare the ground for developing methods to enhance the political information processing of people from different social-class backgrounds.

3 Data and Methods

We preregistered all our analyses on the open science framework. The preregistration can be accessed on osf.io (https://osf.io/g6r7v/).

3.1 Data

We examined the moderating effect of social class on the relationship between the perceived majority intention and individual voting intention based on data from the GLES RCS 2021.² The RCS is a large-scale, cross-sectional survey based on phone interviews. It comprises German-speaking respondents living in private households who have a landline telephone connection or a mobile number within the Federal Republic of Germany and were eligible to vote in the federal election of 2021 (GLES 2022). Adjustment weights were included based on sociodemographic characteristics

² The dataset and its documentation are available via the GESIS data archive after one-time registration.

(i.e., gender, age, and education) as well as regional characteristics (e.g., municipal regions, east–west comparison; for detailed information on the survey design, see GLES 2022).

Specifically, we used data from the preelection survey collected before the German parliamentary election between August 2, 2021, and September 25, 2021. For each of the 55 field days, the goal was to realize 130 interviews. This resulted in a total sample size of 7068 respondents. To ensure that there were no missing data at the construct level, we listwise excluded all cases with missing data on the focal measures (i.e., educational attainment and current gainful employment status as indicators of objective social class, individual voting intention [and the actual majority's voting intention]).

The RCS data are highly suitable for our research question for three reasons. First, the RCS covers a relatively long time span before the German election and captures voting intentions. Second, the RCS comprises all relevant information for our research question: respondents' educational attainment and their current gainful employment status as proxy for social class, respondents' voting intention, and respondents' perception of poll results in the previous week, i.e. the perception of a majority's voting intention. Third, the RCS includes data from a random sample of the population described above on each field day (GLES 2022).

To examine the potential influences of poll results (i.e., the current majority's voting intention) on individual voting intentions, we combined the RCS data with results of published preelection polls (Blais et al. 2006; Faas et al. 2008). This procedure made it possible to investigate influences of published poll results on voting intentions on a daily basis over the last 8 weeks of the election campaign before the 2021 German federal election (Faas et al. 2008). We used data from the eight leading polling institutes in Germany (in alphabetical order): Allensbach, Forsa, Forschungsgruppe Wahlen, GMS, Infratest dimap, INSA, Kantar Emnid, and Yougov. The data were obtained from Wahlrecht.de (2021).³ We matched each field day's RCS data with the results of preelection polls published 1 day before the RCS field day for each of the six political parties currently represented in the German Bundestag. Because some polls were published in the evening, which could result in some respondents not having the chance to see these polls, a lag of 1 day was chosen. In the event that no new polls were published on a particular day, we matched the RCS data with the most recently published poll results. If results from two or more polling firms were published on the same day, we used the average of these results.

In Fig. 1, the assumed main effect of poll results on voting intentions is displayed as a directed acyclic graph (see Panel a). Even though we had no specific hypotheses about the main effect of social class on voting intentions, we have displayed this path in the model for the sake of completeness. One could argue that our study design only allows examination of correlations. Potentially, unobserved confounders such as external events during the investigated time span could have influenced poll results and the voting intentions measured in the RCS data. Nevertheless, this appears to be unlikely for several reasons. First, the data collection of the polling institutes

³ For each polling institute, data were manually assembled into a Microsoft Excel sheet for processing in R.

60	F. Unkelbach et al.
а	b
Social Class	
Poll Result	Social Class → ΔVoting Intention _{Poll Result}

Fig. 1 a Directed acyclic graph (DAG) for the main effects of social class and poll result on voting intention. b Interaction DAG (IDAG, Nilsson et al. 2021) for the hypothesized moderation effect. This states that social class influences the effect of poll results on voting intentions as measured in the 2021 rolling cross-section study

took place a few days before the assessment of the RCS data. Second, each polling institute collected the poll data over a different number of days and published the data one or several days after data collection. Thus, an external event that might have influenced the poll results would have needed to exert an influence over several days. Even if this were the case, other events that happened temporally closer to the RCS data collection would have been more salient for the RCS respondents and thus would have overshadowed the effect of the first event on voting intention. Accordingly, we concluded that it would have been rather unlikely for an external event to have caused both short-term fluctuations in the respective poll result and short-term fluctuations in the matched RCS data.

Furthermore, the main interest of this paper is to investigate whether the effect of poll results on voting intentions was moderated by social class (Fig. 1b). Inferences regarding this moderation are not necessarily affected by an external event that confounds the relationship between poll results and voting intentions. Instead, a confounder would have had to influence both social class and the effect of poll results on voting intentions at the same time. We argue that such a confounder is unlikely because voters' social class is rather stable over the investigated time span and varies mainly between, rather than within, respondents (cf. OECD 2018).

3.2 Variables

The following variables from the 2021 GLES RCS were used:

Individual Voting Intention Individuals' voting intention was assessed by the single item: "Which party will you vote for in the federal election?" Respondents were instructed to name one political party to which they would give the second vote (i.e., the vote that determines the proportions by which the political parties are represented in the German parliament). The individual voting intention was recoded in dummy-variables for the six political parties currently represented in the German Bundestag (CDU/Christian Social Union [CSU], SPD, Alternative for Germany [AfD], Free Democratic Party [FDP], the Left, Alliance 90/the Greens). Thereby, we treated the three separate answer options "CDU/CSU," "CDU," and "CSU" as a joint category. The reference category was defined as the voting intention for another political party or respondents who were still undecided, who would not cast a vote, or who would cast an invalid vote.

Objective Socioeconomic Status We used two variables assessed in the RCS to compute a composite score of objective SES:

We used respondents' highest level of general education as one indicator for objective social class, as we assumed that an individual's socioeconomic position was (at least partly) given by their educational attainment. To ensure that each level of educational attainment was equally represented, we condensed the seven answer categories of the RCS item into three levels of educational attainment based on the sampling weights in the RCS: *low* (1, 2), *medium* (3, 7), *high* (4, 5). Responses indicating "other school-leaving certificate" (code 6) were coded as missing values.

As a second indicator of objective SES, we used respondents' current gainful employment status. Respondents indicated whether they worked full time, part time, or short time; were in vocational training/studies; or did not work in a paid job at all. We condensed the five answer categories into three levels of employment: full-time (1), part-time/short-time (2, 3), and little to no gainful employment (4, 5).

Following the procedure used in previous studies on social class (e.g., Kraus and Keltner 2009), we coded and standardized educational attainment (0 for low education, 1 for medium education, and 2 for high education) and current gainful employment status (0 for little gainful employment, 1 for part-time/short-time gainful employment, and 2 for full-time gainful employment) and created a single index of objective social class by computing a sum score.

Subjective Social Status Respondents' SSS was assessed with a single item.⁴ They were asked to indicate where they thought they stood in relation to other people in Germany on a ladder from 1 to 11.

Perception of the Majority's Voting Intention The perception of the majority's voting intention was assessed by a single item. Respondents were asked, "Did you read or see any results of current opinion polls on the federal election in the last week?" The two response options *yes* (1) and *no* (0) were dummy-coded, with *no* as the reference category.⁵

We included the following variables assessed in the 2021 RCS as control variables:

Interest in Politics Interest in politics was assessed by the single item, "Quite generally, how interested are you in politics?" The response options ranged from 1 (*very interested*) to 5 (*not at all interested*).

⁴ The item was "Please imagine a ladder with 11 rungs to show where people in Germany stand. At the top—on rung 11—are the people with the most money, the highest education and the best jobs. At the bottom—on rung 1—are those with the least money, the lowest education and the worst jobs or no job. Where would you place yourself on the ladder? Please tell me on which rung—from 1 to 11—you think you stand in relation to other people in Germany in your current phase of life." Unlike the other variables used, SSS was assessed in the RCS postelection survey. Approximately 63% of the respondents from the preelection survey also participated in the postelection survey.

⁵ This measure had some limitations. It did not allow us to draw further inferences on the source of the poll or the time point within the last week when respondents had seen the poll(s). Furthermore, respondents might have misremembered the poll results or exactly when they saw the poll.

Interest in the Current Election Campaign General interest in the current election campaign was assessed by the single item, "And how interested are you in particular in this federal election campaign?" The response options ranged from 1 (*very interested*) to 5 (*not at all interested*).

Party Identification Respondents' party identification was assessed by the single item, "In Germany, many people lean toward a particular political party for a long time, although occasionally they vote for another party. How about you, do you lean toward a particular political party?" The response options included the six political parties currently represented in the German Bundestag (CDU/CSU, SPD, AfD, FDP, the Left, Alliance 90/the Greens). We treated the answer options "CDU/CSU," "CDU," and "CSU" as a joint category. For each of the six parties, we created a dummy variable displaying whether a person leaned toward this party (1) or leaned toward another party/did not lean toward any party/did not know (0).

Issue Orientation Respondents were asked to think about the current political situation and to indicate what they thought the most important political problem in Germany currently was. After naming the perceived most important political problem, respondents were asked to indicate which party they thought would be best able to handle this problem. The response options included the six political parties currently represented in the German Bundestag (CDU/CSU, SPD, AfD, FDP, the Left, Alliance 90/the Greens). We treated the answer options "CDU/CSU," "CDU," and "CSU" as one category. For each of the six parties, we created a dummy variable indicating whether the respective party was named (1) or whether the party was not named/respondent answered that all parties were equally good (0).

Candidate Orientation Respondents were asked how they perceived some of the leading politicians from the six parties currently represented in the German Bundestag (A. Laschet, A. Baerbock, O. Scholz, T. Chrupalla, A. Weidel, C. Lindner, J. Wissler, D. Bartsch, A. Merkel). The response options ranged from -5 (*I do not think much of the politician at all*) to 5 (*I think a great deal of the politician*).

Gender Gender was assessed by three categories: *male, female, and nonbinary*. Because we expected the sample size to be too small to lead to meaningful results, we excluded respondents indicating a nonbinary gender from the respective analyses. We dummy-coded the variable (0=male; 1=female) with male as the reference category.

Age Based on the self-reported year of birth, we computed the age of the participants in years (ranging from 18 to 90 years and older).

As explained above, the RCS data were matched with external data, namely the results of polls published over the course of the election campaign (Blais et al. 2006; Faas et al. 2008; Hoffmann and Klein 2013).

Current Majority's Voting Intention The current majority's voting intention (CMVI) was adopted from the results of the so-called Sunday question. In Ger-

3.3 Data Exclusion and Missing Data

days (level 2).

We excluded participants whose postal codes indicated that they cast their vote in the Saarland because not all of the six parties that were currently represented in the Bundestag were eligible with the second vote there. Specifically, because of a formal error, the state list of the party Alliance 90/the Greens was not admitted for election in the Saarland, which means that voters could not cast their second vote for this party. Technically, this was also the case for the CDU in Bavaria and for the CSU in all federal states but Bavaria; however, due to the close cooperation between these two parties, we treated them as one in our analyses.

We did not impute incomplete or missing data and instead used listwise deletion in our analyses. Furthermore, we did not consider statistical outliers to pose a problem in our analyses because each of the variables we used had only a few answer categories. By implication, extreme answer patterns that would significantly distort our results were unlikely.

3.4 Sampling Weights

Since we were mainly concerned with testing our hypothesis, which had to be true in any sample of participants, we did not plan to include any sample weights.

4 Analysis Plan

4.1 Statistical Models, Robustness Testing, and Model Nonconvergence

To test our hypothesis, we used multilevel logistic regressions, as the RCS survey includes data from different respondents every day in a representative cross-sectional design. Consequently, the respondents were nested in different field days. In our planned analysis, we modeled respondents as level 1 units and field days as level 2 units.

The focus of our research was on the cross-level interaction between CMVI (level 2) and social class (level 1) on voting intention. Following the recommendation by Enders and Tofighi (2007), we first centered all level 1 predictors (i.e., all

⁶ The published poll results, however, are not raw marginals. The polling institutes use specific procedures to transform the raw data into distributions that look like plausible election results. However, these procedures are not revealed to the public (Wüst 2003).

variables aside from CMVI) within field days (group-mean centering).⁷ This made a meaningful interpretation of the cross-level interaction effect possible (Enders and Tofighi 2007). The CMVI as level 2 predictor was grand-mean centered. After centering, we *z*-standardized all predictors. This yielded standardized point estimates of our regression coefficients akin to those obtained through standardization in ordinary least squares regression (Snijders and Bosker 2012).

In accordance with Barr et al. (2013), we followed recommendations regarding the model complexity. We included random slopes for all predictors on level 1 and used an unstructured covariance matrix. In case the models did not converge, we first ran them with a different optimizer, which has been used in a previous version of the lme4 R package (Bates et al. 2015) ("bobyqa"). If this did not solve the problem, we simplified the model complexity by omitting the random slopes for the covariates.

The outcome, voting intention, was operationalized through a dummy variable for each of the six parties. For each party, we conducted a separate multilevel model with maximum likelihood estimates and the respective voting intention as outcome. Since each of the dependent variables was binary, the models included a logit link function. We conducted our analyses in R, using the lme4 package and lmerTest (Kuznetsova et al. 2017).

For the model estimation process, we used a multistep approach. In the first step, we estimated an intercept-only model for each party, which yielded the intraclass correlation coefficient (ICC). For each model, the ICC indicated the proportion of variance in our outcomes that was due to variation between field days (level 2 variation) (Hox et al. 2010). Our hypothesis was based on the assumption that there was variance in voting intentions between field days (which could be traced back to changes in CMVI). If the ICC reached statistical significance, we would conclude that the individual voting intention systematically varied between field days.⁸

In the second step, we estimated a model for each of the six parties in which we included CMVI as a level 2 predictor and the index of objective SES as level 1 predictor. Furthermore, we included the cross-level interaction between the CMVI (level 2) and objective social class (level 1). We expected the voting intentions of individuals to be affected by the poll results, especially for individuals of a lower social class compared with individuals of a higher social class. Thus, we expected a significant cross-level interaction as support for our hypothesis.

In a third step, we tested whether the results were robust by including the two indicators of objective SES separately instead of using the composite score. More concretely, we conducted the same analysis as outlined above but included educational attainment instead of the composite score of objective SES as predictor. For this purpose, educational attainment was contrast coded (-1 for low education, 0 for medium education, and 1 for high education). Additionally, we conducted the anal-

⁷ In this case, group-mean centering means that respondents' personal scores were centered around the mean of the respondents of the respective field day.

⁸ If this was not the case, we would still compute the further planned analyses, as level 2 variation could be suppressed by level 1 predictors and therefore increased once level 1 predictors were introduced into the model (cf. Hoffmann and Klein 2013).

ysis including only gainful employment status as predictor instead of the composite score of objective SES as predictor. This variable was contrast coded (-1 for little to no gainful employment, 0 for part-time/short-time gainful employment, and 1 for full-time gainful employment).

In a fourth step, we tested the robustness of the results regarding the addition of relevant covariates. For the selection of covariates, we followed previous studies on bandwagon effects in Germany (Faas et al. 2008; Hoffmann and Klein 2013). Thus, we included standard primarily sociodemographic controls (age, gender, interest in the campaign, general interest in politics) as well as standard predictors of voting behavior in the Michigan model⁹ (cf. Campbell et al. 1960), which included party identification, candidate orientation and issue orientation. We again estimated a model for each of the six parties, but in addition to the variables we added in the second step, we also included these covariates on level 1.

Additional robustness checks were conducted to investigate the assumed causality direction. More concretely, we computed the same models as outlined above but used voting intentions from the day before the polls were published (t-1) instead of voting intentions from the day after the polls were published (t+1) as outcome. Based on the standardized coefficients for the main effect of poll results on voting intentions, we computed a new binary variable. This indicated the number of times that the effect of poll results on voting intentions at t-1. We conducted a binomial test to investigate whether the effect on t+1 was significantly more often larger than the effect on t-1. If this was the case, we interpreted the result as support for the assumed causality direction.¹⁰ Additionally, we investigated the interaction effect between poll results and objective social class on RCS voting intentions at t-1 and at t+1 following the same procedure outlined above. We expected that this interaction effect would more often be larger when the polls were published before (vs. after) the RCS voting intention had been assessed.

As a final robustness check, we included the perception of the CMVI as additional predictor on level 1 in the main analysis, as we assumed that polls could affect voting intentions only when individuals took note of them. We also included all possible two-way interactions and the three-way interaction between objective social class, CMVI, and perception of the CMVI. We assumed that the voting intentions of individuals would be more strongly affected by the poll results they reported to have seen, especially for respondents of lower objective social class.¹¹

⁹ The Michigan model is a dominant model for the explanation and prediction of voting behavior. Indeed, Hoffmann and Klein's (2013) analyses based on the 2009 RCS data show that party identification, candidate orientation, and issue orientation were significant predictors of voting intention.

¹⁰ We argue that such a test constitutes a conservative check of the expected poll effects, as the RCS voting intention on t-1 was assessed with a smaller temporal distance to the data collection by the polling institutes compared with the RCS voting intention on t+1. This smaller temporal distance might have increased their correlation.

¹¹ However, because of power concerns and the previously mentioned limitations of the variable capturing poll perception, we evaluated our hypothesis based on the two-way interaction between objective social class and CMVI.

4.2 Effect Size, Statistical Power, and Inference Criteria

Since all of the models were nested, we used likelihood ratio tests for model comparisons. We considered p < 0.05 as statistically significant. As our hypotheses were directional, we used one-sided tests to assess the significance of the corresponding regression coefficients. No corrections were made for multiple tests. We expected to be able to use data from an average of 130 respondents per field day for 55 field days. Previous studies on the role of social class in the tendency to follow a perceived majority found medium to large effect sizes (e.g., Na et al. 2016). However, our study differs from these studies in several aspects. Most importantly, previous research was focused on the context of product choice, whereas the aim of the present research was to transfer these findings to the political context.¹² Thus, we anticipated a small effect size for the hypothesized cross-level interaction effect between social class and poll results on voting intentions. A power analysis using an online tool for power analysis for multilevel logistic regression (Astivia et al. 2019) that is based on the R packages lme4, simglm (LeBeau 2021) and paramtest (Hugh 2017) resulted in an estimated power > 0.90 to detect a cross-level interaction effect, with beta = 0.1 for small level 1 and level 2 effects (each with beta=0.1) and a slope variance of $0.09.^{13}$

4.3 Exploratory Analyses

The results of the following exploratory analyses are presented in the online appendix. For exploratory purposes, we conducted the same analyses as presented above but used a time lag of zero days and a time lag of 2 days when matching the published poll results with RCS data. Additionally, we conducted our main analyses as specified above using subjective social class as continuous predictor on level 1. Furthermore, we considered to conduct our main analyses as specified above using transformation and adjustment weights provided with the dataset.

To get a better understanding of psychological correlates of social class in the German context, we conducted a further explorative analysis. This could provide insights into mechanisms of the hypothesized moderation of bandwagon effects via social class. The social cognitive model of social class (Kraus et al. 2012) proposes that social class is related to different psychological tendencies when perceiving one's social environment. Therefore, we explored whether objective/subjective social class were correlated with the tendency to "focus on the whole and less on particular details" as measured in the RCS postelection survey.

¹² Additionally, these other studies were conducted in the United States, had smaller sample sizes, and used an experimental manipulation of the perceived majority opinion.

¹³ As the tool did not permit us to conduct 1000 replications in one simulation, we ran the same simulation ten times with 100 replications each and computed the mean value of the power for the cross-level interaction effect, as recommended by Astivia et al. (2019).

5 Results

5.1 Main Analyses

After applying the preregistered exclusion criteria, the analysis dataset consisted of 5291 respondents nested in field days. This dataset was matched with poll results (CMVI) using a time lag of 1 day for the main analyses.

To assess the impact of the nesting of the data, we specified intercept-only models for each party and computed ICCs (see Table 1). For CDU/CSU, the Left, and AfD, the ICCs were zero, indicating that there was negligible variation in the intention to vote for these parties across field days. For the remaining parties, ICCs were close to zero (SPD: ICC=0.005; FDP: ICC=0.024; Alliance 90/the Greens: ICC=0.002). Although the variation across field days was rather small, we could not preclude the existence of variation in the effect of objective SES as predictor on level 1 (see Barr et al. 2013). Thus, we estimated multilevel models with objective SES, CMVI, and their cross-level interaction as predictors of voting intention for each of the six parties.

For the SPD, CMVI was positively associated with voting intention¹⁴: $\beta = 0.102$, SE = 0.041, OR = 1.107, 95% CI [1.022, 1.201], $p = 0.006^{15}$. The hypothesized interaction effect was, however, not significant: $\beta = 0.012$, SE = 0.037, OR = 1.038, 95% CI [0.940, 1.089], p = 0.376. For the other parties, neither the CMVI nor the interaction effect with individuals' objective SES reached significance.¹⁶ The complete results are displayed in Table 2.

5.2 Robustness Checks

As indicated by preregistration, we conducted further analyses to assess the robustness of our results. First, we varied our operationalization of objective SES by evaluating gainful employment and educational attainment separately instead of using a combined index. We found that our results did not change (see online attachment, Tables A1 and A2).¹⁷ The CMVI was not related to voting intention (except in the case of the SPD), nor did gainful employment moderate this relationship. However, we found that education moderated the relationship between CMVI and measured voting intention (lag 1) for Alliance 90/the Greens such that the relationship was stronger among less educated people. Interestingly, the relationships of employment and education with measured voting intention differed in intensity and went in opposite directions in the case of the Left and the AfD. This could indicate

¹⁴ A model including CMVI as single predictor showed a similar result for this main effect: $\beta = 0.096$, *SE* = 0.039, *OR* = 1.100, 95% CI [1.019, 1.188], *p* = 0.007.

¹⁵ All presented *p*-values are one-tailed (as preregistered) unless stated otherwise.

¹⁶ For these parties, models without objective SES CMVI also had no significant main effect on voting intention, all p > 0.05.

¹⁷ Since our main analyses produced insignificant results regarding the hypothesized interaction effect, we included the detailed results in the online supplement. These have also been made available on osf.io (https://osf.io/g6r7v/).

Table 1	Results fron	ı all int	ercept-on	ily models a	and int	raclass c	orrelation 6	soeffici	ients for	voting inten	tion							
	CDU/C:	SU		SPD			FDP			Alliance	90/the (Greens	The Left			AfD		
	β		SE	β		SE	β		SE	β		SE	β	S	E	β		SE
Intercept	-1.775	* * *	0.039	-1.538	* * *	0.041	-2.714	* * *	0.071	-1.540	* * *	0.038	-3.021	0 ***	.065	-3.157	* * *	0.069
ICC	0.000			0.005			0.024			0.002			0.000			0.000		
N (level 2	(1) = 55, N (lev	rel 1)=:	5291 for	each model	_													

CaCII III 101 167C JU, IN (ICVCI I)-

AfD Alternative for Germany, *CDU* Christian Democratic Union, *CSU* Christian Social Union, *Est.* estimate, *FDP* Free Democratic Party, *ICC* intraclass correlation coefficient, *SE* standard error, *SPD* Social Democratic Party $^{*}p < 0.01$, *** $_{p} < 0.01$, *** $_{p} < 0.01$

	CDU/CS	ŝU		SPD			FDP			Alliance 9)0/the (Greens	The Left		AfD		
	β		SE	β		SE	β		SE	β		SE	β	SE	β		SE
Intercept	-1.789	* * *	0.040	-1.581	* * *	0.041	-2.749	* * *	0.074	-1.583	* * *	0.04	-3.027	*** 0.06	5 -3.161	* * *	0.074
CMVI	0.037		0.039	0.102	* *	0.041	0.021		0.071	0.038		0.039	0.094	0.06	5 -0.067		0.069
SES	-0.174	* * *	0.040	-0.343	* * *	0.037	0.276	* * *	0.066	0.356	* * *	0.039	-0.052	0.06	5 -0.010		0.076
CMVI × SE	S -0.038		0.040	0.012		0.037	0.058		0.063	-0.059		0.039	-0.041	0.06	4 0.048		0.070

SE standard error, SES objective socioeconomic status, SPD Social Democratic Party p < 0.05, p < 0.01, p < 0.01, p < 0.001 that the index for objective SES based on employment and education had a reduced reliability.

Second, we included covariates (gender, age, general interest in politics, campaign interest, party identification, issue orientation, and candidate orientation) in the main analyses. With the exception of objective SES, we again found no significant changes to our main results (see Table A3).

To investigate the direction of the proposed moderated effect, we matched the RCS data with polls published a day after the respective field day (see Table A4). For half of the parties (SPD, FDP, the Left), the relationship of CMVI and voting intentions from the day after the polls were published (main analyses; expected temporal direction) was stronger than the relationship of CMVI and voting intentions from the day before the polls were published. The binomial test did not reach significance: P(effect in expected temporal direction > effect in contrary temporal direction) = 0.500, <math>p = 0.500. For the interaction effects, the pattern was similar; the standardized coefficients were larger for the expected temporal direction for three of the parties (SPD, FDP, AfD). Again, this was not significant: P(effect in expected temporal direction) = 0.500, p = 0.500. Thus, these results did not sufficiently support the assumed direction of effects. We will further discuss the implications of this finding below.

When including the reported perception of CMVI as well as all possible two-way interactions and the three-way interaction between objective SES, CMVI, and CMVI perception, the pattern of results did not change for the relationship of CMVI and voting intention or its interaction with objective SES (see Table A5). Additionally, the three-way interaction did not reach significance for any of the parties. This indicates that voting intentions were not related more strongly with the perceived CMVI for respondents with lower objective SES. However, supporting the notion of a poll effect for the SPD, we found a significant interaction between CMVI and CMVI perception: $\beta = 0.091$, SE = 0.041, OR = 1.095, 95% CI [1.011, 1.184], p = 0.014. This result suggests that the positive relationship between CMVI and voting intention for the SPD was greater when polls were perceived earlier than the respective field day.

5.3 Exploratory Results

Instead of a time lag of 1 day, other time lags can be used to investigate bandwagon effects. Thus, we first computed the main analyses again for a time lag of zero days, which implies investigating the effects of poll results published on the same day as the collection of the matched RCS data. Paralleling the results for a time lag of 1 day, the relationship of poll results and voting intention remained significant for the SPD: $\beta = 0.109$, SE = 0.041, OR = 1.115, 95% CI [1.029, 1.209], p = 0.004. We did not find a significant interaction between objective SES and CMVI for any of the parties.

Additionally, we computed the same analyses for a time lag of 2 days, which means that we investigated the effects of CMVI published 2 days before the assessed voting intention. Again, CMVI was significantly associated with voting intention for the SPD— β =0.105, *SE*=0.040, *OR*=1.111, 95% CI [1.025, 1.203],

p = 0.005—whereas the interaction between objective SES and CMVI did not reach significance for any of the parties. The complete results for the analyses using a time lag of zero and two are presented in the online supplement (Table A6).¹⁸

Investigating SSS instead of objective SES led to the same pattern of results (see Table A7). The CMVI and voting intention remained significantly associated for the SPD— $\beta = 0.082$, SE = 0.046, OR = 1.085, 95% CI [0.992, 1.189], p = 0.038—whereas the interaction between SSS and CMVI did not reach significance for any of the parties.

We further explored whether objective SES and SSS were related to the tendency to perceive the world in a holistic way, or whether the focus on specific details as social cognitive models of social class postulate related differences in psychological tendencies. Interestingly, we found that people with a high objective SES tended to perceive the world in a more holistic way than people with low objective SES—r(3378)=0.086, p (two-tailed) <0.001—whereas SSS had a negative non-significant correlation with this psychological tendency: r(3342)=-0.027, p (two-tailed) =0.119.

5.4 Nonregistered Analyses

To further examine the validity of our conceptualization of objective SES, we computed Spearman's rank correlation between educational attainment and gainful employment. The indicators merely had a small positive correlation— $r_s(5289)=0.150$, p<0.001—which suggests a low reliability of the score for objective SES. Additionally, we computed the correlation between objective SES and SSS and found a small correlation: r(3372)=0.240, p<0.001. This result indicates that people's perceived standing in society was determined by more factors than their objective living conditions.

6 Discussion and Concluding Remarks

Political bandwagon effects have received considerable attention in past research. Previous findings are, however, decidedly inconclusive (Barnfield 2020). The present study improves on prior research in several aspects. First, we integrated theories from political science and social psychology to introduce social class as a possible moderator of bandwagon effects. Second, we investigated this hypothesized moderation in a multiparty system. Third, we relied on high-quality, representative RCS data preceding an exceptionally strongly contested election in Germany. Fourth, we employed a rigorous methodological approach based on adequate modeling of the time-dependency of the data as well as preregistered analyses.

We found only limited evidence for bandwagon effects. Merely for the SPD, a higher vote share in polls (i.e., CMVI) was associated with a higher voting intention 1 day after polls were published. Supporting the notion of a bandwagon effect,

¹⁸ As the interaction of interest did not reach significance in any of the analyses, we abstained from conducting an additional analysis with weights.

this association was stronger for voters reporting to have seen polls previously. Contradicting our hypothesis, there was no evidence for moderation of the association by voters' objective SES for any of the parties. In contrast to previous findings on consumer decisions, social class was not related to the tendency to follow a perceived majority regarding voting intentions. Exploratory analyses with different time lags or with SSS as indicator of social class resulted in the same pattern of results. However, for Alliance 90/the Greens we found first evidence that the association between CMVI and voting intention was greater for voters with less educational attainment.

6.1 Limitations and Future Research

Although we did not find moderation of the association between CMVI and voting intentions by social class, our results provide important insights into bandwagon effects in multiparty systems as well as social class effects in the German context. Thus, our results can serve as guidance for fruitful future research in this field.

Looking at the results, it becomes clear that voting intentions as measured in the RCS data showed little variation across field days. From a statistical perspective, this makes it unlikely to find potential bandwagon effects. This finding is surprising because the 2021 election was characterized by a dynamic election campaign (e.g., Grahn and Süßmann 2021) and comparably volatile poll results. The dynamic of the poll results was, however, not reflected in the pattern of voting intentions from the RCS data (except for the SPD). There are methodological aspects that should be taken into account to understand this result.

Specifically, there were differences in the way the voting intention was assessed in the RCS data and the way the CMVI had been presented by the polling institutes. Importantly, polls did not report on the share of people who did not plan to cast a vote. We, however, included these participants in our assessment of voting intentions because bandwagon effects could potentially also affect them. Additionally, most polling institutes published so-called projected vote shares instead of raw data. Both aspects could have decreased the association between CMVI and measured voting intentions.

Furthermore, the absence of bandwagon effects might be linked with the investigated time period. Previous social psychological research has shown that majority influence is especially strong when people feel a high level of uncertainty (Deutsch and Gerard 1955). Because we investigated poll effects briefly before the election, most respondents had established their preferences already, which might have reduced the tendency to follow others' opinions.

To summarize, our results indicate that even though the investigated election apparently provided optimal circumstances, bandwagon effects did not occur for the majority of parties in Germany. This implies that most voters did not conform to impersonal social influences exerted by poll results but arrived at their voting intention based on factors such as party identification, issue orientation, and candidate orientation (see Table A3). For these parties, public opinion did not impinge on itself.

Nevertheless, we found a positive relationship between the CMVI and measured voting intentions for the SPD. Looking at the pattern of the published poll results gives further insights (see online attachment, Figs. A1–A6): The SPD was the party with the largest increase in projected vote share during the 55 days before the election across the different polling institutes. Additionally, the party was leading the polls for approximately the last month before the election. During the investigated time span, poll results for the FDP, the Left, and AfD were relatively stable, whereas Alliance 90/the Greens and especially the CDU/CSU lost ground continuously. Accordingly, a bandwagon effect, which is defined as voters switching to a party perceived as the majority choice (Barnfield 2020; Schmitt-Beck 2015), is most plausible for the SPD.¹⁹ This result is in line with the findings by Faas et al. (2008), who identified a bandwagon effect for the SPD in the German federal election of 2005. Importantly, falling poll figures did not accelerate the downward trend for Alliance 90/the Greens or the CDU/CSU.

At the same time, the results provide insights into the factors determining the perception of a "winner" in poll results, which has been considered central to bandwagon effects. In the beginning of the investigated time span, the SPD probably did not appear to be a winner to most voters, as the CDU/CSU was leading the polls. This possibly changed when the poll results for the SPD started to rise and continued until the party took the lead in polls and remained at this position until the election. Thus, either the large increase in poll results over time or the leading position in polls (or a combination of both) might have elicited the perception of the SPD as the winner, which potentially motivated voters to switch to the winning side (Meffert et al. 2011; van der Meer et al. 2016). Because previous conceptualizations of a perceived winner were developed for two-party systems, they cannot be adequately applied to multiparty systems. Future research should further distinguish the characteristics of poll results eliciting bandwagon effects in multiparty systems.

However, we did not find evidence for the assumed causality direction of a bandwagon effect for the SPD, which limits the interpretability of the association between CMVI and voting intention. More concretely, voting intention was also associated with poll results published the day after the respective field day for the SPD (see online appendix, Table A4). This finding may indicate that there is no bandwagon effect and that both measures simply reflect, for example, general swings of public opinion. However, it can also be explained by the fact that CMVI and measured voting intention generally reflect the same underlying core construct. Additionally, polls from the day later are based on data that were collected a few days earlier—that is, temporally close to the respective RCS field day. Thus, the found association does not necessarily prove that there is no (causal) bandwagon effect.

Furthermore, in the context of cross-lagged panel designs, researchers have developed the concept of "causal dominance" to interpret similar patterns of associations (e.g., Schuurman et al. 2016). Accordingly, the strongest association between variables measured at different measurement occasions is thought to be the "causally dominant" one, as it exerts the most important causal effect and drives the mecha-

¹⁹ Additionally, it should be noted that poll effects are less likely to be detected with only little variation in poll results across field days.

nism. Following this argument, our goal was to identify which of the paths between poll results and voting intentions was the "causally dominant" (i.e., the larger) one. Our robustness check showed that the association between CMVI and voting intention at t+1 was not significantly larger than the association with voting intention at t-1. However, this test was quite underpowered, as it was based on only six cases (i.e., the parties), which prevented us from identifying the causally dominant path. Future studies should further tap into the causality direction of bandwagon effects by using longitudinal data with multiple measurement occasions. This would allow for the implementation of complete cross-lagged panel designs and the test of causal dominance for poll effects on voting intentions one or more days later.

Regarding our proposed moderation, we found that the CMVI was associated with voting intentions of voters across the social class spectrum equally. There are different potential explanations for the lack of moderation of bandwagon effects by social class. First, social class effects on majority influence might depend on the national context. So far, most evidence for effects of social class on majority influence comes from the United States (e.g., Na et al. 2016; Stephens et al. 2011, 2007). Compared with the United States, Germany is characterized by a lower income inequality (OECD 2022). Assuming that inequality is related to the salience of social class differences (Schneider 2019), social class might have weaker effects on individuals' self-concepts and, consequently, the tendency to follow a perceived majority in Germany compared with the United States. To ensure the generalizability of social class effects, future research could benefit from further cross-national replications of social class effects on majority influence.

Second, the type of decision might play a role in the investigated effects. Whereas people of lower social class show a greater tendency to follow the majority in consumer choices (e.g., Na et al. 2016; Stephens et al. 2011, 2007), this might not be transferable to voting decisions. Possibly, lower-class voters do not follow a perceived majority in the voting context because they perceive the majority of voters (of a potentially higher social class) as less indicative of their own political interests.

Third, the operationalization of objective SES did not include income level, which has been defined as a relevant indicator of social class alongside educational attainment (e.g., Kraus et al. 2012). Because of a lack of a more suitable measure of income in the RCS study, we used gainful employment status as a proxy. There are, however, some indications that this variable did not adequately capture income level. For example, employment status was only weakly correlated with educational attainment, as a relatively large number of people worked full time and had a low educational level. Additionally, it had a weaker association with SSS (r(3372)=0.119, p<0.001) than educational attainment did (r(3372)=0.279, p<0.001). Interestingly, investigating educational attainment separately suggested that effects of CMVI were greater among less educated voters for Alliance 90/the Greens. This finding can be interpreted as first evidence that aspects of objective SES play a role in the size of bandwagon effects and is in line with results by Schmitt-Beck (1996) for the German parliamentary election of 1990. However, this result should be interpreted cautiously, as poll results showed a declining trend over time for this party. To

provide a complete picture of the role of social class in bandwagon effects, future research should include a measure of income level.

Finally, some questions regarding the underlying psychological mechanisms of bandwagon effects remain open because of the use of cross-sectional, correlational survey data. For example, it is unclear how voters' interpretation of poll results is influenced when they are exposed to several polls over the course of a few weeks. Based on social cognitive research, a so-called cumulative redundancy bias (Alves and Mata 2019) might occur, suggesting that voters' impressions about the winner in polls are biased by how the poll results develop over time. Additionally, it remains open how accurately voters remember currently perceived poll results in multiparty contexts. One reason for a lack of evidence for bandwagon effects might be that voters' memory for published polls is biased such that they remember the results for some parties more accurately than others. This would also distort the effects of poll results on voting intentions. Assessing recalled poll results during the election campaign could help shed light on the psychological underpinnings of bandwagon effects. Furthermore, due to the use of cross-sectional data, we were not able to model time-varying individual effects. Again, we think that future research in this field would benefit from using longitudinal data.

7 Conclusion

The present study adds to research on the political bandwagon effect in multiparty systems and the moderating effects of social class. We found limited evidence that published polls were associated with voting intentions in the German parliamentary election of 2021. Only for the SPD were higher poll results associated with higher voting intentions among those who had previously seen the polls. Voters' social class could not be identified as a boundary condition of bandwagon effects. Consequently, taking into account this sociodemographic characteristic of voters does not appear to resolve the mixed evidence in this field of research. Aside from calling into question the generalizability of previously found social cognitive effects of social class across different contexts, our research helps to identify avenues for future research on bandwagon effects in multiparty systems.

Supplementary Information The online version of this article (https://doi.org/10.1007/s11615-022-00417-3) contains supplementary material, which is available to authorized users.

Funding Open Access funding enabled and organized by Projekt DEAL.

Conflict of interest F. Unkelbach, M. John, and V. Vogel declare that they have no competing interests.

Open Access This article is licensed under a Creative Commons Attribution 4.0 International License, which permits use, sharing, adaptation, distribution and reproduction in any medium or format, as long as you give appropriate credit to the original author(s) and the source, provide a link to the Creative Commons licence, and indicate if changes were made. The images or other third party material in this article are included in the article's Creative Commons licence, unless indicated otherwise in a credit line to the material. If material is not included in the article's Creative Commons licence and your intended use is not permitted by statutory regulation or exceeds the permitted use, you will need to obtain permission directly from the copyright holder. To view a copy of this licence, visit http://creativecommons.org/licenses/by/4. 0/.

References

- Adler, Nancy E., Elissa S. Epel, Grace Castellazzo, and Jeannette R. Ickovics. 2000. Relationship of subjective and objective social status with psychological and physiological functioning: preliminary data in healthy, white women. *Health Psychology* 19(6):586–592. https://doi.org/10.1037//0278-6133.19. 6.586.
- Alves, Hans, and Andre Mata. 2019. The redundancy in cumulative information and how it biases impressions. *Journal of Personality and Social Psychology* 117(6):1035–1060. https://doi.org/10.1037/pspa0000169.
- Asch, S.E. 1956. Studies of independence and conformity: I. A minority of one against a unanimous majority. *Psychological Monographs: General and Applied* 70:1–70. https://doi.org/10.1037/h0093718.
- Astivia, Oscar L.Olvera, Anne Gadermann, and Martin Guhn. 2019. The relationship between statistical power and predictor distribution in multilevel logistic regression: a simulation-based approach. BMC Med Res Methodol https://doi.org/10.1186/s12874-019-0742-8.
- Barnfield, Matthew. 2020. Think twice before jumping on the bandwagon: clarifying concepts in research on the bandwagon effect. *Political Studies Review* 18(4):553–574. https://doi.org/10.1177/ 1478929919870691.
- Barr, Dale J., Roger Levy, Christoph Scheepers, and Harry J. Tily. 2013. Random effects structure for confirmatory hypothesis testing: keep it maximal. *Journal of Memory and Language* https://doi.org/ 10.1016/j.jml.2012.11.001.
- Bates, Douglas, Martin Mächler, Ben Bolker, and Steve Walker. 2015. Fitting linear mixed-effects models using lme4. *Journal of Statistical Software* 67(1): 1–48. https://doi.org/10.18637/jss.v067.i01.
- Blais, André, Elisabeth Gidengil, and Neil Nevitte. 2006. Do polls influence the vote? In *Capturing campaign effects*, ed. Henry E. Brady, Richard Johnston, 263–279. University of Michigan Press. https://doi.org/10.3998/mpub.132252.14.
- Brettschneider, F. 2008. The news media's use of opinion polls. In *The SAGE handbook of public opinion research*, ed. W. Donsbach, M.W. Traugott, 479–487. London: SAGE. https://doi.org/10.4135/ 9781848607910.
- Bundeszentrale für politische Bildung. 2021. Datenreport 2021. Ein Sozialbericht für die Bundesrepublik Deutschland. Retrieved from https://www.bpb.de/nachschlagen/datenreport-2021/. Accessed 22 Sept 2021.
- Callander, Stephen. 2007. Bandwagons and momentum in sequential voting. *The Review of Economic Studies* 73(3):653–684.
- Campbell, A., P.E. Converse, W.E. Miller, and D.E. Stokes. 1960. The American voter. John Wiley.
- Cuomo, Andrew. 2020. The virus is the great equalizer. Twitter. https://twitter.com/nygovcuomo/status/ 1245021319646904320?lang=de. Accessed 20 Sept 2021.
- Dahlhaus, Carolina, and Thomas Schlösser. 2021. The interplay of social status and trust: A critical review of concepts, operationalizations, and findings. *Review of General Psychology* 25(2):203–220. https://doi.org/10.1177/10892680211007405.
- Deutsch, Morton, and Harold B. Gerard. 1955. A study of normative and informational social influences upon individual judgment. *The Journal of Abnormal and Social Psychology* 51(3):629–636. https://doi.org/10.1037/h0046408.
- Enders, Craig K., and Davood Tofighi. 2007. Centering predictor variables in cross-sectional multilevel models: a new look at an old issue. *Psychological Methods* 12(2):121–138. https://doi.org/10.1037/1082-989X.12.2.121.
- Eom, Kimin, S. Kim Heejung, and David K. Sherman. 2018. Social class, control, and action: socioeconomic status differences in antecedents of support for pro-environmental action. *Journal of Experimental Social Psychology* 77:60–75. https://doi.org/10.1016/j.jesp.2018.03.009.
- Erb, Hans-Peter, and Gerd Bohner. 2010. Consensus as the key: Towards parsimony in explaining minority and majority influence. In *Minority influence and innovation: Antecedents, processes and consequences*. https://pub.uni-bielefeld.de/record/1938041.
- Erb, Hans-Peter, Gerd Bohner, Knut Schmilzle, and Susanne Rank. 1998. Beyond conflict and discrepancy: cognitive bias in minority and majority influence. *Personality and Social Psychology Bulletin* 24(6):620–633. https://doi.org/10.1177/0146167298246006.
- Faas, T., C. Mackenrodt, and R. Schmitt-Beck. 2008. Polls that mattered: effects of media polls on voters' coalition expectations and party preferences in the 2005 German parliamentary election. *International Journal of Public Opinion Research* 20(3):299–325. https://doi.org/10.1093/ijpor/edn034.

- Genz, Andreas, Klaus Schönbach, and Holli A. Semetko. 2001. "Amerikanisierung"? Politik in den Fernsehnachrichten während der Bundestagswahlkämpfe 1990–1998. In Wahlen und Wähler: Analysen aus Anlass der Bundestagswahl 1998, ed. Hans-Dieter Klingemann, Max Kaase, 401–413. Wiesbaden: VS. https://doi.org/10.1007/978-3-322-95630-9_11.
- GLES. 2022. *Rolling cross-section 2021. GESIS Datenarchiv.* Köln: GLES. https://doi.org/10.4232/1. 13876. ZA7703 Datenfile Version 2.0.0.
- Grahn, Sarah L., and Johannes Süßmann. 2021. SPD liegt in Umfrage vor Union. https://www.zeit.de/ politik/deutschland/2021-08/sonntagsfrage-forsa-wahlumfrage-spd-union-bundestagswahl-2021. Accessed 22 Sept 2021.
- Grossmann, Igor, and Michael E.W. Varnum. 2011. Social class, culture, and cognition. *Social Psychological and Personality Science* 2(1):81–89.
- Hardmeier, S. 2008. The effects of published polls on citizens. In *The SAGE handbook of public opinion research*, ed. W. Donsbach, M.W. Traugott, 504–515. London: SAGE.
- Hardmeier, Sibylle, and Hubert Roth. 2003. Die Erforschung der Wirkung politischer Meinungsumfragen: Lehren vom "Sonderfall" Schweiz. *Politische Vierteljahresschrift* 44(2):174–195. https://doi.org/10. 1007/s11615-003-0037-9.
- Hoffmann, Hanna, and Markus Klein. 2013. Wirkungen von veröffentlichten Wahlumfragen auf die Koalitionserwartung, die Wahlbeteiligung und die Wahlentscheidung bei der Bundestagswahl 2009: Eine Mehrebenenanalyse auf der Grundlage der Rolling Cross-Section-Befragung der German Longitudinal Election Study (GLES). In Koalitionen, Kandidaten, Kommunikation: Analysen zur Bundestagswahl 2009, eds. Thorsten Faas et al., 221–246. Wiesbaden: Springer Fachmedien Wiesbaden. https://doi.org/10.1007/978-3-531-94010-6_11
- Hox, Joop, Mirjam Moerbeek, and Rens van de Schoot. 2010. *Multilevel analysis*. Routledge. https://doi. org/10.4324/9780203852279.
- Hugh, J. 2017. paramtest: R package for varying parameters in simulations or other iterative processes. https://github.com/jeff-hughes/paramtest. Accessed 25 Feb 2022.
- Kim, Youngju, Nicolas Sommet, Na Jinkyung, and Dario Spini. 2022. Social class—not income inequality—predicts social and institutional trust. *Social Psychological and Personality Science* 13(1):186–198. https://doi.org/10.1177/1948550621999272.
- Kraus, Michael W., and Dacher Keltner. 2009. Signs of socioeconomic status: a thin-slicing approach. *Psychological Science* 20(1):99–106. https://doi.org/10.1111/j.1467-9280.2008.02251.x.
- Kraus, Michael W., Paul K. Piff, Rodolfo Mendoza-Denton, Michelle L. Rheinschmidt, and Dacher Keltner. 2012. Social class, solipsism, and contextualism: how the rich are different from the poor. *Psychological Review* 119(3):546–572. https://doi.org/10.1037/a0028756.
- Kraus, Michael W., Jun W. Park, and J. X. Tan Jacinth. 2017. Signs of social class: the experience of economic inequality in everyday life. *Perspect Psychol Sci* 12(3):422–435. https://doi.org/10.1177/ 1745691616673192.
- Kuznetsova, Alexandra, Per B. Brockhoff, and Rune H.B. Christensen. 2017. ImerTest package: tests in linear mixed effects models. *Journal of Statistical Software* 82(13):1–26. https://doi.org/10.18637/ jss.v082.i13.
- LeBeau, Brandon. 2021. simglm: Simulate models based on the generalized linear model. https://CRAN. R-project.org/package=simglm. Accessed 25 Feb 2022.
- Manstead, Antony S.R. 2018. The psychology of social class: How socioeconomic status impacts thought, feelings, and behaviour. *The British journal of social psychology* 57(2):267–291. https://doi.org/10. 1111/bjso.12251.
- Markus, Hazel R., and Shinobu Kitayama. 1991. Culture and the self: Implications for cognition, emotion, and motivation. *Psychological Review* 98(2):224–253.
- Markus, Hazel R., and Shinobu Kitayama. 2010. Cultures and selves: a cycle of mutual constitution. *Perspectives on Psychological Science* 5(4):420–430. https://doi.org/10.1177/1745691610375557.
- Meffert, Michael F., Sascha Huber, Thomas Gschwend, and Franz U. Pappi. 2011. More than wishful thinking: causes and consequences of voters' electoral expectations about parties and coalitions. *Electoral Studies* 30(4):804–815. https://doi.org/10.1016/j.electstud.2011.08.001.
- Morton, Rebecca B., Daniel Muller, Lionel Page, and Benno Torgler. 2015. Exit polls, turnout, and bandwagon voting: evidence from a natural experiment. *European Economic Review* 77:65–81. https:// doi.org/10.1016/j.euroecorev.2015.03.012.
- Moy, Patricia, and Eike M. Rinke. 2012. Attitudinal and behavioral consequences of published opinion polls. In *Opinion polls and the media*, ed. Christina Holtz-Bacha, Jesper Strömbäck, 225–245. London: Palgrave Macmillan UK. https://doi.org/10.1057/9780230374959_11.

- Mutz, Diana C. 1998. *Impersonal influence: How perceptions of mass collectives affect political attitudes*. Cambridge: Cambridge University Press.
- Na, Jinkyung, Ian M. McDonough, Micaela Y. Chan, and Denise C. Park. 2016. Social-class differences in consumer choices: working-class individuals are more sensitive to choices of others than middleclass individuals. *Personality & Social Psychology Bulletin* 42(4):430–443. https://doi.org/10.1177/ 0146167216634043.
- Nilsson, Anton, Carl Bonander, Ulf Strömberg, and Jonas Björk. 2021. A directed acyclic graph for interactions. *International Journal of Epidemiology* 50(2):613–619. https://doi.org/10.1093/ije/dyaa211.
- Oakes, J. Michael, and Peter H. Rossi. 2003. The measurement of SES in health research: current practice and steps toward a new approach. *Social Science & Medicine* 56(4):769–784. https://doi.org/10.1016/S0277-9536(02)00073-4.
- OECD. 2018. A broken social elevator? How to promote social mobility. Paris: OECD Publishing. https:// doi.org/10.1787/9789264301085-en.
- OECD. 2022. Income inequality (indicator). https://doi.org/10.1787/459aa7f1-en.
- Partheymüller, Julia, and Anne Schäfer. 2013. Das Informationsverhalten der Bürger im Bundestagswahlkampf 2013. *Media Perspektiven* (12):574–588.
- Riambau, Guillem. 2018. Do citizens vote for parties, policies or the expected winner in proportional representation systems? Evidence from four different countries using a multiple-type model. *Party Politics* 24(5):549–562. https://doi.org/10.1177/13540688166688669.
- Schmitt-Beck, Rüdiger. 1996. Mass media, the electorate, and the bandwagon. A study of communication effects on vote choice in Germany. *International Journal of Public Opinion Research* 8(3):266–291. https://doi.org/10.1093/ijpor/8.3.266.
- Schmitt-Beck, Rüdiger. 2015. Bandwagon effect. In *The international encyclopedia of political communi*cation, ed. Gianpietro Mazzoleni, 1–5. Wiley. https://doi.org/10.1002/9781118541555.wbiepc015.
- Schneider, Simone M. 2019. Why income inequality is dissatisfying—perceptions of social status and the inequality-satisfaction link in Europe. *European Sociological Review* 35(3):409–430. https://doi.org/ 10.1093/esr/jcz003.
- Schuurman, N.K., E. Ferrer, M.M. de Boer-Sonnenschein, and E.L. Hamaker. 2016. How to compare crosslagged associations in a multilevel autoregressive model. *Psychol Methods* 21(2):206–221. https:// doi.org/10.1037/met0000062.
- Snijders, Tom A.B., and Roel J. Bosker. 2012. Multilevel analysis: an introduction to basic and advanced multilevel modeling. SAGE. https://lib.ugent.be/catalog/rug01:001698339.
- Stephens, Nicole M., R. Markus Hazel, and Sarah S.M. Townsend. 2007. Choice as an act of meaning: the case of social class. *Journal of Personality and Social Psychology* 93(5):814–830. https://doi.org/10. 1037/0022-3514.93.5.814.
- Stephens, Nicole M., Stephanie A. Fryberg, and Hazel R. Markus. 2011. When choice does not equal freedom. Social Psychological and Personality Science 2(1):33–41. https://doi.org/10.1177/ 1948550610378757.
- Stephens, Nicole M., Hazel R. Markus, and L.T. Phillips. 2014. Social class culture cycles: how three gateway contexts shape selves and fuel inequality. *Annual review of psychology* 65:611–634. https:// doi.org/10.1146/annurev-psych-010213-115143.
- Stolwijk, Sjoerd B., Andreas R.T. Schuck, and Claes H. de Vreese. 2016. How anxiety and enthusiasm help explain the bandwagon effect. *International Journal of Public Opinion Research* https://doi.org/ 10.1093/ijpor/edw018.
- Thiel, Georg. 2021. Forsa hat mit Klage gegen Bundeswahlleiter Erfolg. https://www.zeit.de/politik/ deutschland/2021-09/forsa-georg-thiel-briefwahl-wahlumfragen-sonntagsfrage-streit-meinungs forschungsinstitut-gerichtsurteil. Accessed on Sep 23 2021.
- Van der Meer, Tom W.G., Armen Hakhverdian, and Loes Aaldering. 2016. Off the fence, Onto the bandwagon? A large-scale survey experiment on effect of real-life poll outcomes on subsequent vote intentions. *International Journal of Public Opinion Research* 28(1):46–72. https://doi.org/10.1093/ ijpor/edu041.
- Wahlrecht.de. 2021. Sonntagsfrage Bundestagswahl. [Data set]. https://www.wahlrecht.de/umfragen/. Accessed 26 Nov 2021.
- Wüst, Andreas M. 2003. Stimmung, Projektion, Prognose? In *Politbarometer*, ed. Andreas M. Wüst, 83–107. Wiesbaden: VS. https://doi.org/10.1007/978-3-663-11058-3_7.

Do Consumers of All Social Classes Prefer Best Sellers? Three Preregistered Replication Studies of Na et al. (2016)

Fabienne Unkelbach^a, Pia Kehder^b, and Michaela Wänke^a

^aChair for Consumer and Economic Psychology, University of Mannheim, A5, 6, 68159 Mannheim, Germany; ^bUniversity of Giessen, Erwin Stein Building, Goethestrasse 58, 35390 Giessen, Germany

Author Note

Correspondence concerning this article should be addressed to Fabienne Unkelbach (https://orcid.org/0000-0002-9107-3533), Chair for Consumer and Economic Psychology, University of Mannheim, A5, 6, 68159 Mannheim, Germany. The authors declare that there are no potential conflicts of interest with respect to the research, authorship, and/or publication of this article.

All materials (exception: product pictures (can be shared upon request)), R code for all reported analyses, raw data and codebooks are available on the OSF: https://osf.io/fmc29/?view_only=34bc5c1f50df4bb9b46c33eb5b81e50d

Abstract

This report presents three pre-registered replication studies on the role of social class in the tendency to align one's product choices with those of others. The original research found that working-class Americans were more likely to conform to the majority's product choice compared to middle-class Americans. These class-based differences in conformity were explained by a more pronounced interdependent self-construal among lower-class compared to higher-class individuals. However, empirical evidence for a negative relationship between social class and an interdependent self-construal is mixed, which also calls into question the robustness of its relationship with conformity. The aim of the present research was to clarify the generalizability and replicability of the role of social class in conformity in product choices, while taking into account the cultural context. Thus, we conducted three preregistered replications of the original experiment, two of them with German samples and one with a U.S. sample (total N = 592). None of the studies were able to replicate the negative relationship between social class and the tendency to adjust one's product choices to the preferences of the majority. The relationship was not significant in the German samples and it even pointed in the opposite direction as expected in the U.S. sample. These results suggest that the generalizability of the relationship between social class and the tendency to follow social preferences is more limited than previously thought. We highlight the importance of conducting conceptual replication studies using different operationalizations of conformity and adopting a cross-cultural perspective on social class.

Keywords: social influence, social class, conformity, consumer choices, replication

Do Consumers of All Social Classes Prefer Best Sellers? Three Pre-registered Replication Studies of Na et al. (2016)

In modern consumer contexts, companies often use the strategy of communicating information about which product the majority of consumers ostensibly like in order to influence consumers' decisions. This is done, for example, by labeling products as 'best sellers' in advertising campaigns. This marketing tactic is based on several theoretical underpinnings. Social psychology has long identified social influence as an important predictor of individuals' choices (e.g., Asch, 1956). In marketing research, studies have confirmed that presenting information about majority preferences for products (i.e., suggesting that a majority of consumers have purchased this product on previous occasions) can effectively shift consumer decisions in various contexts (e.g., Rao et al., 2001; Roethke et al., 2020; Salazar et al., 2013; Salmon et al., 2015; Thomas et al., 2017). Within this field, social influence has often been conceptualized as *herd behavior* or *bandwagon effect* (Leibenstein, 1950) which describes the tendency of individuals to adopt the viewpoint of the majority even if it differs from their own (for an overview, see Bindra et al., 2022).

Yet, social influence does not appear to be equally large among all individuals. Aside from contextual factors, several inter-individual difference variables have been identified that moderate the strength of social influence. For example, people that perceive a high level of uncertainty (Deutsch & Gerard, 1955) or a low self-esteem (e.g., Chou et al., 2013; Tainaka et al., 2014) show a larger tendency to follow others' preferences.

Extending this line of research, Na et al. (2016) introduced consumers' social class as a moderator of majority influence on purchase decisions. Using U.S. student samples, their research showed that individuals from lower social class (i.e., working class) expressed a larger tendency to adapt their product choice to a majority preference compared to those from higher social class (i.e., middle class). Notably, the authors established that social class influences the extent of majority influence through its negative relationship with an interdependent self-construal, supporting the so-called social cognitive theory of social class (Kraus et al., 2012). Na et al.'s paper, published in a prestigious social psychological journal, holds significance as it is frequently cited in articles focusing on the psychological foundations of social class (e.g., Carey & Markus, 2017) and applied studies on environmentally conscious consumer behavior (Eom et al., 2018; Sparkman et al., 2020). This recognition is reflected in its placement within the top 25% of research outputs tracked by Altmetric (as of February 2024, https://sage.altmetric.com/details/6201240#score).

However, more recent social psychological research on social class has raised concerns about potential methodological shortcomings of prevailing practices generally and, consequentially, about the generalizability of previously found social class effects. Specifically, the usage of inadequate measures of social class has been criticized (Antonoplis, 2023) as well as the reliance on rather small samples consisting mainly of U.S. students (see e.g., Gobel & Miyamoto, 2024). Indeed, studies have shown that particularly the association between social class and an interdependent self-construal appears to be less robust than expected (Boileau, 2022; see also Stephens et al., 2007). This also calls into question the generalizability of the previously found negative relationship between social class and conformity.

By conducting three pre-registered replication studies of Na et al.'s (2016) Study 1, two of them in the German context and one in the US context, with more diverse samples in terms of social class and considering multiple indicators of the construct, the present research contributes to clarifying the generalizability of the role of social class in majority influence. Moreover, a subordinate goal was to shed light on the assumed underlying mechanism, classbased differences in self-construal.

Social Class and Self-Construal

Na et al.'s (2016) theoretical argument is based on a social cognitive perspective on social class which postulates that repeated experiences in social class contexts shape individuals' psychological tendencies (Markus & Stephens, 2017; Kraus et al., 2012; Kraus & Stephens, 2012). Within these models, social class is defined by a) objective components, i.e., ones' access to (material) resources (socio-economic status, SES), and b) subjective components, i.e., perceived standing in the societal hierarchy (subjective social status, SSS). A main premise of these models is that the material constraints experienced by lower-class individuals limit the opportunities to follow own ideals and shift the focus on the feelings and behaviors of others (e.g., Kraus et al., 2012; Kraus & Stephens, 2012). In contrast, individuals from higher-class contexts experience less constraints and a larger freedom to pursue independent choices and are, thus, more oriented toward their self. Among effects on other psychological tendencies, these class-based differences are expected to be reflected in the self-construal: People from lower social classes are predicted to develop a more interdependent self-construal, i.e., a view of their self as embedded in their social relations, whereas people from higher social classes are predicted to develop a more independent selfconstrual, i.e., a view of their self as unique, separate entity (e.g., Carey & Markus, 2017; Kraus et al., 2012).

Even though the assumed link between social class and an interdependent selfconstrual constitutes a central premise of social cognitive models of social class, the empirical evidence does not appear consistent: While research by Fernández et al. (2005) and Grossmann and Varnum (2011) showed that students of lower social class (measured via parental education) reported a more interdependent self-construal across different cultures, another study could not replicate this relationship (Stephens et al., 2007). Additionally, Na et al. (2010) proposed class-based differences in social orientation, yet a closer look at their results reveals a dependence on the specific measures employed. There was no significant class-based (measured via own education) difference in an interdependent self-construal as measured via the Self-Construal Scale by Singelis (1994) and the difference was even significant in the opposite direction as expected when using the Twenty Statement Task (Kuhn & McPartland, 1954) (see Table S4, Na et al., 2010), suggesting that higher-class individuals have a more interdependent self-construal. Furthermore, recent large-scale studies based on representative samples from the U.S. and Germany (A. Batruch, personal communication, December 6, 2023) using several of the most common self-report scales did not support the assumption that lower-class individuals exhibit a more interdependent selfconstrual. In sum, class-based differences in self-construal which were proposed as mediator in the context of conformity in product choices are less robust than proposed.

Social Class and (Product) Choices

Independent of the assumed psychological underpinnings, however, several studies on product choices have yielded results in line with the predictions of social cognitive models of social class. In particular, lower-class individuals were more likely to be influenced by social preferences in their product choices and reacted more positively to reduced individuation compared to higher-class individuals (Stephens et al., 2007). For example, people of lower social class (as measured via parental educational attainment) more often chose the same pen as an ostensible former participant and liked their choice more when another person apparently made the same choice (vs. a different choice) (Stephens et al., 2007). In contrast, another person's choice did not affect the ratings of the chosen pen for participants with higher social class. These findings suggest that individuals from lower social classes might be relatively more likely to make choices which promote similarity to and connection with others whereas individuals from higher social classes might be relatively more likely to make choices and differentiation from others (Stephens et al., 2011; Stephens et al., 2007).

Building on this line of research, studies by Na and colleagues (2016) showed that a higher sensitivity to social preferences among lower-class individuals could even override personal preferences in product choice scenarios. The authors conducted three quasiexperimental studies based on U.S. student samples (Study 1: N = 43, Study 2: N = 107, Study 3: N = 101). Within these studies, participants firstly made 60 product choices among equally attractive alternatives and received manipulated feedback indicating that the majority of previous participants had either chosen the same product (consistent trials) or a different one (conflicting trials). Then, they were shown the product pairs again and asked to indicate which product they would purchase based on their current feelings.¹² The number of *changes* in product choice in conflicting trials relative to consistent trials served as a measure of conformity. Na and colleagues found that members of the working class (as operationalized by maternal education and SSS) made more changes compared to members of the middle class when the majority had ostensibly made a deviating choice. There was no significant difference between middle- and working-class participants when the majority had ostensibly made the same choice (effect size of the interaction effect between SES and type of feedback (Study 1): $\eta_p^2 = .29$). The authors concluded that working-class Americans show a larger tendency to align their product choices with a perceived majority choice than middle-class Americans.

Na and colleagues (2016) replicated their main finding across two additional studies aimed at testing the cross-cultural robustness and participants' self-construal as potential underlying mechanism. In Study 2, Na et al. found that the effect of social class on majority

¹² In a final recognition phase participants were presented with the product pairs from the previous phases as well as with 37 new pairs. They had to complete an old-new-recognition task and indicate which item was the more popular one for the old pairs. The results showed that middle-class participants had worse memory for the feedback that the majority had made a different choice than they had, and no such effect was found for working-class participants ($\eta_p^2 = .17$).

influence was diminished when investigating students with East-Asian background which they explained with a generally larger tendency to follow social preferences in interdependent cultures. In Study 3, a similarity vs. difference priming was used to imitate the effects of an interdependent vs. independent self-construal among an American sample. Social class had no significant effect on the tendency to follow a majority product choice over and above this priming which the authors interpreted as support for class-based differences in self-construal as underlying psychological mechanism. While this approach was beneficial for understanding the causal effect of self-construal (as mediator) on the tendency to follow majority preferences, it omitted investigating the relationship between social class and selfconstrual, and, thus, did not allow for an analysis of the entire indirect effect (see Pirlott & MacKinnon, 2016). Considering the above-mentioned mixed evidence for the negative relationship between an interdependent self-construal and social class, it appears crucial to clarify the robustness of the relationship between social class and choice behavior under social influence based on (cross-cultural) replication studies with larger, more diverse samples.

The Present Research

After Study 1a (pre-registered) was planned as a first attempt to replicate class-based differences in conformity in the German context as part of an originally different research project, we conducted two more targeted, pre-registered replication studies of Study 1 by Na and colleagues (2016). These aimed at providing diagnostic information on the replicability and generalizability of the proposed negative effect of social class on the tendency to follow a majority in product choices in a different cultural context. We hereby focused on Na et al.'s product choice change paradigm to investigate the level of conformity with a majority.¹³ While Studies 1a and 1b were conducted with German samples, Study 2 constituted a high-powered replication study with a U.S. sample to more closely follow the study of Na et al. Additionally, Study 1b and Study 2 aimed to investigate the relationship between social class and chronic interdependent self-construal, which was proposed by Na et al. as a potential mediator for the social class effect.

Aside from implications for external validity, a replication of the social class effect on conformity outside the U.S.A. appears to be relevant for theoretical considerations. Specifically, we argue that societal characteristics of the U.S.A. may have enhanced the link between social class and conformity in prior studies. Previous research shows that the positive association between social rank and individualistic traits is particularly pronounced in cultures in which individualism is highly valued (Gobel & Miyamoto, 2024; Zhang et al., 2021). As the U.S.A. is the most individualistic country in the world (Hofstede et al., 2010), it can be expected that American individuals from higher social classes express a particularly lower tendency to adjust their product choices to social preferences. In Germany as less individualistic product choices. Secondly, the U.S.A. is characterized by a higher level of income inequality compared to Germany (Gini index 2021: 37.4 vs. 29.7; Solt, 2019).

¹³ Due to pragmatic considerations concerning the study length, we abstained from additionally assessing memory performance in Studies 1b and 2.

As a higher inequality can increase the salience of one's social class and consequently social class effects (Cheung & Lucas, 2016; Schneider, 2019), the link between social class and the tendency to follow a majority product choice may be smaller in Germany. Consequently, conducting replication studies for Na et al.'s (2016) research in the German context constitutes a conservative test of the robustness of the investigated social class effects.

Thus, if we could replicate that people with lower social class show a larger tendency to follow a perceived majority product choice in all three studies independently from the cultural setting, this would support the robustness of the finding across Western countries. The absence of such an effect in the German studies but not the U.S. study would indicate that the generalizability of the finding by Na et al. (2016) might be more limited by cultural context than expected. Finally, if we could not replicate the original finding in any of the studies, this would cast doubts on the robustness of effects of social class for majority influence generally.

All of the reported studies in this manuscript were pre-registered (Study 1a: <u>https://aspredicted.org/45V_LBL</u>, Study 1b: <u>https://aspredicted.org/7VC_8JD</u>, Study 2: <u>https://aspredicted.org/Q47_GYZ</u>) and all preregistrations included the study design, planned sample size, exclusion criteria, and planned primary analyses. All preregistrations adhere to the disclosure requirements of aspredicted.org. All pre-registered analyses are reported in the manuscript or Supplement. All materials (exception: product pictures (shared upon request)), R code for all reported analyses, data and codebooks are available at https://osf.io/fmc29/?view_only=34bc5c1f50df4bb9b46c33eb5b81e50d

Study 1a

Study 1a constituted a first attempt to replicate the first study by Na et al. (2016) with a German sample and had the preliminary goal to explore class-based differences in conformity in another, less individualistic cultural context. We followed Na et al.'s methodology as closely as possible but adapted the study in the following aspects: First, as we did not have access to the original stimulus material¹⁴ and in order to use products common to German consumers, we used standardized images of products taken from German online stores. Second, while the original study was conducted as laboratory experiment, we collected data online as in-person data collection was restricted in spring 2020 due to pandemic regulations in Germany. Therefore, unlike the original study, we implemented pre-registered exclusion criteria to ensure high data quality, such as failing a seriousness check and answering too few attention checks correctly. Nevertheless, this approach allowed us to collect data from a larger, more diverse sample of participants and, thus, with larger statistical power. Third, we went beyond Na et al. by not only measuring SSS and maternal education as indicators of working vs. middle class membership, but also assessing and analyzing additional indicators of SES such as participants' own educational attainment and household income level, in order to gain a more nuanced understanding of social class effects. If the findings of Na et al. can be replicated, only participants with lower SES (working class) respectively lower SSS should be responsive to social preferences, i.e., make more changes

¹⁴ We had contacted the first author of the original paper, Jinkyung Na, in March 2020 to borrow their study material but did not receive a response.

when a perceived majority of others has made a different product choice than when a perceived majority of others has made the same product choice. Participants with higher SES (middle class) respectively SSS should not be influenced by social preferences in their product choices.

Method

Materials

Following the procedure employed by Na et al. (2016), participants were presented with 60 product pairs individually and asked to choose the product they would like to purchase for each pair. Within each pair, the products were in the same product category (e.g., two watches) and we ensured that they differed only in their color or design. The images of the products were taken from German online retail websites and had a standardized white background. Most of the products were utilitarian everyday objects and did not show the brand name at all or only in an unobtrusive way¹⁵.

Participants

Na et al. (2016) investigated their *change of choice* – paradigm with student samples from the University of Texas at Dallas (Study 1: N = 43, Study 2: N = 107, Study 3: N = 101). Since Study 1a was not originally planned as a conclusive replication study, we relied on an a priori power analysis to determine the sample size. Using G*Power (Faul et al., 2009) for alpha = .05 and a power of 90% to detect a medium effect size of $f^2 = 0.15$ (original effect size Study 1: $\eta_p^2 = .29$) in a multiple regression analysis with one predictor resulted in a sample size of 73 participants. To investigate effects of social class, it seems crucial to have a diverse sample which is why we collected data via prolific.co. Data was obtained from 86 German participants. After excluding participants according to the pre-registered exclusion criteria¹⁶ (failing the seriousness check or more than 12 of 60 attention check items), we arrived at a final sample size of 77^{17} participants (18 female, 57 male, 2 diverse; $M_{age} = 28.13$ years, SD = 8.88). Only about one third of the participants were students (n = 24), suggesting that this sample was more diverse in this respect than the original study.

Procedure

Study 1a was conducted via SoSci Survey (Leiner, 2019). Although it was conducted online instead of in the lab, we closely followed the procedure described by Na et al. (2016). At the beginning of the study, participants were informed that the study was about consumer choices. In choice phase 1, participants viewed the 60 product pairs and indicated for each pair which of the two products they would purchase if they had to make a decision. It was

¹⁵ With the exception of one product pair, when a brand name was visible, it was identical for both products in a pair. This was the case for 20 of the 60 product pairs. For the product pair with different visible brand names, we ensured that the price of the two products was comparable at the time of data collection. Excluding this product pair did not change the pattern of results.

¹⁶ One person did not answer the seriousness check but provided answers on all other measures. Excluding this person from the dataset did not change the pattern of results.

¹⁷ Notably, this sample size is slightly below the pre-registered target sample size of 80 but was still considered sufficient based on the power analysis.

randomized which of the two products in a pair was presented on which side of the screen. This was kept constant over the course of the experiment. After each product choice participants received manipulated feedback regarding the popularity of the chosen product. The same item as in Na et al. was used, reading that x% of previous participants had made the same choice. In half of the trials, a random percentage between 75% and 95% was presented, reflecting a majority (consistent trials). In the other half, a random percentage between 5% and 25% was presented, reflecting a minority (conflicting trials). Following each feedback, participants were asked to indicate the more popular product, serving as an attention check.

In choice phase 2, participants were presented with the same product pairs as in the previous phase in an individualized randomized order and asked again which product they would purchase. We presented a German translation of the instruction used by Na et al. (2016), asking the participants to rely on their current feelings instead of attempting to recall their previous choices. Furthermore, we added that sometimes peoples' taste changes when looking at products a second time but that the first impression remains in other cases.

In the last phase, participants completed an old/new-recognition task with the 60 old and 37 new product pairs. If a participant identified a product pair as old, they additionally indicated which item was more popular according to the feedback presented before.

Finally, we measured participants' SSS using the MacArthur Scale (e.g., Adler et al., 2000) (M = 5.65, SD = 1.61, range: 1-8). Among further demographics (age, gender, country of education, country of birth, nationality, years of living in Germany, parents' and grandparents' country of birth, ethnicity¹⁸, German language proficiency) (see Supplement (1) for descriptives), indicators of SES were assessed. These items were adapted to the German context. Specifically, participants indicated their educational level on a scale with 8 options plus an "other" option (*Median* = high school diploma) and their current annual gross household income on category options from 1 (*below 15,000€*) to 8 (*over 150,000€*) (*Median* = 35,001€ - 50,000€). Furthermore, participants indicated their parents' educational attainment, their own occupational status, occupation and the number of people living in the household. After indicating whether they had answered all questions seriously and honestly participants were debriefed.

Results

Changes in Choice Predicted by Social Class

We computed one variable for the number of changes in product choice in consistent trials (M = 2.29, SD = 2.37) and a variable for the number of changes in conflicting trials (M = 3.45, SD = 3.12). A paired samples *t*-test showed that participants made significantly more changes in conflicting trials than in consistent trials, t(76) = 4.98, p < .001, d = 0.57, 95% CI for d [0.32, 0.81]. This suggests that the manipulation of social preferences was successful, and people generally followed the product choice of a perceived majority. Following Na et al.'s analytical approach (2016), we subtracted the number of changes in consistent trials from the number of changes in conflicting trials to obtain a conformity score (M = 1.17, SD = 2.06) with higher values reflecting a larger conformity with the majority choice. In the following,

¹⁸ The ethnicity item used by Na et al. (2016) for U.S. samples was adapted such that it included ethnic groups that were relevant in the German context.

we firstly report the findings from the pre-registered analyses using SSS respectively a score of SES (based on education and household income) as predictor. Then, we provide an exact replication of the analyses of the original research, categorizing participants as either working or middle class based on maternal education.

Subjective Social Status. Firstly, the conformity score was regressed on SSS as single predictor. As a lower social class was expected to be associated with more conformity in product choices, a negative association would be in line with the original results.¹⁹ Additionally, this relationship was expected to be robust to controlling for participants' gender and ethnicity. For SSS, we found a non-significant association with conformity, $\beta = 0.16$, SE = 0.11, t = 1.37, p = .174, 95% CI [-0.07, 0.38], JZS BF = 0.53 (controlling for gender: $\beta = 0.14$, SE = 0.11, t = 1.19, p = .240, 95% CI [-0.09, 0.37], JZS BF = 0.48). After excluding outliers on the dependent variable, SSS was significantly associated with the conformity score, but contrary to the findings by Na et al. (2016) individuals with higher SSS indicated more conformity, $\beta = 0.24$, SE = 0.11, t = 2.07, p = .042, 95% CI [0.01, 0.47], JZS BF = 1.48.

Objective SES. We repeated the analysis with a single index of SES. For this purpose, we coded and z-standardized educational attainment and household income and took their mean (see Kraus & Keltner, 2009). For students, pupils, and people in training, we created an adjusted SES index based on their parents' educational level (mean of father's and mother's educational level) and their current family household income. Contrary to Na et al.'s (2016) finding, SES was not significantly associated with the conformity score when included as single predictor, $\beta = 0.05$, SE = 0.12, t = 0.44, p = .663, 95% CI [-0.18, 0.28], Jeffrey–Zellner–Siow (JZS) BF = 0.26. Controlling for gender²⁰ did not change this pattern, $\beta = 0.03$, SE = 0.12, t = 0.27, p = .788, 95% CI [-0.20, 0.26], JZS BF = 0.25. Furthermore, using own education or household income as separate indicators did not change the pattern of results (see Supplement (1)). Finally, neither SES nor SSS were significantly associated with the total number of choices (SES: p = .325; SSS: p = .403).

Replication of Na et al.' s (2016) Analysis. To directly replicate the analyses of the original article, we used maternal education as a binary indicator of social class. Participants whose mother did not have a university degree (n = 60) were labeled as working class and participants whose mother had a university degree or a doctoral degree were labeled as middle class (n = 17). Indeed, working-class participants indicated a significantly lower SSS (M = 5.43, SD = 1.63) than middle-class participants (M = 6.41, SD = 1.33), t(75) = -2.27, p = .026, d = -0.62, 95% CI for d [-1.17, -0.07]. A 2 (social class: working vs. middle) x 2 (trial type: consistent vs. conflicting) mixed ANOVA (with Greenhouse-Geisser corrections, R package

¹⁹ We deviated from the pre-registration in using two-sided instead of one-sided p-values to remain consistent to the original study and across the present studies. This did not change the pattern of significance for any of the main results.

²⁰ To include gender as covariate, we excluded two participants that had indicated "diverse" for their gender. Different from our pre-registration, we did not include ethnicity as covariate as the large majority of participants identified as German (n = 74) rendering any effect of this variable unlikely to find.

11

rstatix, Kassambara, 2023) resulted in a non-significant interaction effect between social class and type of feedback, F(1, 75) = 0.67, p = .417, $\eta_p^2 = .01$. Again, only the main effect of feedback was significant, F(1, 75) = 20.93, p < .001, $\eta_p^2 = .22$ (main effect of maternal education: p = .656). Excluding outliers on the dependent variable did not change the pattern of results.

Additional Analyses

Memory Accuracy for Social Feedback Predicted by Social Class. Paralleling the approach of the original study, we subtracted the memory accuracy for feedback on social preferences (i.e., the probability of accurately identifying the more popular product in each pair that was correctly recognized as old) in conflicting trials from the respective memory accuracy in consistent trials and regressed this difference score separately on SES and on SSS. Na et al. (2016) had found that participants were overall better at remembering feedback on social preferences in consistent compared to conflicting trials. Importantly, this effect was moderated by social class such that individuals from the middle class had worse memory for conflicting social feedback compared to consistent social feedback while there was no such difference among working-class individuals.

In the present study, memory accuracy for social preferences (M = 0.69, SD = 0.08) was significantly higher than chance level (.50), p < .001. Similar to the original research, we found that participants showed a higher memory accuracy for social preferences in trials with consistent (M = 0.80, SD = 0.13) compared to conflicting feedback (M = 0.58, SD = 0.17), t(76) = 7.38, p < .001, d = 0.84, 95% CI for d [0.58, 1.10]. However, different from Na et al.'s (2016) results, a difference score of memory accuracy (computed as memory accuracy for trials with consistent minus memory accuracy for trials with conflicting feedback) was not significantly associated with SES, p = .871 (measured via maternal education: p = .556), or SSS, $p = .548^{21}$. Controlling for gender did not change this pattern of results, p = .977, /p = .504.

Discussion

In contrast to Na et al.'s (2016) findings, we did not find a moderation of the tendency to follow social preferences in product choices by indicators of SES. This suggests that individuals across the social class spectrum did not differ in this regard in the German context. Preliminary evidence suggests that individuals with higher SSS may be more inclined to follow the perceived majority choice, contrary to previous findings. Furthermore, we did not find any class-based differences in memory accuracy for perceived social preferences. In conclusion, the social class effect reported by Na et al. was not replicated in Study 1a. Notably, the manipulation of social preferences was successful: In line with previous research on bandwagon effects (e.g. Cho et al., 2022; Leibenstein, 1950), we found that people generally changed their product choice more often to follow the majority when the majority

²¹ Additionally, paralleling Na et al.'s findings (2016), neither SES nor SSS predicted hit rates (old pairs identified as old), false alarm rates (new pairs misidentified as old) or discrimination indices (see Supplement (1)), suggesting that people across the social class spectrum differentiated old and new product pairs equally well.

had preferred a product they had not originally chosen. This result suggests that the general paradigm was adapted successfully to investigate social influence in the German context.

However, although the present study assessed a larger and more diverse sample compared to the original research, it was not sufficiently powered to provide conclusive evidence for a non-replication of the interaction effect between social class and feedback on social preferences. To be able to make a meaningful statement about the replicability of the role of social class in choice behavior in the German context, and thus, the cultural context as potential boundary condition for social class effects, we took several measures to increase power in the following study.

Study 1b

In Study 1b, we made several improvements to the design from Study 1a. We assessed a larger German sample to increase statistical power and adhered more closely to the approach by Na et al. (2016) by focusing on a predominantly student sample. Additionally, we manipulated temporary perceptions of SSS instead of measuring chronic SSS. Thus, we were able to examine rank-based, social comparative aspects of social class and investigate causality beyond the original research. In the first choice phase of Study 1a, participants' answers suggested that for some product pairs, the products were not perceived as equally attractive (see Supplemental Table S1). Therefore, we limited the number of product choice trials to the 26 product pairs that were closest to being perceived as equally attractive in Study 1a. In Study 1b, we omitted the assessment of memory of the feedback, as this part of the study had substantially increased its length while being less informative for the role of social class in majority influence per se.

Additionally, we attempted to shed some light on the psychological mechanism underlying social class differences in conformity by assessing the perceived similarity with other participants. Na et al. (2016) found that experimentally priming a higher (vs. lower) perceived similarity with others as a proxy for an interdependent self-construal diminished class-based differences in the tendency to conform to social preferences (Study 3). The authors concluded that individuals from lower social classes are more likely to follow a majority due to a higher perceived similarity with others. To clarify if this can be generalized to the German context, we assessed how similar participants perceived themselves to other participants and explored potential class-based differences in this perception.

Method

Materials

To ensure that the products within each product pair were equally attractive, we selected the 26 product pairs that best fulfilled this criterion in Study 1a based on the decisions of the participants in the first choice phase. Specifically, we selected 21 pairs based on binomial tests and an additional 5 pairs that came close to achieving similar levels of attractiveness across the respective products (see Supplemental Table S1).

Participants

Taking into account the small effect size found in Study 1a, we conducted an a-priori power analysis using G*Power (Faul et al., 2009) for alpha = .05 and a power of 80% to
detect a small effect size of f = 0.10 for the interaction effect of a 2 x 2 mixed ANOVA which resulted in a sample size of 200. We pre-registered a minimum sample size of 200 valid cases and collected data from 242 participants. In doing so, we also met the sample size recommendation by Simonsohn (2015) for replication studies (i.e., at least 2.5 times the original sample size). Participants were recruited from social media platforms and were offered course credit or the chance to take part in a raffle. After applying pre-registered exclusion criteria (failing more than 6 of the 26 attention check items, failing a seriousness check), we arrived at a sample of 207 German participants. Additionally, we excluded participants with invalid values (i.e., answers outside the range of 1 to 10) on their SSS (n =4). The final sample consisted of 203 participants (159 female, 43 male, 1 diverse; $M_{age} =$ 24.75 years, SD = 10.52). The majority of the sample were students (n = 166, including 101 psychology majors).

Procedure

Study 1b was conducted online. The procedure of Study 1b differed to the one from Study 1a in some respects. In the beginning of the study, we manipulated participants' SSS through downward versus upward comparisons (Aydin et al., 2019; Piff et al., 2010). Participants were presented with the MacArthur Scale, an image of a ladder with 10 rungs representing German society (Adler et al., 2000). They were asked to compare themselves to people at the top (n = 98) vs. bottom (n = 105) of the ladder which was meant to induce the feeling of having a lower vs. higher perceived standing in society. To strengthen the manipulation, participants were asked to indicate differences between themselves and people from the respective end of the ladder. As a manipulation check, they answered on which ladder rung they feel they stand within German society.

Then, the first choice phase started and was set up like in Study 1a, but with a reduced number of 26 product pairs. Again, after each product choice, participants received feedback on the ostensible majority preference followed by an attention check item asking to indicate which of the two products was the more popular one. In the second choice phase, participants were again presented with the same 26 product pairs and indicated how they felt about the products now. Finally, participants completed the measures of SES among further demographics (gender, age, German language skills). Parallel to Study 1a, SES was assessed via educational level (*Median* = high school diploma) and current household income (8-point scale plus "don't know" option²²) (*Median* = 50,001 \in - 75,000 \in). Additionally, we assessed parents' educational attainment, participants' occupational status, occupation and the field of studies for students. Different from Study 1a and to further explore potential social class differences in the perception of the feedback on the majority choice, participants rated their agreement to the statement that the other participants of the study are similar to them on a scale from 1 (*disagree completely*) to 7 (*agree completely*) (*M* = 3.95, *SD* = 1.14).²³ After a seriouness check participants were debriefed.

https://osf.io/fmc29/?view_only=34bc5c1f50df4bb9b46c33eb5b81e50d).

²² 56 participants indicated that they did not know their household income.

²³ At the end of the study, participants were asked to answer several items in English that were part of a different, unrelated research project (see study material at

Results

Manipulation Check

Participants who compared themselves to people from the bottom of the ladder indicated a significantly higher perceived standing in society (M = 6.49, SD = 1.28) than participants that compared themselves to people from the top of the ladder (M = 5.92, SD = 1.45), t(201) = -2.96, $p^{24} = .003$, d = -0.42, 95% CI [-0.69, -0.14]. This result suggests that the manipulation of SSS was successful.

Changes in Choice Predicted by Social Class

Overall, the number of changes in product choice made by participants was rather low (M = 2.01, SD = 2.15) and about a third of the participants (n = 65) did not make any changes in product choice.

Subjective Social Status. A 2-between (SSS: low vs. high) x 2-within (feedback: conflicting vs. consistent) mixed ANOVA with the number of changes in product choice as dependent variable was computed (with Greenhouse-Geisser corrections, R package afex, Singmann et al., 2022). Replicating the result of Study 1, feedback type had a significant main effect, F(1, 201) = 29.36, p < .001, $\eta_p^2 = .13$, 95% CI [0.05, 0.22], as participants made significantly more changes in product choice after conflicting (M = 1.31, SD = 1.56) than after consistent feedback (M = 0.70, SD = 1.07). Different from Study 1a, SSS had a small but significant main effect on the total number of choices, F(1, 201) = 4.85, p = .029, $\eta_p^2 = .02$, 95% CI [0.00, 0.08], indicating that, averaged over feedback type, participants in the high SSS condition made significantly more changes in product choice (M = 1.17, SD = 1.48) than participants in the low SSS condition (M = 0.84, SD = 1.23). As in Study 1a and not in line with the tested hypothesis, the interaction effect between SSS and feedback type was not significant, F(1, 201) = 0.43, p = .514, $\eta_p^2 = .002$, 95% CI [0.00, 0.03] (see Figure 1). When conducting the analysis with excluded outliers on the changes in choice (n = 20) the main effect of SSS was not significant, p = .123, whereas the rest of the results remained robust.

 $^{^{24}}$ Again, different from the pre-registration, we used two-sided instead of one-sided *p*-values to remain consistent across studies.

Figure 1

Number of Changes in Product Choice Depending on Manipulated Subjective Social Status and Type of Feedback on Majority Preferences



Note. N = 203. SSS = Subjective social status. In trials with consistent feedback, participants received the feedback that a majority of participants (a random number between 75 and 95%) had made the same product choice as they have. In trials with conflicting feedback, participants read that a minority of participants (a random number between 5 and 25%) had made the same product choice as they have. Error bars represent standard errors of the mean.

In addition to the pre-registered analyses, we included SSS as measured in the manipulation check as continuous predictor of conformity in a linear regression. To obtain the criterion, we subtracted the number of changes in consistent trials from the number of changes in conflicting trials as in Study 1a (M = 0.61, SD = 1.58). While SSS was descriptively negatively associated with this difference score, indicating that participants with lower SSS made more changes after conflicting compared to consistent feedback, this effect was not significant, $\beta = -0.08$, SE = 0.07, t = -1.19, p = .237, 95% CI [-0.22, 0.06], JZS BF = 0.30.

Objective SES. Parallel to Study 1a, we created an index of SES based on educational attainment and household income (for students and apprentices: based on their parents' education and family household income). Importantly, SES did not differ significantly between the two experimental groups, p > .420, suggesting that the SSS manipulation did not affect the objective indicators of social class.

While SES was negatively associated with the above-mentioned difference score of changes in choice, $\beta = -0.13$, SE = 0.08, t(145) = -1.61, p = .109, 95% CI [-0.30, 0.03], JZS BF = 0.58, this association failed to reach significance at conventional significance levels.

This result was robust to controlling for gender²⁵, $\beta = -0.10$, SE = 0.08, t(143) = -1.22, p = .226, 95% CI [-0.27, 0.06], JZS BF = 0.47. Furthermore, SES was not associated with the total number of changes (SES: p = .195). Using own education or household income as separate predictors each did not change the pattern of results (see Supplement (2)).

Replication of Na et al.' s (2016) Analysis. Replicating the analyses by Na et al. (2016) by using maternal education as binary indicator of social class (working class: n = 117, middle class: n = 82) resulted in a non-significant interaction between social class and type of feedback on the number of changes in product choice, F(1, 197) = 1.86, p = .174, $\eta_p^2 = .01$, 95% CI [0.00, 0.05]. Again, merely feedback type had a significant main effect on the number of changes in choice, F(1, 197) = 24.88, p < .001, $\eta_p^2 = .11$, 95% CI [0.04, 0.20] (for maternal education p > .101). The results for the different indicators of SES were robust to excluding outliers on the dependent variable.

Exploratory Analyses

Perceived Similarity. As expected, participants who perceived themselves as more similar with the other participants showed a higher level of conformity with the majority preference, i.e., they changed their product choice significantly more often after conflicting compared to consistent feedback, $\beta = 0.17$, SE = 0.07, t = 2.37, p = .019, 95% CI [0.03, 0.30]). The relationship between perceived similarity and conformity did not vary significantly across SSS conditions, $\beta = 0.01$, SE = 0.07, t = 0.08, p = .939, 95% CI [-0.13, 0.14]), or the SES spectrum, $\beta = 0.01$, SE = 0.08, t = 0.15, p = .883, 95% CI [-0.14, 0.17]).

Additionally, participants in the low SSS condition did not perceive themselves as more similar to other participants (M = 3.92, SD = 1.11) compared to participants in the high SSS condition (M = 3.98, SD = 1.17), t(201) = -0.39, p = .696, d = -0.05, 95% CI [-0.33, 0.22]). Similarly, SES was not significantly correlated with the perceived similarity with other participants, r(145) = -0.02, p = .764 (for education and income: ps > .330).

Discussion

In contrast to Na et al.'s findings (2016), Study 1b replicated the results of Study 1a and showed that neither temporarily shifted SSS nor chronic SES were associated with the tendency to follow a perceived majority when making product choices in the German context. Surprisingly, individuals with higher SSS expressed a generally higher tendency to change their product choice independent of feedback on social norms. Consistent with the findings of Study 1a, social feedback influenced consumer choices, with individuals generally conforming to the product choices of others. This result supports the assumption that the experimental paradigm was adequately adapted to the German context and highlights the importance of social influence in consumer choices when strong prior preferences are absent (e.g., Leibenstein, 1950).

²⁵ One participant who indicated "diverse" for their gender was excluded when we included gender as covariate

Additionally, Study 1b shed light on the role of perceived similarity in conformity and social class effects. Consistent with prior research on social norms (e.g., Neighbors et al., 2010), our results suggest that individuals show a larger tendency to follow others who they perceive as similar to themselves. However, we found no significant relationship between social class and the perceived similarity with other participants which challenges the notion of perceived similarity as mediator of class-based differences in conformity.

In sum, the results of Study 1a and 1b did not support the notion that individuals of lower social class are more likely to follow social preferences in consumer choices. As outlined above, the diverging results of our research compared to Na et al.'s (2016) findings might be due to the different cultural context of our research. Possibly, social class effects on conformity are less pronounced in Germany compared to the U.S.A. due to the country's less individualistic societal norms and lower income inequality. To determine if the effect of social class is dependent on the cultural context, we conducted a third replication study with a U.S. sample.

Study 2

Study 2 aimed to replicate the finding by Na et al. (2016) that social class moderates the effect of majority influence on product choices with a U.S. sample. As we did not find this effect in German samples in Studies 1a/b, Study 2 was conducted to clarify the potential role of the cultural context as a boundary condition for this social class effect.

Furthermore, Study 2 had the goal to shed light on its underlying psychological mechanism. Na et al. (2016) argued that class-based differences in conformity can be explained by a more interdependent self-construal among lower-class individuals. They showed that the link between social class and the sensitivity to social preferences was diminished among people from more interdependent cultures (Study 2) and under conditions in which an interdependent (compared to an independent) self-construal was temporally induced (Study 3). While these studies provided some evidence for the role of self-construal in the responsiveness to social preferences, they did not show that individuals from different social classes – but from the same cultural background – differ in their chronic interdependent self-construal. As outlined above, the empirical evidence regarding this association is mixed. To advance the understanding of the potential mediating role of self-construal, we went beyond prior research and assessed individuals' independent and interdependent self-construal.

Finally, we attempted to explore the association between social class and other potentially relevant psychological characteristics. Specifically, we assessed self-esteem as lower-class individuals tend to report lower self-esteem (Rosenberg & Pearlin, 1978; Twenge & Campbell, 2002), which, in turn, may reduce the tendency to follow a perceived majority (e.g., Chou et al., 2013; Tainaka et al., 2014). Additionally, we explored the role of the need for uniqueness which is defined as the desire to distinguish oneself from others and is often reflected in unusual consumer choices such as a preference for rare products (Lynn & Harris, 1997; Snyder & Fromkin, 1977) as well as nonconformity (Imhoff & Erb, 2009). Importantly, prior research also found that a higher need for uniqueness is associated with a more pronounced independent self-construal (Kastanakis & Balabanis, 2012).

Method

Materials

In order to stick with the procedure employed by Na et al. (2016) as closely as possible, to ensure a higher level of reliability and to maintain comparability to Studies 1a/b, we used the pictures of the 60 product pairs from Study 1a.

Participants

In light of the small effect sizes found in the German samples, we conducted an apriori power analysis using G*Power for alpha = .05 and a power of 80% to detect a small effect size of $f^2 = 0.03$ for a multiple regression with one predictor which resulted in a required sample of 264. To maximize power in light of our resources, we pre-registered a minimum sample size of 303 valid cases and collected complete data from 327 US participants on prolific.co (with incomplete submissions: 350)²⁶.

Following our pre-registration, we excluded participants who failed more than 12 of the 60 attention checks and arrived at a final sample of 312 US participants (154 female, 148 male, 6 diverse, 4 no indication; $M_{age} = 32.41$ years, SD = 11.06). Among them, 230 participants identified as White/Caucasian, 55 as African American, 18 as Asian American, 20 as Hispanic/Latino and 5 as Native American (and 9 used an open answer format; multiple mentions were possible). Different from the original study, students made up merely 15.7% of the present sample. However, the sample appeared to be similar to the one investigated by Na et al. (2016) with regard to some aspects: Operationalizing SES via maternal education (i.e., no bachelor's degree/bachelor's degree or higher) led to an almost equal number of participants being classified as working class (47.3%) and middle class (52.7%) as found in the original study (original study: working class: 48.8%, middle class: 51.2%). Whereas our sample was also approximately balanced for gender (49.4% females), Na et al.'s sample had a slightly larger proportion of female participants (62.8%). With regard to ethnicity, both in our sample and in the one by Na et al., the majority of participants (73.7% vs 79.1%) identified as White/Caucasian whereas African Americans constituted the second largest group (17.6% vs 11.6%).

Procedure

Study 2 was conducted online. The beginning of the study was identical to Study 1a as we again attempted to follow the procedure by Na et al. (2016) very closely. Participants saw the 60 product pairs as described in the previous studies and indicated for each pair which of the two products they would purchase if they had to choose. Following each choice, manipulated feedback on social preferences (conflicting vs. consistent trials) was presented as described above. After the feedback participants indicated which of the two products was more popular as an attention check. Then, the second choice phase followed like in Study 1a.

Then, several psychological characteristics were assessed to advance the understanding of class-based differences in the psychological make up which might underly

ensure a sufficient sample size considering the pre-registered exclusion criteria.

²⁶ We arrived at a slightly larger final sample than pre-registered (312 instead of 303 participants) as we aimed to

its relationship with conformity. Firstly, participants' self-esteem was assessed using the 10item Rosenberg scale (1965). Participants expressed agreement with the statements on a scale from 1 (*strongly disagree*) to 4 (*strongly agree*) (Cronbach's $\alpha = .90$).

Next, participants completed a short version of the Singelis' self-construal scale (D'Amico & Scrima, 2016) measuring interdependent and independent self-construal (Cronbach's $\alpha = .73/.77$) with 5 items per dimension on a scale from 1 (*strongly disagree*) to 7 (*strongly agree*). We used this measure to examine the assumption by Na et al. (2016) that lower class individuals were more responsive to social preferences because of their more pronounced interdependent self-construal.

Then, need for uniqueness was assessed, both generally and specifically in the consumption context. As a measure of general need for uniqueness, participants completed the 12-item subscale on lack of concern regarding others' reactions to one's different ideas and actions of the need for uniqueness scale (Snyder & Fromkin, 1977; see also Schumpe et al., 2016; scale: 1 (*strongly disagree*) to 5 (*strongly agree*), Cronbach's $\alpha = .87$). Additionally, we included three items from the scale on the *desire for unique consumer products* by Lynn and Harris (1997) which appeared specifically suited to the consumption context (e.g., "*I enjoy having things that others do not*") (Cronbach's $\alpha = .53$).

Finally, participants indicated their SSS on the MacArthur Scale (M = 5.50, SD = 1.79) and among further demographics (gender, age, country of education, country of birth, nationality, years of living in the U.S.A., number of parents and grandparents born in the U.S.A., ethnicity), indicators of SES were assessed. Specifically, participants reported their educational level on a scale with 8 options plus an "other" option (*Median* = Bachelor's degree) which was adapted from the U.S. Census Bureau (2021). Current annual household income was assessed on category options from 1 (*below 15,000\$*) to 8 (*over 150,000\$*)²⁷ (*Median* = \$50,001 und \$75,000). Parallel to Na et al. (2016), we also assessed parents' educational attainment, participants' employment status, their occupation and the number of people living in the household.

Results

Changes in Choice Predicted by Social Class

Paralleling Studies 1a/b, we computed one variable for the number of changes in product choice in consistent trials (M = 2.96, SD = 3.03) and a variable for the number of changes in conflicting trials (M = 6.10, SD = 5.85). Based on these variables, we computed a conformity score as in Studies 1a/b (M = 3.14, SD = 5.09). Replicating the effect found in the first studies, participants made overall more changes in product choice in conflicting trials than in consistent trials, t(311) = 10.88, p < .001, d = 0.62, 95% CI [0.49, 0.74], indicating that they generally followed social preferences.

Subjective Social Status. We regressed the difference score as measure of conformity on SSS. Contrary to our expectation, we found a significant *positive* association with SSS, $\beta = 0.30$, SE = 0.05, t = 5.57, p < .001, 95% CI [0.20, 0.41], JZS BF > 199657, suggesting that people with higher SSS made relatively more changes in product choice after conflicting than

²⁷ To assess income, we used the same number of category options as Na et al. (2016) but adjusted the income categories to better distinguish between lower income levels (see Piff et al., 2010).

after consistent feedback. This result was robust when controlling for gender²⁸ (female = 0, male = 1) and ethnicity (0 = other, 1 = White/Caucasian) as dummy-coded predictors²⁹, β = 0.29, *SE* = 0.05, *t* = 5.34, *p* < .001, 95% CI [0.18, 0.40], JZS *BF* > 86723. Additionally, the main effect of SSS on the number of choices was significant, β = 0.24, *SE* = 0.06, *t* = 4.40, *p* < .001, 95% CI [0.13, 0.35].

Objective SES. Paralleling Study 1a/b, we computed a composite score of SES. Similar to SSS, a higher SES was significantly positively associated with a larger difference score of changes in product choice, $\beta = 0.30$, SE = 0.05, t = 5.63, p < .001, 95% CI [0.20, 0.41], JZS BF > 270681. This association remained robust when controlling for gender and ethnicity³⁰ as described above, $\beta = 0.28$, SE = 0.05, t = 5.09, p < .001, 95% CI [0.17, 0.39], JZS BF > 106971. Furthermore, the main effect of SES on the number of choices was significant, $\beta = 0.32$, SE = 0.05, t = 5.88, p < .001, 95% CI [0.21, 0.42]. Excluding outliers on the conformity score (n = 4) or using own education respectively household income as separate predictors each did not change the pattern of results (see Supplement (3)).

Replication of Na et al.' s (2016) Analysis. As direct replication of the analyses of Na et al. (2016), we used maternal education as binary indicator of social class as in Study 1a/b (working class: n = 148; middle class: n = 163; one person did not indicate maternal education)³¹. Paralleling the result for the SES score, a 2 (social class: working vs. middle) x 2 (trial type: consistent vs. conflicting) mixed ANOVA indicated that individuals made more changes in conflicting compared to consistent trials, F(1, 309) = 119.99, p < .001, $\eta_p^2 = .28$, 95% CI [0.20, 0.36]. Additionally, this effect was moderated by social class, F(1, 309) = 17.07, p < .001, $\eta_p^2 = .05$, 95% CI [0.01, 0.11]. Post hoc comparisons revealed that, contrary to the results found by Na et al. (2016), the effect of feedback on social preferences was larger among middle-class participants, t(309) = -10.93, p < .001, d = -0.62, 95% CI [-0.74, -0.50], compared to working-class participants, t(309) = -4.26, p < .001, d = -0.27, 95% CI [-0.38, -0.15], suggesting that people from the middle class showed a higher level of conformity. Furthermore, different from the original study, the main effect of social class was significant, F(1, 309) = 25.87, p < .001, $\eta_p^2 = .08$, 95% CI [0.03, 0.14], indicating that people from the middle class made overall more changes compared to people from the working class.

²⁸ To include gender as covariate, we excluded six participants who had indicated "diverse" for gender and four participants that had not answered this item.

²⁹ Surprisingly, male participants showed a higher level of conformity compared to female participants, $\beta = 0.15$, *SE* = 0.05, *t* = 2.74, *p* = .006, 95% CI [0.04, 0.26], and participants that did not identify as White/Caucasian

showed a higher level of conformity compared to White/Caucasian participants, $\beta = -0.13$, SE = 0.05, t = -2.43, p = .016, 95% CI [-0.24, -0.03].

³⁰ Again, gender significantly predicted conformity, $\beta = 0.13$, *SE* = 0.05, *t* = 244, *p* = .016, 95% CI [0.03, 0.24] (for ethnicity: *p* = .060).

³¹ The differentiation was reflected in the reported SSS, with working-class participants indicating a significantly lower SSS (M = 5.03, SD = 1.88) compared to middle-class participants (M = 5.94, SD = 1.58), t(288) = -4.64, p < .001, d = -0.53, 95% CI for d [-0.75, -0.30].

The Role of Self-Construal

Na et al. (2016) argued that people with a lower social class have a more pronounced interdependent self-construal which would mediate their higher tendency to follow a majority in their product choices. Contradicting this pattern, a lower SES was significantly associated with a *lower* (not higher) interdependent self-construal, r(308) = 0.13, 95% CI [0.02, 0.24], p = .021 in our sample. For SSS, the association with an interdependent self-construal was not significant, r(310) = 0.03, 95% CI [-0.08, 0.14], p = .561. Regarding an independent self-construal, there was no significant association with SES, r(308) = 0.09, 95% CI [-0.02, 0.20], p = .108. However, in line with our expectations, a higher SSS was significantly associated with a more independent self-construal, r(310) = 0.17, 95% CI [0.06, 0.28], p = .002.

Contrary to the assumption that a more interdependent self-construal would lead to a higher responsiveness to social preferences, it did not significantly predict the number of changes in choice in conflicting vs. consistent trials, $\beta = 0.10$, SE = 0.06, t = 1.73, p = .085, 95% CI [-0.01, 0.21] (for separate correlations per trial type, see Table 1). Instead, a more pronounced independent self-construal significantly predicted the conformity score, $\beta = 0.21$, SE = 0.06, t = 3.80, p < .001, 95% CI [0.10, 0.32].

Y
H
Ŧ
$\boldsymbol{\triangleleft}$
G
Ĕ
Z
0
$\overline{\mathbf{O}}$
3
4
S
Ā
Ľ,
\mathbf{O}
Ţ
I
Ũ
0
0
$\tilde{\mathbf{\omega}}$
H
Ř
\mathcal{O}
5
Ę
A
Z
_

Table 1

Descriptive Statistics and Bivariate Correlations among Objective SES, Subjective Social Status, Gender, Ethnicity, Self-Esteem, Self-Construal, Need for Uniqueness, and the Number of Changes in Product Choice After Conflicting and Consistent Feedback (Study 2)

	1	2	3	4	5	9	7	8	6	10	11	12
1. SES												
2. SSS	.62											
3. Gender	.11	90.										
4. Ethnicity	04	.03	09									
5. Self-Esteem	.29	.31	00.	08								
6. Independent Self	60.	.17	.11	03	.47							
7. Interdependent Self	.13	.03	.06	.08	17	10						
8. Need for Uniqueness - G	02	.08	.04	01	.42	.32	44					
9. Need for Uniqueness - C	90.	.04	00	12	08	.08	.21	16				
10. Changes Conflicting	.34	.29	.19	13	.15	.24	.17	10	00.			
11. Changes Consistent	.15	90.	.07	02	60.	.10	.16	05	.02	.49		
12. Conformity Score	.31	.30	.18	14	.12	.21	.10	08	01	.86	03	
W	0.08	5.50	ı		3.04	4.68	4.20	3.12	3.46	6.10	2.96	3.14
SD	0.86	1.79	ı	ı	0.66	1.26	1.16	0.76	0.80	5.85	3.03	5.09

with conflicting feedback, i.e., feedback that a minority of others had chosen the same product as oneself. Changes Consistent = Number of changes in product choice in trials American/Asian American, Hispanic/Latino and open answers, Native American, 1 = White/Caucasian). Changes Conflicting = Number of changes in product choice in trials with consistent feedback, i.e., feedback that a majority of others had chosen the same product as oneself. Conformity Score = Number of changes in product choice in trials Note. N = 300-312. SES = Socioeconomic status. SSS = Subjective social status. Gender (0 = female, 1 = male) and ethnicity were dummy-coded (0 = African with conflicting feedback minus number of changes in product choice in trials with consistent feedback. All correlations \geq .12 or \leq -.11 are statistically significant (p < .05).

22

Self-Esteem and Need for Uniqueness

People with higher social class indicated a higher self-esteem (for SES: r(308) = 0.29, 95% CI [0.18, 0.39], p < .001, for SSS: r(310) = 0.31, 95% CI [0.20, 0.41], p < .001). However, surprisingly, a higher self-esteem significantly predicted a *higher* conformity score, $\beta = 0.12$, SE = 0.06, t = 2.11, p = .036, 95% CI [0.01, 0.23]. When controlling for self-esteem, the positive relationship between social class and conformity remained significant (for SES/SSS: p < .001).

Need for uniqueness (generally and specifically in the consumer context) was neither significantly associated with social class (for SES: p > .280; for SSS: p > .160) nor with the conformity score (p > .160) (for details, see Table 1).

Discussion

Following closely the experimental procedure of Na et al. (2016) and using data from a diverse U.S. sample, Study 2 did not replicate the finding that people with lower social class show a larger tendency to change their product choice to follow an apparent majority. Contrary to the findings of the original research, American participants with *higher* social class were more likely to conform to social preferences. Additionally, the present research examined the role of chronic interdependent self-construal as mediator of social class effects on conformity. In contrast to the assumed negative relationship between social class and an interdependent self-construal (e.g., Kraus et al., 2012), SES was *positively* related with this type of self-construal. This finding is consistent with recent large-scale cross-national replication studies (A. Batruch, personal communication, December 6, 2023).

Furthermore, although participants demonstrated a general tendency to follow the majority in their product choices, the conformity score showed an unexpected pattern of correlations not only with social class but also with other interindividual difference variables. Specifically, a more pronounced independent self-construal and a higher self-esteem predicted more changes in choice in conflicting trials compared to consistent trials which appears to contradict prior research on the effectiveness of social norms (e.g., Chou et al., 2013; Tainaka et al., 2014). However, as this study was designed to examine the replicability of the original results by Na et al. (2016), the evidence for countervailing effects on conformity is only preliminary and should be interpreted with caution. We further discuss the divergent pattern for social class effects in the U.S. and the German context in the "General Discussion". Nevertheless, the pattern emphasizes that the relationship between social class and conformity is less generalizable than expected based on previous research.

General Discussion

Across three replication studies (two with German samples, one with a U.S. sample), we did not find support for the notion that people of lower social class show a higher level of conformity with a majority in product choices compared to people of higher social class as found by Na et al. (2016). The effect sizes for this association were either small and non-significant (Studies 1a/b) or, interestingly, even pointed significantly in the opposite direction (Study 2), suggesting that people with higher social class showed more conformity. In the following, we interpret the results separately by cultural context.

Studies 1a/b constituted close replications of Na et al.'s (2016) Study 1 in a Western European context. We deviated from the approach by Na et al. mainly in the following aspects: In Study 1a, we collected data from a more diverse sample instead of a student sample, whereas in Study 1b, we manipulated (instead of measured) SSS to be able to investigate its potential causal effect. In both German samples, social class was not significantly associated with conformity in product choices. This result was found for several continuous measures of objective SES as well as for chronic and experimentally manipulated SSS. In line with this result, people with lower class did also not indicate a higher perceived similarity with other participants which was proposed as potential mediator by Na et al. These results are consistent with recent research from the German political context which failed to find class-based differences in the tendency to follow a majority of voters as perceived from poll results in their voting intentions (Unkelbach et al., 2023). Possibly, these results can be explained by structural or cultural differences between the German and the U.S. context. As outlined above, the lower level of individualism (e.g., Hofstede et al., 2010) or the lower level of income inequality in Germany compared to the U.S. might explain why we could not find class-based differences in the level of conformity in Studies 1a/b.

Noteworthy, however, with a U.S. sample we did also not find a negative relationship between social class and conformity with a majority product choice (Study 2). Instead, people with a lower social class (as measured via SES and SSS) showed even less conformity with a perceived majority. However, this finding is preliminary, and further research is needed to replicate and explain this effect. Additionally, the relationship between social class and self-construal which was proposed as mediator in previous research deviated from the expected pattern. People with a lower SES even indicated a less interdependent self while the relationship between SES and an independent self-construal was small and nonsignificant. For SSS, merely a significant positive association with an independent self-construal was found. Overall, the associations of social class indicators and self-construal were weak at best. Taken together with the results from a recent large-scale study which failed to replicate a link between a lower social class and a more interdependent self-construal across the U.S. and Germany (A. Batruch, personal communication, December 6, 2023), our results raise further doubts about the proposed link between social class and self-construal.

To test the robustness of our findings, we conducted some exploratory analyses. As we were not able to use the product pictures from the corresponding original study for the change of choice task, we repeated the analyses with different types of subsets of pictures for Study 1a and Study 2 such as a) only product pairs in which both products were equally attractive in the first choice phase or b) only the first 20 of 60 product pairs in order to eliminate potential effects of fatigue or loss of motivation (see Supplement (4)). In none of these analyses did we find a significant negative relationship between social class and conformity. We also note that we were not able to follow Na et al.'s (2016) approach to investigate the relationship between social class and the number of changes in product choice when controlling for ethnicity in Study 1a/b. However, we do not think that the lack of this additional robustness check limits the interpretability of our results as it is unclear how the role of ethnicity in U.S. samples relates to its role in German samples.

Finally, we need to acknowledge that the possibility remains that factors like the mode of data collection might explain why our pattern of results differs from the one of Na et al.

(2016). While Na et al. conducted their studies as laboratory experiments, we conducted our experiments online. This also made it possible to collect data from samples which were more diverse in terms of SES compared to the original research. However, collecting data online might have, for example, lead to a reduced level of attention in comparison to a laboratory experiment. One way to investigate this potential explanation is to look at the attention checks included after each (manipulated) feedback on social preferences. Even though Na et al. (2016) also included these attention checks to make sure that participants understood the feedback, they did not report their results. Thus, these data cannot be used to qualify the results of our replication studies. In order to ensure a high data quality, we decided to exclude participants who failed a pre-registered number of attention checks. Including all participants regardless of their performance in these checks did not change the pattern of results regarding the relationship between social class and conformity.

Possibly, the different results may be due to differences in the type of products, samples, or the study mode. Still the finding of class-based differences in conformity seems limited in robustness and generalizability. Social influence might be equally strong across the social class spectrum suggesting that, for example, different than expected ads or public campaigns could employ descriptive norms to change individuals' behavior independently of the social class of the target group. Importantly, aside from questioning the robustness of class-based differences in conformity, our findings suggest that the relationship between social class and an interdependent self-construal is less robust as expected. One important avenue for future research on class-based differences in the view on the self may be to first clarify which factors determine if the relationship between social class and self-construal can be found.

Furthermore, follow-up studies that test the generalizability of the link between social class and conformity should be conducted. Specifically, future research should use psychometrically established measures of conformity and different choice contexts to establish the conditions under which a lower social class is related with a higher level of conformity. Even though the manipulation of social preferences developed by Na et al. (2016) was successful in evoking changes in choice, this paradigm has some drawbacks. Firstly, the desire to remain consistent with one's first choice potentially diminished the influence of the social feedback on the tendency to follow a majority choice. This might also explain the relatively low overall number of changes in choice in our studies as well as in the corresponding original study. Secondly, the paradigm only included hypothetical product choices without any real-life consequences. Previous research suggests that a lack of consequences following experimental product choice tasks can indeed affect product preferences (Klein & Hilbig, 2019). In order to arrive at a conclusive picture, it is essential that future research uses further, reliable measures of conformity.

Conclusion

Based on German and U.S. samples, the present research fails to replicate the negative relationship between social class and the tendency to follow a majority when it comes to product choices which was originally found by Na et al. (2016). There was also no empirical support for a negative relationship between social class and an interdependent self-construal which was proposed as underlying mechanism of the original findings. Social psychological

research has only recently begun to recognize the fundamental role of social class in the psychological make up and choice behavior of individuals. However, empirical evidence for class-based differences in conformity and an interdependent self-construal is scarce and faces some methodological shortcomings. The present research adds to recent attempts to assess the generalizability of social class effects and underscores the need to further clarify under which conditions class-based differences in conformity and self-construal can be found.

Funding Information

This work was supported by a scholarship from the Graduate School of Economic and Social Sciences at the University of Mannheim to the first author and by departmental budget resources of the Chair for Consumer and Economic Psychology, University of Mannheim. Additionally, this research was supported by the Open Access Publication Funds of the University of Mannheim.

Competing Interests

The authors declare that there are no potential conflicts of interest with respect to the research, authorship, and/or publication of this article.

Supplemental Material

Supplemental material to this article can be found on this paper's project page on the OSF: https://osf.io/fmc29/?view_only=34bc5c1f50df4bb9b46c33eb5b81e50d

Data Accessibility Statement

All materials, except for product pictures (which can be shared upon request), R code for all reported analyses, raw data, and codebooks are available on this paper's project page on the OSF: <u>https://osf.io/fmc29/?view_only=34bc5c1f50df4bb9b46c33eb5b81e50d</u> To obtain access to the product pictures used in the studies, please send an email to the corresponding author.

References

- Adler, N. E., Epel, E. S., Castellazzo, G., & Ickovics, J. R. (2000). Relationship of subjective and objective social status with psychological and physiological functioning:
 Preliminary data in healthy, White women. *Health Psychology*, *19*(6), 586–592. https://doi.org/10.1037/0278-6133.19.6.586
- Antonoplis, S. (2023). Studying socioeconomic status: Conceptual problems and an alternative path forward. *Perspectives on Psychological Science*, *18*(2), 275–292. https://doi.org/10.1177/17456916221093615
- Asch, S. E. (1956). Studies of independence and conformity: I. A minority of one against a unanimous majority. *Psychological Monographs: General and Applied*, 70, 1-70. https://doi.org/10.1037/h0093718
- Aydin, A. L., Ullrich, J., Siem, B., Locke, K. D., & Shnabel, N. (2019). The effect of social class on agency and communion: Reconciling identity-based and rank-based perspectives. *Social Psychological and Personality Science*, 10(6), 735–745. https://doi.org/10.1177/1948550618785162
- Bindra, S., Sharma, D., Parameswar, N., Dhir, S., & Paul, J. (2022). Bandwagon effect revisited: A systematic review to develop future research agenda. *Journal of Business Research*, 143, 305–317. https://doi.org/10.1016/j.jbusres.2022.01.085
- Boileau L.-A. (2022). Social class and the big two: Investigating a widespread yet understudied assumption on social class differences in self-concepts. [Doctoral dissertation, University of Mannheim].
- Carey, R. M., & Markus, H. R. (2017). Social class shapes the form and function of relationships and selves. *Current opinion in psychology*, 18, 123–130. https://doi.org/10.1016/j.copsyc.2017.08.031
- Cheung, F., & Lucas, R. E. (2016). Income inequality is associated with stronger social comparison effects: The effect of relative income on life satisfaction. *Journal of Personality and Social Psychology*, *110*(2), 332–341. https://doi.org/10.1037/pspp0000059
- Cho, E., Kim-Vick, J., & Yu, U.-J. (2022) Unveiling motivation for luxury fashion purchase among Gen Z consumers: need for uniqueness versus bandwagon effect. *International Journal of Fashion Design, Technology and Education, 15*, 24-34. https://doi.org/10.1080/17543266.2021.1973580
- Chou, T.-J., Dai, Q., Chang, E.-C., & Wong, V. (2013). Replacement between conformity and counter-conformity in consumption decisions. *Psychological Reports*, 112(1), 125– 150. https://doi.org/10.2466/01.07.09.PR0.112.1.125-150
- D'Amico, A., & Scrima, F. (2016). The Italian validation of Singelis's self-construal scale (SCS): A short 10-Item version shows improved psychometric properties. *Current Psychology*, 35, 159–168. https://doi.org/10.1007/s12144-015-9378-y
- Deutsch, M., & Gerard, H. B. (1955). A study of normative and informational social influences upon individual judgement. *Journal of Abnormal Psychology*, 51(3), 629– 636. https://doi.org/10.1037/h0046408
- Eom, K., Kim, H. S., & Sherman, D. K. (2018). Social class, control, and action: Socioeconomic status differences in antecedents of support for pro-environmental

action. *Journal of Experimental Social Psychology*, 77, 60–75. https://doi.org/10.1016/j.jesp.2018.03.009

- Faul, F., Erdfelder, E., Buchner, A., & Lang, A.-G. (2009). Statistical power analyses using G*Power 3.1: Tests for correlation and regression analyses. *Behavior Research Methods*, 41(4), 1149–1160. https://doi.org/10.3758/BRM.41.4.1149
- Fernández, I., Páez, D., & González, J. L. (2005). Independent and interdependent selfconstruals and socio-cultural factors in 29 nations [Le concept de soi indepéndant et interdépendent et les facteurs socio-culturels dans 29 nations]. *Revue Internationale de Psychologie Sociale*, 18(1-2), 35–63.
- Gobel, M. S., & Miyamoto, Y. (2024). Self- and other-orientation in high rank: A cultural psychological approach to social hierarchy. *Personality and Social Psychology Review*, 28(1), 54-80. https://doi.org/10.1177/10888683231172252
- Grossmann, I., & Varnum, M. E. W. (2011). Social class, culture, and cognition. Social Psychological and Personality Science, 2(1), 81–89. https://doi.org/10.1177/1948550610377119
- Hofstede G., Hofstede G. J., & Minkov M. (2010). *Cultures and organizations—Software of the mind: Intercultural cooperation and its importance for survival* (3rd ed.). McGraw-Hill.
- Imhoff, R., & Erb, H.-P. (2009). What motivates nonconformity? Uniqueness seeking blocks majority influence. *Personality and Social Psychology Bulletin*, 35(3), 309–320. https://doi.org/10.1177/0146167208328166
- Kassambara, A. (2023). rstatix: Pipe-friendly framework for basic statistical tests. R package version 0.7.2. https://CRAN.R-project.org/package=rstatix
- Kastanakis, M. N., & Balabanis, G. (2012). Between the mass and the class: Antecedents of the "bandwagon" luxury consumption behavior. *Journal of Business Research*, 65(10), 1399–1407. https://doi.org/10.1016/j.jbusres.2011.10.005
- Klein, S. A., & Hilbig, B. E. (2019). On the lack of real consequences in consumer choice research. And its consequences. *Experimental Psychology*, 66(1), 68–76. https://doi.org/10.1027/1618-3169/a000420
- Kraus, M. W., & Keltner, D. (2009). Signs of socioeconomic status: A thin-slicing approach. *Psychological Science*, 20(1), 99–106. https://doi.org/10.1111/j.1467-9280.2008.02251.x
- Kraus, M. W., Piff, P. K., Mendoza-Denton, R., Rheinschmidt, M. L., & Keltner, D. (2012). Social class, solipsism, and contextualism: How the rich are different from the poor. *Psychological Review*, 119(3), 546–572. https://doi.org/10.1037/a0028756
- Kraus, M. W., & Stephens, N. M. (2012). A road map for an emerging psychology of social class. Social and Personality Psychology Compass, 6(9), 642–656. https://doi.org/10.1111/j.1751-9004.2012.00453.x
- Kuhn, M. H., & McPartland, T. S. (1954). *Twenty Statements Test* [Database record]. APA PsycTests. https://doi.org/10.1037/t05100-000
- Leibenstein, H. (1950). Bandwagon, snob, and veblen effects in the theory of consumers' demand. *The Quarterly Journal of Economics*, 64, 183–207. https://doi.org/10.2307/1882692

- Leiner D. J. (2019). *SoSci survey* (Version 3.1.06) [Computer software]. https://www.soscisurvey.de
- Lynn, M., & Harris, J. (1997). The desire for unique consumer products: A new individual differences scale. *Psychology & Marketing*, *14(6)*, 601-616. https://doi.org/10.1002/(SICI)1520-6793(199709)14:6<601::AID-MAR5>3.0.CO;2-B
- Markus, H. R., & Stephens, N. M. (2017). Editorial overview: Inequality and social class: The psychological and behavioral consequences of inequality and social class: a theoretical integration. *Current opinion in psychology*, 18, iv–xii. https://doi.org/10.1016/j.copsyc.2017.11.001
- Na, J., Grossmann, I., Varnum, M. E., Kitayama, S., Gonzalez, R., & Nisbett, R. E. (2010). Cultural differences are not always reducible to individual differences. *Proceedings of the National Academy of Sciences of the United States of America*, 107(14), 6192– 6197. https://doi.org/10.1073/pnas.1001911107
- Na, J., McDonough, I. M., Chan, M. Y., & Park, D. C. (2016). Social-class differences in consumer choices: Working-class individuals are more sensitive to choices of others than middle-class individuals. *Personality & Social Psychology Bulletin*, 42(4), 430– 443. https://doi.org/10.1177/0146167216634043
- Neighbors, C., LaBrie, J. W., Hummer, J. F., Lewis, M. A., Lee, C. M., Desai, S., Kilmer, J.R., & Larimer, M. E. (2010). Group identification as a moderator of the relationship between perceived social norms and alcohol consumption. *Psychology of Addictive Behaviors, 24*, 522-528. http://dx.doi.org/10.1037/a0019944
- Piff, P. K., Kraus, M. W., Côté, S., Cheng, B. H., & Keltner, D. (2010). Having less, giving more: the influence of social class on prosocial behavior. *Journal of Personality and Social Psychology*, 99, 771-784. http://dx.doi.org/10.1037/a0020092
- Pirlott, A. G., & MacKinnon, D. P. (2016). Design approaches to experimental mediation. Journal of Experimental Social Psychology, 66, 29–38. https://doi.org/10.1016/j.jesp.2015.09.012
- Rao, H., Greve, H. R., & Davis, G. F. (2001). Fool's gold: Social proof in the initiation and abandonment of coverage by wall street analysts. *Administrative Science Quarterly*, 46(3), 502–526. https://doi.org/10.2307/3094873
- Roethke, K., Klumpe, J., Adam, M., & Benlian, A. (2020). Social influence tactics in ecommerce onboarding: The role of social proof and reciprocity in affecting user registrations. *Decision Support Systems*, 131, https://doi.org/10.1016/j.dss.2020.113268
- Rosenberg, M. (1965). Society and the adolescent self-image. Princeton University Press.
- Rosenberg, M., & Pearlin, L. I. (1978). Social class and self-esteem among children and adults. *American Journal of Sociology*, 84(1), 53–77. https://doi.org/10.1086/226740
- Salazar, H.A., Oerlemans, L., & van Stroe-Biezen, S. (2013). Social influence on sustainable consumption: Evidence from a behavioural experiment. *International Journal of Consumer Studies*, 37, 172-180. https://doi.org/10.1111/j.1470-6431.2012.01110.x
- Salmon, S. J., De Vet, E., Adriaanse, M. A., Fennis, B. M., Veltkamp, M., & De Ridder, D. T. D. (2015). Social proof in the supermarket: Promoting healthy choices under low selfcontrol conditions. *Food Quality and Preference*, 45, 113-120. https://doi.org/10.1016/j.foodqual.2015.06.004

- Schneider, S. M. (2019). Why income inequality is dissatisfying—perceptions of social status and the inequality-satisfaction link in Europe. *European Sociological Review* 35(3), 409–430. https://doi.org/10.1093/esr/jcz003
- Schumpe, B. M., Herzberg, P. Y., & Erb, H.-P. (2016). Assessing the need for uniqueness: Validation of the German NfU-G scale. *Personality and Individual Differences*, 90, 231-237. https://doi.org/10.1016/j.paid.2015.11.012
- Simonsohn, U. (2015). Small Telescopes: Detectability and the evaluation of replication results. *Psychological Science*, 26(5), 559–569. https://doi.org/10.1177/0956797614567341
- Singelis, T. M. (1994). The measurement of independent and interdependent self-construals. *Personality and Social Psychology Bulletin, 20*(5), 580–591. https://doi.org/10.1177/0146167294205014
- Singmann H., Bolker B., Westfall J., Aust F., & Ben-Shachar M. (2022). afex: Analysis of factorial experiments. R package version 1.1-1. https://CRAN.Rproject.org/package=afex
- Snyder, C. R., & Fromkin, H. L. (1977). Abnormality as a positive characteristic: The development and validation of a scale measuring need for uniqueness. *Journal of Abnormal Psychology*, 86, 518–527. http://dx.doi.org/10.1037/0021-843X.86.5.518
- Solt, F. (2019). The standardized world income inequality database, Versions 8-9. *Harvard Dataverse, V11*. https://doi.org/10.7910/DVN/LM4OWF
- Sparkman, G., Weitz, E., Robinson, T. N., Malhotra, N., & Walton, G. M. (2020). Developing a scalable dynamic norm menu-based intervention to reduce meat consumption. *Sustainability*, *12*(6), 2453. MDPI AG. http://dx.doi.org/10.3390/su12062453
- Stephens, N. M., Fryberg, S. A., & Markus, H. R. (2011). When choice does not equal freedom. Social Psychological and Personality Science, 2(1), 33–41. https://doi.org/10.1177/1948550610378757
- Stephens, N. M., Markus, H. R., & Townsend, S. S. M. (2007). Choice as an act of meaning: The case of social class. *Journal of Personality and Social Psychology*, 93(5), 814– 830. https://doi.org/10.1037/0022-3514.93.5.814
- Tainaka, T., Miyoshi, T., & Mori, K. (2014). Conformity of witnesses with low self-esteem to their co-witnesses. *Psychology*, 5, 1695-1701. https://doi.org/10.4236/psych.2014.515177
- Thomas, J. M., Ursell, A., Robinson, E. L., Aveyard, P., Jebb, S. A., Herman, C. P., & Higgs, S. (2017). Using a descriptive social norm to increase vegetable selection in workplace restaurant settings. *Health Psychology*, 36(11), 1026–1033. https://doi.org/10.1037/hea0000478
- Twenge, J. M., & Campbell, W. K. (2002). Self-esteem and socioeconomic status: A metaanalytic review. *Personality and Social Psychology Review*, 6(1), 59–71. https://doi.org/10.1207/S15327957PSPR0601_3
- Unkelbach, F., John, M., & Vogel, V. (2023). Jumping on the bandwagon: The role of voters' social class in poll effects in the context of the 2021 German federal election. *Politische Vierteljahresschrift, 64*, 51–78. https://doi.org/10.1007/s11615-022-00417-3

- U.S. Census Bureau (2021). Educational Attainment in the United States: 2020. Retrieved January 24, 2022, https://www.census.gov/data/tables/2020/demo/educational-attainment/cps-detailed-tables.html
- Zhang H., Talhelm T., Yang Q., & Hu C. S. (2021). High-status people are more individualistic and analytic-thinking in the west and wheat-farming areas, but not ricefarming areas. *European Journal of Social Psychology*, 51, 878–895. https://doi.org/10.1002/ejsp.2778